# Quantum-Inspired Resource Optimization for 6G Networks: A Survey

Muhammad Omair Butt, Nazar Waheed, Trung Q. Duong, and Waleed Ejaz

Abstract—The Internet of things (IoT) drives an exponential surge in computing and communication devices. Consequently, it triggers capacity, coverage, interference, latency, and security issues in the existing communication networks. The forthcoming sixth-generation (6G) networks will address these issues by providing comprehensive solutions. In particular, quantum communication technology can potentially address the challenges of 6G networks. However, its implementation requires substantial infrastructure upgrades. Therefore, the quantum-inspired (QI) techniques offer an intermediate resort due to their ability to utilize the classical communication infrastructure for design and implementation. Hence, we review QI techniques in this survey that address radio resource optimization challenges across various communication aspects, including channel assignment, reconfigurable intelligent surfaces, spectrum sensing, unmanned aerial vehicle-assisted networks, and related areas. The analysis explores diverse aspects, including objectives, constraints, problem characterization, proposed solutions, and lessons learnt. Research indicates that QI techniques offer advantages such as faster convergence and reduced complexity, providing promising solutions to complex optimization problems in communication networks. Furthermore, we identify the future directions, research gaps, and ongoing challenges from the QI radio resource optimization dataset.

*Index Terms*—Machine learning, meta-heuristics, optimization, quantum communication, quantum computation, quantuminspired, sixth-generation

#### I. INTRODUCTION

Sixth-generation (6G) networks set ambitious performance goals, promising significantly enhanced data rates, lower latency, and greater capacity than fifth-generation (5G) standards. Fig. 1 illustrates key performance indicators, enabling technologies, and diverse use cases for 6G networks. The key performance indicators are categorized into various aspects such as capacity, coverage, data rate, and others [1]. 6G networks aim for peak data rates in the terabits per second range, a substantial leap from 5G's peak data rate of 20 gigabits

M. O. Butt and W. Ejaz are with the Department of Electrical Engineering, Lakehead University, ON, Canada. (e-mail: mbutt1@lakeheadu.ca; wejaz@lakeheadu.ca).

Nazar Waheed is with the Faculty of Computing Information Systems, Higher Colleges of Technology, UAE. (e-mail: nwaheed@hct.ac.ae).

T. Q. Duong is with the Faculty of Engineering and Applied Science, Memorial University, St. Johnâs, NL A1C 5S7, Canada, and with the School of Electronics, Electrical Engineering and Computer Science, Queenâs University Belfast, BT7 1NN Belfast, U.K., and also with the Department of Electronic Engineering, Kyung Hee University, Yongin-si, Gyeonggi-do 17104, South Korea (e-mail: tduong@mun.ca).

We acknowledge the support of the Natural Sciences and Engineering Research Council of Canada (NSERC). The work of T. Q. Duong was supported in part by the Canada Excellence Research Chair (CERC) Program CERC-2022-00109.

Corresponding author is Waleed Ejaz.

per second. While 5G offers user-experienced data rates in the range of megabits per second, 6G targets 1 gigabit per second. Additionally, 6G aspires to achieve sub-millisecond latency, enabling real-time applications, compared to 5G's 1-millisecond latency. Moreover, 6G seeks to support the connection density of ten times higher with up to 10 million connections per square kilometer and double the mobility speed, reaching 1000 km/h. It also aims for centimeter-level positioning accuracy, a hundredfold improvement in energy efficiency, 99.9999% availability, and ultra-reliable connectivity. This can be achieved by integrating heterogeneous networks encompassing terrestrial and non-terrestrial elements such as cellular, fiber optic, satellite, and unmanned aerial vehicles (UAVs). However, achieving optimized interoperability among these diverse networks poses significant challenges [2].

The authors of [3] investigated resource allocation strategies for space-air-ground integrated networks, which are critical for achieving the service-everywhere requirement in the 6G networks. As radio resource optimization complexities increase with the advent of 6G networks, innovative approaches such as dynamic optimization, mathematical optimization, game theory, and artificial intelligence-based methods become essential. Similarly, the authors of [4] highlighted the limitations of current communication systems regarding capacity and reliability, proposing a cell-less networking approach for radio resource management in 6G networks. This approach enables user connectivity with multiple access points, improving system capacity and mitigating interference while achieving energy efficiency and optimized latency. Consequently, the authors of [5] introduced an optimization framework focused on minimizing delay and energy consumption for emergency vehicle users in UAV-assisted air-ground integrated networks, enhancing emergency response through federated learning and prioritized experience replay.

The optimization of constrained wireless resources is necessary to support the evolving requirements of users and applications. Despite the significant progress of wireless networks, achieving the desired key performance indicators of 6G networks requires addressing several issues that include devising an efficient resource optimization technique while considering different parameters such as limited bandwidth, high data rate, low latency, extended range, and ensuring unprecedented security. The high density of nodes in Industry 4.0 and the Internet of things (IoT) complicates large-scale resource optimization. The 6G networks can incorporate quantum communication and computing technologies to ensure seamless and massive device connectivity [6].

Quantum technology is in the early stages of develop-



Fig. 1. The key performance indicators of 6G, its enabling technologies and use cases [1].

ment and can potentially optimize radio resources in tasks such as channel assignment, IoT, UAV trajectory planning, and related areas. Transitioning from classical to quantum communication paradigms is challenging because it involves integrating quantum elements into existing communication infrastructure [7]. The lack of infrastructure and expertise makes large-scale adoption of quantum communication difficult. Therefore, quantum-inspired (QI) techniques are advantageous due to their compatibility with existing systems and incremental improvements, enabling a gradual transition to true quantum communication. QI optimization techniques offer a promising approach to handle the complexity and scalability requirements of 6G networks, which will integrate numerous IoT devices, UAVs, and ultra-reliable low latency (URLLC) applications. In such a complex scenario, classical optimization methods may struggle to optimize radio resources in dynamic environments, such as spectrum allocation and energy usage. For instance, clustering numerous ground users based on constraints like latency, channel gain, and quality of experience can be managed more effectively. Recent research views QI techniques as a middle ground, providing quantumlike enhancements while being implementable on classical communication networks [8]. Table I lists the acronyms used in this survey paper.

Fig. 2 demonstrates that quantum information science is a key driver of QI optimization, enhancing classical metaheuristics and machine learning (ML) methods. By incorporating quantum-like enhancements, it improves the optimization of classical communication networks. QI techniques are preferred for their ability to enhance efficiency and performance without requiring infrastructure upgrades. These techniques

		TAE	BLE I			
List	OF	ACRONYMS	USED	IN	THIS	PAPER

	Acronyms
5G	Fifth-generation
6G	Sixth-generation
AI	Artificial intelligence
AR	Augmented reality
BS	Base station
GPS	Global positioning system
IoT	Internet of things
IoV	Internet of vehicles
IRS	Intelligent reflective surfaces
MIMO	Multiple input multiple output
ML	Machine learning
NN	Neural network
NOMA	Non-orthogonal multiple access
NP-hard	Non-deterministic polynomial time hard
PSO	Particle swarm optimization
QI	Quantum-inspired
QKD	Quantum key distribution
QML	Quantum machine learning
QNN	Quantum neural network
QPSO	Quantum particle swarm optimization
QRL	Quantum reinforcement learning
QUBO	Quadratic unconstrained binary optimization
RIS	Reconfigurable intelligent surface
RL	Reinforcement learning
UAV	Unmanned aerial vehicles
URLLC	Ultra-reliable low latency communication
VR	Virtual reality



Fig. 2. Quantum-inspired optimization connecting domains.

are typically classified as meta-heuristics and ML approaches. Despite using classical infrastructure, they significantly enhance scalability and the ability to handle complexity, which is crucial for 6G networks integrating IoT devices, UAVs, and applications requiring URLLC. QI techniques can efficiently manage numerous ground users in dynamic scenarios such as spectrum allocation and energy usage, focusing on constraints like latency, channel gain, and quality of experience. Recent research positions QI techniques as a middle ground, incorporating quantum-like enhancements while remaining adaptable to classical communication networks [8]. These advancements in QI algorithms enable enhancement in training duration and capacity within ML-based supervised learning frameworks, like support vector machines. QI techniques improve communication, computation, and privacy by integrating quantum concepts, including parallelism and superposition. They can be applied to diverse domains, including cryptography, financial modeling, ML, and optimization, by reducing computational complexity, exploring large solution spaces, implementing post-quantum cryptography, and mitigating noise and errors typical in quantum-like systems.

#### A. Literature Search Process

This paper identifies various QI techniques available in the literature to address the resource optimization problems in the 6G networks. The prime focus is to analyze objectives, constraints, problem types, and solutions to address concerns arising due to incorporating non-classical and QI computational solutions. A dataset comprising papers from 2019 to 2024 is gathered by employing different queries in the IEEE, Science Direct, Arxiv, American physical society (APS), and optical society of America (OSA) repositories as tabulated

9	Research question: Identify the significance of quantum-inspired optimization in 6G environment?
Ŏ	Searching keywords:
	Quantum Communication, Quantum-Inspired, Quantum Key Distribution, Wireless Networks, 6G, Machine Learning, Meta-Heuristics, Optimization, Neural Network, Reinforcement Learning, Genetic Algorithm
Ŏ	Search databases:
Ţ	IEEE Xplore, Science Direct, Arxiv.org, APS and OSA
Ŏ	Article type:
I	Journal, early access or magazine paper
ŏ	Publication timeframe:
Ť	Paper published between 2019 and 2024
¥	r aper published between 2013 and 2024
Q	Selection criteria:
	The paper is in English
	• The use case involves quantum communication or quantum-inspired optimization
ŏ	Resultant paper:
Ŭ	Covers quantum communication fundamentals and components
	<ul> <li>Compares principles of quantum and classical communication</li> </ul>
	<ul> <li>Offers insights into different quantum-inspired optimization techniques</li> </ul>
	Reviews recent literature (2019-2024) on quantum-inspired optimization
	Categorizes radio resource management strategies into various domains
	Examines a vertical farming case study using quantum-inspired techniques
	Discusses future research directions and challenges

Fig. 3. Selection of related papers in the collection process.

in Table II. The search queries include "quantum communication", "quantum-inspired", "quantum key distribution", "wireless networks", 6G, "machine learning", meta-heuristics, optimization, "neural network", "reinforcement learning", "genetic algorithm", "quantum key distribution", "resource optimization", and closely related keywords. The survey builds on a dataset comprising different types of papers such as surveys, journals, and magazine papers. Fig. 3 depicts paper collection criteria. The technical discussion of mathematical modeling and implementation details available in the literature is beyond the scope of this paper and can be considered as future work. The papers of interest are selected based on their relevance and the optimization problem formulation.

#### B. Contributions

The key contributions of our survey paper are summarized as follows:

• Principles and Components of Quantum Communication: We provide an overview of the fundamentals and components of quantum communication, while providing a comparison with the principles of classical communication systems.

 TABLE II

 Articles search queries and the dataset details.

Query	Database	Total	Filtered
Quantum-Inspired AND (Heuristics OR "Machine Learning" OR "Game Theory"	IEEE	191	31
OR "Federated Learning" OR "Neural Network" OR "Reinforcement Learning" OR	Science Direct	237	9
"Support Vector Machine" OR Optimization OR "Wireless Communication" OR 6G)	Arixv.org	197	6
Quantum-Inspired AND (Wireless OR 6G) AND Optimization	IEEE	14	12
Quantum-Inspired AND Optimization AND "Wireless Communication"	APS	52	9
Quantum-Inspired AND Optimization	OSA	4	4

- Comprehensive Literature Review (2019-2024): We review and analyze recent papers from 2019 to 2024 to study how QI approaches are used in optimizing wireless network resources. This review ensures that our analysis stays up-to-date by incorporating the most recent research and findings.
- Categorization of QI Optimization Techniques: We organize QI optimization techniques into meta-heuristics, ML and specialized solutions. We detail the methods, key factors, challenges, and how these techniques can be applied to the important services expected in the upcoming 6G networks.
- Novel Categorization Framework: To the best of our knowledge, this is the first attempt to arrange the QI optimization approaches systematically across different areas of communication. This involves comparing our work with previous research, demonstrating our paper's new perspectives, and establishing its relation to other studies in the field.
- Lessons Learnt: We provide insights into the research gaps identified in the literature on QI radio resource optimization scenarios, highlighting areas that require further investigation.
- Use Case of Vertical Farming: We provide insights into an intriguing case study of vertical farming, demonstrating how QI optimization techniques can enhance efficiency and productivity.
- Future Research Directions and Challenges: We discussed various potential future avenues of research for QI optimization techniques and outlined the challenges that need to be addressed.

#### C. Organization of paper

Fig. 4 illustrates the roadmap of our paper, highlighting the section numbers and providing the sub-categorizations. Section II provides a detailed review of existing survey studies on quantum communication and computing. Section III presents basic concepts of quantum communication and enlists the quantum communication components to develop an understanding of the reference area. Section IV explains the QI optimization approaches such as meta-heuristics and ML. Section V categorizes and comprehensively analyzes the reviewed resource optimization problems and their solution techniques. Section VI discusses the use case of QI optimization in vertical and digital farming. Section VIII discusses future directions and challenges. Finally, section VIII concludes the survey.

#### II. EXISTING SURVEYS

In this section, we analyze key insights from available surveys on quantum communication and computing for wireless networks. We categorize the existing surveys into different domains such as quantum computing and applications, quantum Internet [9] and security, post-quantum malware detection, resource optimization and network performance. We identify the correlations between key findings from previous surveys and utilize this information to build our survey paper.

Section I: Introduction <ul> <li>Literature search process</li> <li>Contributions</li> <li>Organization of Paper</li> </ul>			
Section II: Existing surveys  • Quantum computing and applications • Quantum Internet and security • Post-quantum malware detection • Resource optimization and network performance			
Section III: Quantum communication <ul> <li>Fundamental principles</li> <li>Components of quantum communication</li> <li>Quantum versus classical communication: Enabling 6G Applications</li> </ul>			
Section IV: Quantum-inspired optimization • Quantum-inspired meta-heuristics • Quantum-inspired machine learning • Specialized quantum-inspired optimization techniques • Quantum-Inspired Optimization: Techniques and Insights			
Section V: Radio resource optimization in 6G networks • Quantum-inspired channel assignment • Quantum-inspired optimization for edge devices • Quantum-inspired optimization of reconfigurable intelligent surfaces • Quantum-inspired spectrum sensing and localization • Quantum-inspired resource optimization in UAV assisted networks • Quantum-inspired power control optimization			
Section VI: Use case: Quantum-inspired vertical farming model			
Section VII: Future research directions and challenges			
Section VIII: Conclusion			

**Roadmap of article** 

Fig. 4. Organization of this paper.

#### A. Quantum Computing and Applications

The authors of [10] presented a survey on quantum computing-based techniques to solve physical and network layer issues in wireless communication systems. This paper identified the application areas and analyzes the complexity of several quantum computing techniques. The target categories for optimization include big data analysis, indoor localization, joint channel estimation, multi-objective routing, multi-user detection, and transmission. The quantum-based solutions can significantly reduce complexity budgets compared to their classical counterparts. A comprehensive survey on OI metaheuristics was presented in [11]. This review encompassed various algorithms, such as evolutionary, multi-objective, social evolution, and swarm intelligence, providing valuable insights into identifying practical QI meta-heuristics candidates for quantum computers. Various benchmark problems, including 0-1/multidimensional knapsack, combinatorial optimization, continuous optimization, logic circuit design, and traveling salesman, were considered to evaluate QI algorithms. The evaluation of the QI Acromyrmex evolutionary algorithm on 15 benchmark test functions showed superior performance compared to the genetic and advanced genetic quantum algorithms. Similarly, the real-QI evolutionary algorithm outperformed fast evolutionary programming, particle swarm optimization (PSO), and stochastic genetic algorithm. Lastly, the quantum artificial bee colony algorithm performed the best compared to the quantum swarm evolutionary algorithm and genetic

algorithm in optimizing the three benchmark test functions.

In [12], an overview of OI-ML was presented for the security and resource optimization of 6G networks. The computational and time complexity of classical ML algorithms often makes them unsuitable for handling large-scale data analytics, which is essential in 6G networks. The QI-ML algorithms are favorable as they leverage the superposition of quantum states and parallelism for exponential speed up in the learning process. An overview of quantum algorithms for realtime optimization was presented in [8]. The authors considered quantum optimization primarily for solving combinatorial problems using heuristic approaches. A comprehensive survey on QI-ML was presented in [13]. The authors categorized quantum-based ML approaches into quantum ML (QML), QI-ML, and hybrid classical-quantum ML. The QI-ML techniques refer to developing classical ML algorithms while incorporating the principles of quantum mechanics and do not require quantum computers. The authors have highlighted resources to implement OI-ML approaches and open issues.

The authors of [14] reviewed the rapid advancements in quantum technologies and emphasized their significance in academia and industry. The paper categorized quantum computing components into quantum computers, quantum cryptography, QML, and quantum networks. The foundational quantum mechanics principles, such as superposition and entanglement, were explained. The difference between universal quantum computers and quantum annealers was highlighted. Furthermore, the implications of decoherence and quantum error correction were discussed. The paper served as a comprehensive resource for identifying future research directions and potential applications of quantum computing and communication in various domains.

The authors of [15] explored advanced concepts in quantum communications, including superadditivity, superactivation, and causal activation. These phenomena challenge the classical notions of channel capacity. The paper provided insights into using these phenomena to enhance communication protocols and contribute to the development of future quantum networks. It was highlighted that quantum channels can transmit information through entangled states, whereas classical channels cannot. This paper can be considered for understanding the implications of quantum mechanics in communication technologies.

#### B. Quantum Internet and Security

The authors of [6] comprehensively reviewed the infrastructure requirements and technological advancements required for the quantum Internet. The authors anticipated that the quantum Internet would incorporate security measures based on quantum key distribution (QKD) algorithms for networking classical and quantum entities. However, the fragility of quantum information due to decoherence is a significant challenge in realizing the quantum Internet. Similarly, the interfacing of discrete and continuous variables is essential, where the discrete variables refer to data represented by discrete features, such as the polarization of single photons detected by single-photon detectors, and the continuous variable refers to information encoded onto an optical field constituting infinite Hilbert space (to analyze vectors using inner products).

The authors of [16] analyzed the integration of quantum computing with drone technology, identifying quantum algorithms and architectures that enabled efficient and secure communication within the Internet of quantum drones. The roles of quantum drones and satellites were examined in futuristic scenarios and real-time applications. The integration of artificial intelligence (AI) and ML with quantum drones was also analyzed. The survey highlighted technological advancements in quantum computing and drone technology while reviewing the security implications and investigating the effects of quantum attacks on different cryptosystems.

The authors of [17] reviewed quantum secure direct communication and considered it a promising method to communicate securely. This approach allowed secret messages to be sent directly over quantum channels without requiring pre-shared cryptographic keys. The increase in threats to the classical cryptographic systems from quantum computing necessitates novel communication methods. The paper discussed conceptual foundations, experimental implementations, and different secure communication protocols, showing their advantages over QKD. A hybrid approach was proposed, combining quantum secure direct communication with classical systems to improve the security of next-generation communications, highlighting the potential of the quantum Internet.

In [18], the authors reviewed the cybersecurity enhancements required in satellite-based communication. The authors dissect the literature on satellite communication security into physical layer security and cryptographic techniques. They preferred the symmetric cryptographic solutions over publickey infrastructure setups due to the swiftness of encryption and decryption procedures. The key replacement and update are time-expensive tasks, and the key establishment is usually performed on satellite-to-ground and satellite-to-satellite links in satellite communication. They highlighted that incorporating quantum computation in satellite links can offer a promising level of security. However, the passive-side channel attacks can compromise the physical layer security of the QKD signal, and researchers have yet to explore this aspect in the literature. Furthermore, the authors of [19] considered physical layer security an emerging and alternative method to classical cryptography. The authors highlighted that quantum cryptography utilizes Heisenberg's uncertainty principle and avoids using public keys. They classified the physical layer security schemes into artificial noise, channel adaptation, and code. A security and privacy-based overview of 6G networks was given while focusing on the underlying technologies of physical, connection, and service layers.

#### C. Post-Quantum Malware Detection

The authors of [20] overviewed AI-driven malware detection techniques and categorized them into continuous learning and enhanced explainability through visualization. They considered quantum technology an enabler of unprecedented security and a strong candidate for future communication networks. The survey emphasized the necessity of balanced datasets to learn and accurately detect malware. Similarly, the privacy-enhancing technologies were comprehensively reviewed in [21] based on post-quantum cryptographic primitives, particularly when quantum computers become a reality. They provide insights into various threats in IoT and intelligent infrastructures. They preferred digital signatures and latticebased schemes in comparison to other post-quantum cryptographic schemes for multiple constrained IoT and intelligent infrastructure platforms.

#### D. Resource Optimization and Network Performance

In [22], the authors provided an overview of the quantum search algorithms for maximizing wireless system performance. Additionally, they presented a feasibility analysis of QML techniques across various use cases. The optimization techniques were oracle-based, variational quantum eigensolver, and quantum approximate optimization algorithm. These techniques find application in energy efficiency, error correction, physical layer, security, and signal intelligence scenarios. In [7], the authors comprehensively reviewed the quantum network structure and its development stages. They highlighted the key differences compared to the classical networks in terms of the routing metrics. A robust and scalable entanglement-assisted network design incorporating heterogeneity and cost-effectiveness was deemed necessary. Furthermore, the standardization of such networks is essential to successfully transitioning towards heterogeneous networking schemes. The authors in [2] reviewed the challenges of resource optimization in integrated 6G networks. They only considered the optimization of classical networking elements. Their proposed hierarchical paradigm developed on four integration levels, including space-air-ground-sea network segments.

Lessons Learnt: Table III provides a summary of the contributions of the existing papers surveyed in relation to QI optimization. A critical analysis is performed to identify the current scope of research work that includes quantum basics, quantum components, optimization techniques, and our remarks. This analysis forms the foundation of our paper, which focuses on comprehensively reviewing and categorizing the QI optimization schemes. The QI communication and computation is an initial step towards a fully realizable integrated quantum-classical network topology. This domain provides several benefits over classical communication, such as lowering the computational complexity, improving energy efficiency, and incorporating the quantum effects for better security.

#### **III. QUANTUM COMMUNICATION**

This section introduces basic principles governing quantum communication and provides an overview of the key components to enable quantum communication networks. Fig. 5 illustrates a quantum communication system composed of various elements based on the logic of experimental setups presented in [23], [24]. The tunable laser generates unpolarized photons, which then pass through a polarization state generator to become oriented into specific polarization states relevant to quantum information. This information can be transmitted through an optical fiber or free-space channel, where noise can adversely affect the transmitted qubits, potentially leading to decoherence. Following this, quarter-wave and half-wave plates are used to modify the polarization states of the photons, converting circular and diagonal polarizations into linear states. Finally, a beam splitter separates the linear polarization states into their respective paths, where photon detectors can measure the incoming photons.

#### A. Fundamental Principles

Quantum communication is built on the principles of quantum mechanics, enabling secure and efficient information transfer. Quantum communication systems offer significantly higher capabilities compared to classical communication systems due to the unique properties of quantum states. This section discusses foundational concepts such as qubits, superposition, quantum entanglement, the no-cloning theorem, and quantum teleportation. These principles support quantum communication's advantages and potential applications in modern technology.

1) Quantum Bits: Qubits are the basic informational unit in quantum computing. They can coexist simultaneously in multiple states, allowing for a higher degree of efficiency in computations compared to classical bits. Environmental factors such as noise and temperature fluctuations can impair delicate qubits, leading to different errors or decoherence of entangled states [25]. An n-qubit quantum system can simultaneously represent  $2^n$  different states. In contrast, classical bits in classical communication systems only exist in one of two possible states of 0 or 1, and they do not experience quantum impairments such as decoherence.

The Dirac notation concisely represents quantum states, probability distributions, normalization, and operators. Dirac notation can analyze quantum states through the vectors of bra  $(\langle x |)$  and ket  $(|y\rangle)$ , where x and y are basis states. The inner product of bra and ket vectors helps compute probabilities and determine overlaps between quantum states, revealing their essential properties. This can be represented as [26]:

$$\langle x|y \rangle = \begin{bmatrix} x_1 & x_2 & \dots & x_n \end{bmatrix} \times \begin{bmatrix} y_1 \\ y_2 \\ \vdots \\ y_n \end{bmatrix},$$
 (1)

$$\langle x|y\rangle = (x_1y_1) + (x_2y_2) + \dots + (x_ny_n),$$
 (2)

where  $x_1, ..., x_n$  and  $y_1, ..., y_n$  can represent the coefficients or amplitudes of quantum states and *n* represents the dimension of vectors. Similarly, the qubits can be expressed in terms of their basis states, and density matrix ( $\rho$ ) can be used to represent both mixed and pure quantum states as [27]:

$$|\psi\rangle = \alpha |x\rangle + \beta |y\rangle, \tag{3}$$

TABLE III Summary of existing surveys on QI radio resource optimization. Ref.: Reference; QB: Quantum Basics; QC: Quantum Components; QIO: Quantum-Inspired Optimization; MH: Meta-heuristic; ML: Machine Learning; RRO: Radio Resource Optimization.

Year	Ref.	QB	QC	Q MH	IO ML	RRM	Remarks
		I	1	1	Qu	iantum (	Computing and Applications
2010	[10]		~		v		Reviewed different types of quantum algorithms for wireless commu-
2019	[10]	<b>~</b>	^	~	^		nication, including cryptographic, meta-heuristics and search.
2020	[11]	X	X	1	X	X	Comprehensively reviewed the QI meta-heuristics algorithms.
							Reviewed quantum computing and its potential through ML and dis-
2022	[12]	1	X	X	1	X	cussed the suitable applications for enhancing resource optimization
							and network security in 6G networks.
							Analyzed the potential of QI real-time optimization in the context
2022	[8]	1	X	1	1	×	of 6G and provided a brief discussion on heuristic and ML-based
							optimization.
2022	[15]	1	x	x	x	x	Reviewed advanced phenomena in quantum communication towards
2022	[15]						channel capacity enhancement and does not focus on QI optimization.
							Comprehensively reviewed the QI-ML while explaining its different
2023	[13]	x	x	x		x	research areas, recent progress, real-world uses, and future directions,
2025	[15]				•		thereby the paper served as a valuable resource for both researchers
							and practitioners.
2023	[14]	1	1	x	x	x	Comprehensively reviewed the components of quantum Internet, and
2025	[1]	•	•				does not focus on QI optimization approaches.
						Quantu	Im Internet and Security
							Comprehensively reviewed the applications, enabling technologies,
2021	[6]	1	1	X	X	X	functions, and open challenges of the quantum Internet and there is
							no discussion on optimization.
2021	[16]	x	x	x	x	x	The paper focused on quantum drones-based communication while
2021	[10]						considering post-quantum cryptography and security.
							Comprehensively reviewed the link-layer security threats, solutions,
2022	[18]	X	X	X	X	X	and challenges in deploying and operating satellite communication
							systems while considering QKD and other cryptographic techniques.
2024	[17]	1	x	x	x	x	The paper reviewed quantum secure direct communication and its
	[]				•		integration with classical systems for the quantum Internet.
					-	Post-Qu	antum Malware Detection
							Comprehensively reviewed data-driven malware detection literature,
							with emphasis towards continuous learning techniques and novel
2022	[20]	X	X	X	X	×	visualization methods to efficiently detect malware and enhance
							malware identification and its interpretability in large-scale, data-
							intensive environments.
				Re	esourc	e Optim	ization and Network Performance
							The paper examined how QML and quantum search algorithms can
2023	[22]		x	×			enhance wireless system performance, explored potential solutions,
2025	[22]	•			•	•	and addressed their applicability and feasibility challenges in this
							context while briefly discussing security and error correction.
							Provided a comprehensive review of entanglement-assisted quantum
2023	[7]	1	1	x	x	1	networks while discussing the network structure, development stages,
2025	[']	•	•			•	and challenges in constructing wide area networks and does not focus
							on QI optimization.
Provided a categorization of resource opti		Provided a categorization of resource optimization techniques, dis-					
2023	[2]	X	X	X	X	×	cussed challenges and future directions for space, aerial, ground, and
					<u> </u>		sea integrated networks and there is no discussion on QI optimization.
	Our						Provided a detailed analysis of the fundamentals of quantum com-
2024	Survey	1	1	1	1	1	munication, categorization of various QI solutions available in the
	Survey						literature, prospective use cases, and future directions.



Fig. 5. Example of transmission and detection in quantum communication.

$$\rho = |\Psi\rangle\langle\Psi| = \begin{pmatrix} |\alpha|^2 & \alpha\beta^*\\ \alpha^*\beta & |\beta|^2 \end{pmatrix}, \tag{4}$$

where  $\psi$ , x and y denote the quantum state, horizontal and vertical polarization, respectively. The coefficients  $\alpha$  and  $\beta$ are the amplitudes for respective basis states. Hence, (3) demonstrates the qubit's capacity to embody both polarizations simultaneously. The commonly used polarizations in quantum communication include linear, diagonal, and circular polarizations. These are analytically expressed in (5) to (10) [27]:

$$\rightarrow = |x\rangle,$$
 (5)

$$\uparrow = |y\rangle, \tag{6}$$

$$\nearrow = \frac{1}{\sqrt{2}} \left( |x\rangle + |y\rangle \right),\tag{7}$$

$$\searrow = \frac{1}{\sqrt{2}} (|x\rangle - |y\rangle),$$
 (8)

$$= \frac{1}{\sqrt{2}} \left( |x\rangle - i|y\rangle \right), \tag{9}$$

$$= \frac{1}{\sqrt{2}} \left( |x\rangle + i|y\rangle \right), \tag{10}$$

where  $\rightarrow$ ,  $\uparrow$ ,  $\nearrow$ ,  $\searrow$ ,  $\curvearrowright$ , and  $\curvearrowright$  represent horizontal, vertical, 45° diagonal, -45° diagonal, clockwise, and anticlockwise polarizations, respectively. Furthermore, the measurement of polarization states affects the qubits, and the probability of obtaining a particular polarization state can be computed through the inner product of the corresponding bra and ket vectors. Specifically, (11) can be used to measure the probabilities for other polarization states can be derived from their respective bra and ket representations. Unlike qubits, the state of classical bits can be read without disturbing the system.

$$P(\to) = |\langle \to |\psi \rangle|^2. \tag{11}$$

To enhance the control and manipulation of quantum states, the quantum Hamiltonian learning algorithm (a function representing the kinetic and potential energy of the system) [28] and quantum state tomography (reconstructing a quantum state by measuring a system in different bases) [29] can be employed. These techniques facilitate the efficient management of quantum systems and the accurate reconstruction of quantum states. 2) Superposition: According to this principle, a qubit can exist in multiple states simultaneously. It enables parallelism by simultaneously allowing computations on multiple inputs and empowers quantum and QI algorithms to explore various potential solutions more efficiently [27]. In contrast, a classical bit can only be in a 0 or 1 state, leading to sequential computations by processing one input at a time. Although both quantum and classical communication systems share foundational principles of information processing, the superposition of qubits represents a transformative shift in how information is processed and transmitted. It offers significant advantages in computational efficiency and security. Fig. 6 depicts a qubit in a superposition state, with the  $(|0\rangle)$ ,  $(|1\rangle)$  and z denote the horizontal, vertical, and superposition dimensions, respectively.

A qubit's state remains uncertain until measured, illustrating that quantum states are inherently unpredictable. This uncertainty before measurement distinguishes it from classical methods, where the state is always clear and definite. For instance, Grover's algorithm provides a quadratic speedup for searching an unsorted database, allowing it to search through N items in approximately  $\sqrt{N}$  steps, compared to N steps in a classical search [30]. Similarly, Shor's algorithm, designed for factoring large integers, achieves exponential speedup compared to classical algorithms [25]. These quantum algorithms exploit the principles of superposition and interference to evaluate multiple possibilities simultaneously, a capability beyond the reach of classical systems.

3) Quantum Entanglement: It allows qubits to become correlated with each other, regardless of the distance separating them. The state of one qubit directly influences the state of its entangled partner. This property enables secure data transfer through techniques like QKD to create unbreakable keys and enhances computational power by allowing multiple simultaneous calculations [9]. Careful management is required to maintain the entangled states of qubits, and it is indeed possible to entangle multiple qubits, creating complex states that increase computational capabilities. Although it is theoretically feasible to entangle multiple qubits, its practical implementation is challenging due to decoherence and noise.

A density matrix captures the intrinsic correlation between entangled qubits. Bell states are specific two-qubit states that exhibit maximal entanglement, making them vital for quantum information protocols such as cryptography and teleportation [26]. These states can be generated using quantum gates or spontaneous parametric down-conversion in optical experiments. The analytical representation of a Bell state, expressed via the outer product, illustrates the entangled nature of qubits.



Fig. 6. Qubit representation through Bloch sphere, showing its existence in a superposition state  $(|\psi\rangle)$ .

In (12), an entangled Bell state  $(|\psi^+\rangle)$  is generated from two qubits  $|\psi_1\rangle$  and  $|\psi_2\rangle$ , that are initially in  $|0\rangle$  state [27]. Furthermore, the quantum circuits are computational models manipulating qubits using quantum gates, effectively demonstrating quantum entanglement. Fig. 7 shows that applying a Hadamard gate to qubit  $|\psi_1\rangle$  and a controlled-NOT gate with  $|\psi_1\rangle$  as control and  $|\psi_2\rangle$  as target leads to the maximally entangled Bell state as [25]:

$$|\Psi^{+}\rangle = \frac{1}{\sqrt{2}}(|00\rangle + |11\rangle).$$
 (12)

Furthermore, developing robust quantum communication networks requires addressing challenges like reliable entanglement creation and quantum memory management [9]. In contrast, classical bits are independent and do not exhibit the interconnected behavior seen in quantum systems. Classical communication faces issues like signal degradation over long distances and the need for effective security measures to protect data during transmission. Classical networks employ error correction codes and parallel processing to improve reliability and efficiency. Both classical and quantum systems utilize optimization techniques to tackle the challenges of respective domains. Classical algorithms, such as simulated annealing and genetic algorithms, can enhance communication processes, while quantum techniques, such as quantum annealing and quantum neural networks (QNN), improve entanglement protocols. Furthermore, hybrid approaches combining both systems can leverage classical computation for tasks that do not require quantum properties while employing effective entanglement protocols.

The authors of [31] formulated a mixed integer non-linear programming problem to maximize the successful generation of entangled qubit pairs. They solved it using an interior-point algorithm while considering constrained quantum memory capacity. Similarly, the authors of [32] minimized the latency of entangled pair generation with linear time complexity by selecting optimal swapping trees while accounting for a 3/2 factor for throttled trees and capacity violations. Furthermore, the authors of [33] maximized the entanglement distribution rate by designing a remote entanglement distribution protocol incorporating graph theory and classical flow networks for a chain of homogeneous repeaters. The authors of [34]



Fig. 7. Quantum entanglement of qubits.

experimentally demonstrated high visibility entanglement of photon-pair shared among three users within  $17 \ km$ reach for a tree-shaped optical network. Furthermore, while entanglement assistance significantly boosts communication rates, its unreliability in entanglement-breaking channels renders the combination of the assisted and unassisted coding suboptimal [35].

4) No-Cloning Theorem: This theorem states that creating an exact copy of an arbitrary quantum state is impossible. This limitation arises from the principle of linearity, which requires that if a quantum copier gets a superposition of states as input, then the output must also reflect that superposition [25]. Therefore, any attempts to clone quantum states disrupt this requirement, as illustrated in Fig. 8. Consequently, the eavesdroppers attempting to intercept or duplicate quantum information can be swiftly detected through the QKD protocol. However, cloning specific states, such as orthogonal states, is possible, but the quantum copier must be designed explicitly for such states. The classical communication systems do not possess this property, and an eavesdropper can easily copy the classical information without detection, compromising information confidentiality.

5) Quantum Teleportation: It involves transferring quantum information between two distant locations without physically moving the quantum state [26]. Fig. 9 shows that the process of teleportation begins with entangled qubits at both the source and the destination nodes. The source gets a third qubit whose information needs to be transmitted. It performs a Bell measurement on the entangled partner and the incoming qubit. This measurement collapses the quantum states and produces a classical outcome. The measurement result is then sent to the destination node via a classical communication channel, adhering to the speed of light limit. At the destination, the entangled qubit undergoes a quantum operation based on the received classical information and transforms into a replica of the third qubit's state. Therefore, quantum teleportation depends on transmitting measurement results through classical communication methods. Ouantum teleportation can alleviate the challenges of a data-driven metaverse through quantum anonymous communication and variational quantum computing [36]. In contrast, classical communication systems allow easy copying and transmission of information without quantum considerations.

#### B. Components of Quantum Communication

Classical communication systems can incorporate QI optimization techniques without imposing infrastructural updates on network providers. However, the cohesive



Fig. 8. Failure in cloning arbitrary quantum states.

operation of classical and quantum systems requires only software-based updates, such as encoding and decoding procedures. In contrast, implementing purely quantum communication requires specific enabling components, explained in this subsection.

1) Qubit Source and Detector: Quantum communication depends on accurate generation and detection of qubit states. Common qubit sources include coherent states, entangled photons, single photons, and squeezed states, each designed for the respective quantum communication protocols. These sources utilize devices such as lasers, non-linear crystals, parametric oscillators, and quantum dots to generate the required quantum states [27]. The quantum dot chains can be employed for wireless communication due to reduced threshold current densities and strong vertical emission for terahertz quantum cascade lasers at room temperature [37]. On the contrary, classical communication systems utilize continuous signals, such as electrical pulses or radio waves, to generate classical bits, which do not require the same level of precision.

Similarly, the quantum detection methods vary depending on the qubit generation technique. Devices such as charged coupled device cameras, quasiparticle traps, radio frequency superconducting quantum interference devices, single-photon detectors, and transition edge sensors are commonly used to detect quantum states with high sensitivity [26]. Classical bit detectors are relatively simpler and typically use photodiodes or other conventional detectors that do not require the same level of precision in resolving states.

2) Quantum Gates: Quantum gates perform operations on qubits that exist in superposition. They serve as the fundamental building blocks of quantum circuits, enabling various computational tasks on qubits. Common types of quantum gates include the controlled-NOT, Hadamard, Pauli, Rotation, Swap, and Toffoli gates, each designed for specific functions within quantum computations [26]. In contrast, classical gates process information in a binary manner by operating on binary values of bits. Unlike classical gates, quantum gates exploit the principles of quantum mechanics, such as entanglement and superposition, to perform complex operations and enhance computational power. It allows quantum algorithms to tackle tasks often inefficient for classical systems, such as factoring large numbers or simulating quantum systems, which would require impractical amounts of time and resources with classical computing methods.

The gate set tomography can also be employed to evaluate the performance of a group of quantum gates through



Fig. 9. The process of teleporting quantum information.

experimental measurement data. In this regard, authors of [38] proposed a compressive gate set tomography technique by employing low-rank approximations and random gate sequences to enhance the measurement efficiency and reduce experimental complexity. QI optimization approaches view quantum gates as essential conceptual tools rather than merely physical operations. In this framework, quantum gates are crucial in implementing quantum algorithms and protocols that enhance overall performance and efficiency. By harnessing the unique properties of quantum gates, QI strategies can enable higher computational capability, even within classical computing environments.

3) Quantum Repeaters: Quantum and classical repeaters primarily aim to extend communication distances. However, they use different principles and have distinct features. Classical repeaters amplify and regenerate signals to mitigate transmission losses and noise impairments. In contrast, quantum repeaters entangle quantum states across different segments of the communication link and perform entanglement swapping to establish a direct link between distant nodes without amplifying or copying the quantum signal [27]. Fig. 10 illustrates a chain of quantum repeaters connecting two distant quantum nodes, with the quantum channel being either fiber optic or free space. Quantum repeaters mitigate signal impairments in extended quantum channels by utilizing quantum memory to store and synchronize quantum states effectively. Entanglement swapping performed by the repeaters enables reliable and scalable communication between distant nodes. These features make quantum repeaters indispensable for overcoming the limitations of direct quantum transmission.

Quantum repeaters can be classified into trusted and non-trusted types. Trusted quantum repeaters assume secure internal operations and merely facilitate the connection. Non-trusted quantum repeaters do not rely on assumptions of trustworthiness and instead utilize cryptographic methods to safeguard against potential eavesdropping. Ultimately, quantum repeaters maintain the integrity of delicate quantum information over long distances, enabling quantum Internet and secure quantum communication systems. Different types of quantum repeaters include distributed phase reference, entanglement swapping, error correction-based, hybrid, measurement-based, and memory-based. Furthermore, there is a need to explore the performance of noisy intermediate-scale quantum networks in terms of scalability and integration with classical communication networks to achieve the global quantum Internet [39].



Fig. 10. Quantum repeaters connecting the quantum nodes.

4) Quantum Memory: It stores qubits that exist in superposition, enabling enhanced processing capabilities. However, retrieving data from quantum memory necessitates measuring qubits, which collapses their superposition into a definite state and alters their original condition. Quantum memory operates through various physical systems, such as atoms, ions, or photons, to encode quantum states in qubits. Techniques that regulate light and matter interactions enable quantum memories to store quantum states, such as superpositions and entangled states, for varying durations, depending on the technology and environmental conditions [27].

In contrast, classical memory stores deterministic information in binary form through technologies such as dynamic random-access memory. It relies on magnetic or electronic means for reliable storage. Classical memory has a simpler structure and does not require the advanced capabilities of quantum memory, which must tackle challenges like decoherence and signal loss to preserve quantum states. Overall, while both classical and quantum memories serve the purpose of storing information, they operate through fundamentally different mechanisms.

5) Quantum Information Processing: It pertains to the analysis and operations of quantum information using quantum systems. It encompasses many tasks, including cryptography, data analysis, ML, and optimization. Qubit states are manipulated through quantum gates, which exploit quantum properties such as entanglement, parallelism, and superposition to achieve significant speedups for various computational tasks and algorithms [25]. In contrast, classical communication relies on bits for information processing and often faces limitations in handling complex problems. Many classical algorithms struggle to find efficient solutions for large-scale optimization, factoring large integers, and simulating quantum systems, where the computational complexity can grow exponentially with input size. For instance, the authors of [40] proposed a quantum multi-clustering technique utilizing a variational quantum eigensolver to tackle real-world clustering problems on noisy intermediate-scale quantum devices, significantly reducing computational requirements through the use of non-orthogonal qubit states.

6) *Quantum Security Protocols:* The unique properties of qubits, such as entanglement and the no-cloning theorem, offer security that greatly surpasses classical systems, which can be compromised by cryptanalysis. Quantum security protocols, including QKD, quantum digital signatures, quantum coin flip-

ping, and quantum secure multi-party computation, leverage these quantum properties to achieve unparalleled security [26]. QKD specifically uses qubits to exchange cryptographic keys securely. However, practical QKD systems face challenges like optimizing key rates, extending communication distances, improving scalability, detecting eavesdropping, and managing quantum memory. Addressing these challenges requires advancements in techniques such as quantum error correction, quantum repeaters, and enhanced protocols.

The authors of [41] proposed a network scheduling policy to maximize secure key rates among quantum devices. The secret-key rates were also improved by employing a dynamic attenuation scheme for free-space channels through real-time monitoring and adjusting transmittance [42]. Similarly, in [43], a time-bin entanglement-based QKD scheme was proposed that utilized the E91 protocol on free-space optical channels, aiming to maximize the secret keys rate while considering the system average symbol error rate and outage probability. In [44], upper bounds on device independent keys secure against a non-signaling adversary were obtained through squashed secrecy monotones, which quantify the maximum secure key generation rate from a quantum state. A squashed nonlocalitybased security condition was formulated to identify the domains where secure key distillation was impossible and provide the tightest known upper bounds.

Furthermore, a holistic QKD-based communication technique was introduced in [45] to ensure security for aerial, ground, and space nodes. Additionally, the authors of [46] integrated QKD with classical data traffic to extend transmission distances by minimizing spontaneous Raman scattering noise, enhancing security in wavelength division multiplexing networks. The authors of [47] proposed an algorithmic framework to strengthen the resilience of QKD-integrated optical networks against link failures via a secret-key recovery technique. A three-layered QKD semantic information communication architecture for resource optimization and routing decisions was developed in [48]. Similarly, an NP-complete optimization problem was formulated in [49] to minimize the number of QKD repeaters in fiber optic networks and enable secure communication over wide areas. Lastly, the feasibility of QKD protocols, including discrete variable and distributed phase reference, was analyzed in [50], considering noise and wind impairments on various communication links. A continuous variable OKD scheme for terahertz bands using multicarrier modulation was proposed in [51]. Additionally, the quantum Byzantine fault tolerance protocol can mitigate issues related to security, noise, and faults in applications like the human-centric metaverse [52].

7) Quantum Error Correction: It is essential to maintain the integrity of quantum communication, especially in the presence of various noise impairments. It addresses challenges such as amplitude damping, bit flips, decoherence, depolarization, and phase damping but induces significant overheads. Various techniques for quantum error correction include redundant encoding, stabilization codes, syndrome measurement-based unitary correction, and topological protection [26]. These methods detect and correct errors by employing measurements that provide information about errors without fully collapsing the superposition and entanglement of qubits. A simpler procedure called purification can be used to detect errors [27].

In classical communication, Gaussian noise is the dominant noise-contributing element, and error correction methods such as Hamming and Reed-Solomon codes detect and correct errors by adding redundant bits to the transmitted data. Although both approaches seek to preserve data integrity, quantum error correction functions within a fundamentally different framework that reflects the unique characteristics of quantum mechanics. Adaptive frameworks can effectively estimate timevarying quantum noise in diverse communication scenarios [53], which can be leveraged to devise suitable quantum error correction techniques.

Calderbank-Steane-Shor codes combine two classical linear codes to protect quantum information from bit-flip and phaseflip errors, with the proposed quantum low-density-generatormatrix codes outperforming quantum low-density-parity check codes in correcting errors in phase-flip dominant depolarizing channels [54]. Furthermore, non-Calderbank-Steane-Shor quantum low-density-generator-matrix codes can address the limitations of classical non-Calderbank-Steane-Shor constructions for enhanced error correction. The authors of [55] performed specific row operations on quantum parity check matrices to devise new codes with better performance and increased design flexibility as compared to the existing Calderbank-Steane-Shor quantum low-density generator matrix and lowdensity parity check codes.

## C. Quantum versus Classical Communication: Enabling 6G Applications

This subsection provides a comparative analysis of classical and quantum communication, summarizing key discussions from Section III. The parameters in Table IV highlight the advantages of quantum communication, particularly in its use of qubits, which enable probabilistic representation, superposition, and greater parallelism, while classical communication is limited to deterministic bit transmission. Quantum technologies will play a significant role in 6G networks by offering enhanced security through QKD and facilitating innovations such as quantum-enhanced sensor networks and secure multi-party computations. Alongside classical communication, it remains essential in scenarios where established infrastructure exists, and simplicity of implementation is required. An intelligent trade-off between both paradigms is crucial for maximizing their potential in future networks. Quantum communication can enhance various applications expected in the 6G networks, listed follows:

- **High-speed connectivity:** Through superposition and entanglement to transmit data swiftly.
- Enhanced security: Through QKD relays to protect sensitive information.
- Massive IoT support: Efficient data handling from numerous IoT devices through exponential storage capabilities.
- **Real-Time applications:** High parallelism can aid in meeting the real-time requirements of augmented reality (AR) and virtual reality (VR) experiences.

In summary, Section III has provided a foundational understanding of the differences between classical and quantum communication. It offers a brief overview of the fundamental quantum principles, including the concept of qubits, the phenomenon of superposition, qubits entanglement, and the significance of the no-cloning theorem. Additionally, it has discussed the functionalities of critical components such as quantum gates, repeaters, memory units, and error correction mechanisms, highlighting their roles in the quantum communication scenario. Lastly, a comparative analysis is presented between quantum and classical communication while presenting the plausible 6G applications enabling quantum technologies.

#### IV. QUANTUM-INSPIRED OPTIMIZATION

Rapid growth of edge computing devices, IoT sensors, and mobile users requires significant improvements in wireless communication networks. These improvements include increasing the capacity and data rate and reducing interference, latency, and similar factors. Accommodating the advanced features anticipated in future 6G networks may prove challenging for classical networks. Alongside, implementing a global quantum communication network is challenging due to current infrastructure and technological limitations, including the need for quantum repeaters to maintain entanglement across vast distances, the high sensitivity of quantum states to environmental disturbances, and the absence of a well-established global quantum framework. Furthermore, integrating quantum

	TABLE IV		
COMPARISON BETWEEN	CLASSICAL AND	QUANTUM	COMMUNICATION.

Parameters	<b>Classical Communication</b>	Quantum Communication
Unit	Bits (0 or 1)	0, 1, or superposition of both
Representation	Deterministic	Probabilistic
Parallelism	Limited due to sequential bit operations	High due to superposition
Speed	Slow (signal processing and transmission delays)	Faster (superposition and entanglement)
Measurement	Non-intrusive	Intrusive (qubit collapses to certain state)
Security	Vulnerable to eavesdropping	Enhanced due to no-cloning and QKD
Repeaters	Amplify signals over distances	Extend entanglement over long distances
Gates	Deterministic operations on bits	Complex operations and parallelism on qubits
Memory	Linear information storage in conventional cells	Exponential storage capability due to superposition



Fig. 11. Classification of QI optimization approaches.

systems with classical networks requires significant advances in both hardware and protocols. Therefore, a potential solution lies in applying quantum principles to classical networks, referred to as QI techniques, to achieve quantum-like benefits. Fig. 11 depicts the categorization of QI optimization techniques into meta-heuristics, ML, and specialized algorithms. This section reviews QI algorithms and their conceptual frameworks, includes flowcharts of the optimization process, summarizes key strengths and weaknesses, examines applications in radio resource management, and analyzes the frequency of their use in optimizing radio resources.

#### A. Quantum-Inspired Meta-Heuristics

Meta-heuristics can be broadly categorized as nature-based or biologically-inspired algorithms seeking near-optimal solutions through repeated solution space exploration. QI metaheuristics enhance exploration without compromising exploitation by leveraging quantum principles such as entanglement, tunneling, and superposition. Unlike classical meta-heuristics, which explore solutions sequentially, QI meta-heuristics utilize quantum parallelism to simultaneously evaluate multiple solutions. In a classical register, only one of the  $2^n$  possible values can be stored, whereas an n-qubit register can represent all  $2^n$  states at once, allowing for more efficient searches. This results in superior convergence speed and solution quality for large-scale optimization tasks.

Figs. 12 and 13 illustrate generic flowcharts of classical and QI meta-heuristics optimization algorithms, respectively. Classical meta-heuristics optimization methods rely on deterministic population initialization and solution generation, whereas QI optimization begins by initializing qubit states in superposition, leveraging quantum parallelism to evaluate multiple solutions simultaneously. Furthermore, QI techniques employ probabilistic selection and use quantum gates to update qubit states, enabling enhanced exploration and faster convergence compared to classical methods. Overall, both approaches iteratively generate candidate solutions, evaluate fitness, and check stopping criteria, and QI optimization introduces quantum-specific features. In particular, in QI meta-heuristics, the three-dimensional properties of qubits are simplified into a two-dimensional representation using rotation angles and probability amplitudes to guide the optimization process. These algorithms can also optimize decision variables in collaborative inference frameworks for intelligent mobile services with linear time complexity [56], [57]. Here, we briefly explain the algorithms identified in the reviewed literature and their QI conceptualization.

1) **Biologically-Inspired Meta-Heuristics Optimization**: These optimization algorithms leverage principles from natural phenomena to address complex challenges. This subsection presents a curated list of biologically-inspired meta-heuristics, highlighting key insights and implementations from the filtered literature that specifically focuses on solutions for radio resource optimization.

i. Quantum-Inspired Ant Colony Optimization: It relies on artificial ants depositing pheromones on the paths they traverse, thereby influencing the likelihood of subsequent ants choosing those paths. Each ant probabilistically selects a route by considering the pheromone levels and heuristic information, such as the distance from the food source to their colony. The classical ant colony optimization algorithm has a  $O(M^2)$  complexity, where M refers to the number of IoT nodes. In contrast, its QI implementation demonstrates a lower computational complexity as indicated in [58]. QI ant colony optimization integrates principles of quantum computing by using qubits to represent pheromone information, which is updated through quantum gates. This approach enables parallelism due to the superposition of states, allowing for the exploration of multiple solutions simultaneously.

**ii.** *Quantum Clone Whale Optimization:* The classical whale optimization algorithm enhances local search capability for NP-hard optimization problems while maintaining lower computational complexity. However, it tends to fall into local optima. To address this, the quantum clonal whale optimization algorithm was proposed in [60] for optimized clustering of wireless sensor nodes. This algorithm follows a stepwise approach, including population initialization, fitness



Fig. 12. General flowchart of classical meta-heuristics optimization [59].

calculation, encirclement and predation, bubble-net attacking, a random search for prey, clonal expansion, quantum gate operations, and termination. The quantum clonal whale optimization algorithm is robust and slightly more complex than the classical counterpart due to the clonal expansion, which guides individuals in the population toward the optimal solution. Its computational complexity is  $3 \times i_{max} \times P_s \times M$ , where  $i_{max}$ ,  $P_s$ , and M refer to the maximum number of iterations, population size, and sensor nodes, respectively. Furthermore, it demonstrates superior energy balancing for wireless sensor nodes compared to benchmark algorithms, where noticeable energy fluctuations occur.

**iii.** *Quantum-Inspired Evolutionary Algorithm:* It is a generational, population-based iterative algorithm designed to address complex optimization problems. In each iteration, quantum gates such as Hadamard, NOT, and rotation gates are employed to update individuals in the population, guiding them toward better solutions. Key characteristics of this algorithm include the probabilistic representation of individuals (as qubits), evaluation functions, and population dynamics. The underlying mechanisms leverage quantum principles, such as the probabilistic nature of qubits and quantum gate operations, to effectively explore and exploit the search space of a specific optimization problem. This algorithm is especially useful for solving combinatorial and non-linear integer programming optimization problems, where finding the optimal solution within polynomial time is difficult.

For instance, in [61], joint caching and user association optimization for adaptive bit-rate video streaming was tackled through the QI evolutionary algorithm, resulting in the lowest total content delivery delay and reduced UAV transmission power compared to classical randomized caching and user



Fig. 13. General flowchart of QI meta-heuristics optimization [57].

association methods. Furthermore, the NP-hard optimization problem of reducing the peak-to-average power ratio in multicarrier modulation systems was solved through the QI evolutionary algorithm with a computational complexity of  $P_s \times G \cdot O(N_c \log N_c)$ , where  $P_s$ , G, and  $N_c$  represent the population size, generation count, and the number of sub-carriers, respectively [62]. The benchmarks used for comparisons, such as the artificial bee colony and cross-entropy-based selective mapping algorithms, showed significantly higher complexity regarding the number of fast Fourier transforms required to meet the desired criteria of the peak-to-average power ratio.

iv. Quantum-Inspired Genetic Algorithm: It effectively combines the probabilistic and heuristic strengths of classical genetic algorithms with quantum principles and improves the exploration and exploitation in complex search spaces. The qubit chromosomes are initially considered in a linear superposition of equally likely states having  $1/\sqrt{2}$  value. They provide binary candidate solutions that aid in selecting the best solution. The iterative process of updating candidate solutions continues until a termination condition is met. Quantum gates are used to update qubit chromosomes for the next-generation.

In [63], the overall complexity of the QI genetic algorithm for resource-constrained project scheduling is found to be  $O(G.K.n_d^2)$  through a stepwise approach where G, K, and  $n_d$  represent the number of generations, resource types, and decision variables, respectively. The QI genetic algorithm converges faster than the classical genetic algorithm. Similarly, a quantum-genetic-based optimal link-state routing protocol was proposed for mobile Ad hoc networks to optimize the multi-point relay set selection and perform adaptive routing decisions [64]. This strategy showed the least network topology control overhead compared to benchmarks such as classical optimal link-state routing protocol, optimal link-state routing protocol-neighborhood state self-adaptive update, and cartography-enhanced optimal link-state routing protocol.

v. Quantum Particle Swarm Optimization (QPSO): It is inspired by the foraging behavior of birds and enhances the classical PSO algorithm. Originally, PSO was designed to solve complex optimization tasks by simulating the movement of particles within a multidimensional search space. In QPSO, each particle's position and velocity are initialized, and their movements are guided by both their individual best positions and a global best position, evaluated through a fitness function. The algorithm iterates until a predefined maximum number of iterations is reached. QPSO has a time complexity of  $O(i_{max}N_p)$ , where  $i_{max}$  and  $N_p$  represent the maximum number of iterations and particles, respectively.

Fig. 14 illustrates the iterative optimization process of QPSO, which integrates quantum principles like superposition and tunneling to enhance solution space exploration. Unlike conventional PSO, QPSO employs probabilistic states and QI updates to overcome local convergence issues, ensuring effective exploration and optimal convergence. QPSO algorithm was used in [65] to maximize energy efficiency in device-to-device edge computing, where it significantly outperformed the conventional PSO algorithm due to its ability to converge toward a global suboptimal solution. QPSO can be a powerful tool for solving mixed integer non-linear programming and other complex optimization problems related to resource allocation [66]–[74].

vi. *Quantum-Inspired Seagull Optimization:* The classical seagull optimization algorithm mimics the behavior of seagulls as they move in groups towards a destination while skillfully avoiding collisions with one another. After navigating these challenges, the seagulls optimize their movement based on local information. Additionally, they display a local search behavior characterized by a spiral movement when attacking other birds. In contrast, the QI seagull optimization algorithm incorporates the neighboring seagulls to enhance interaction, thereby influencing each seagull's flight direction.

Therefore, the algorithm proposed in [76] employed a variable angular-distance rotation gate to update the probability magnitudes of the seagulls, allowing for more adaptive behavior. This algorithm utilizes dynamic archives to store Pareto solutions, improving solution space exploration. The algorithm efficiently scales the number of seagulls,



Fig. 14. General flowchart of QPSO algorithm [75].

exhibiting a time complexity of  $O(M_s^2)$ , where  $M_s$  represents the number of seagulls. Experimental results validate the superiority of the QI seagull optimization algorithm over different benchmark algorithms, such as multi-objective PSO and non-dominated sorting genetic algorithm II, particularly in terms of convergence and distribution across diverse multi-objective optimization tasks. Overall, enhanced solution diversity and convergence speed make the QI seagull optimization algorithm a plausible choice for solving complex optimization problems.

vii. Quantum-Inspired Termite Colony Optimization: It employs a colony of quantum termites to navigate among the solutions in a D-dimensional space, where D is the maximal dimension of the optimization task. The positions of individual termites represent potential solutions, and the highest-performing positions are considered the local and global optima. The algorithm iteratively updates the pheromone contents using quantum rotation angles and pheromone levels at each termite's position. All pheromone levels are initially set to zero, and each termite uses its information and that of its neighbors to adjust its trajectory based on the defined search radius. The algorithm concludes upon reaching the maximum number of iterations.

In [77], the quantum termite colony optimization algorithm was applied to maximize the energy efficiency and transmission capacity of MIMO systems, which belong to the category of NP-hard optimization problems. Its computational complexity was  $O(i(5P_s + 3P_sn))$ , where i,  $P_s$ , and n represent the number of iterations, population size, and dimension of each quantum termite (vector), respectively. The quantum termite colony optimization algorithm demonstrated superior convergence speed and population diversity due to its reliance on quantum evolution strategies and global information sharing. It effectively and reliably solved energy efficiency optimization problems with multiple constraints compared to various benchmark algorithms, such as termite colony optimization, PSO, backtracking search optimization, and half-power-random resource allocation.

Table V highlights the advantages and disadvantages of biologically inspired QI meta-heuristics optimization approaches. It offers key insights into their strengths and limitations in diverse scenarios.

2) Nature-Inspired Meta-Heuristics Optimization: These optimization algorithms harness techniques from natural processes to solve optimization problems. This subsection provides insights into the QI nature-inspired algorithms for solving radio resource optimization issues.

i. Quantum-Inspired Backtracking Search: It is an innovative optimization technique that combines the backtracking search optimization algorithm and principles of quantum computing. The algorithm employs L quantum individuals, each defined by n dimensions that reflect the optimization task's complexity. The QI backtracking search evaluates the fitness of each quantum individual based on a specific optimization objective and maintains a record of historically measured states to enhance the search process. The historical memory set updates systematically by recording the current state when a randomly generated variable  $\alpha 1$  is less than another variable  $\alpha 2$  and otherwise keeps the previous state unchanged. This mechanism and the random shuffling of stored solutions promote diversity in the solutions generated during each iteration. New quantum individuals are then produced via evolutionary strategies that leverage historical data and the states of other quantum individuals, ensuring a comprehensive search for optimal solutions.

In [78], the QI backtracking search was employed on massive MIMO systems for joint antenna selection and power allocation, exhibiting a complexity of O(i(5L+4Ln)), where i, L, and n refer to the number of iterations, population size, and dimension of each individual, respectively. The QI backtracking search optimization algorithm demonstrated superior convergence accuracy compared to various benchmarks, including PSO, discrete PSO, artificial physics optimization,

TABLE V
KEY INSIGHTS INTO THE BIOLOGICALLY-INSPIRED META-HEURISTICS QI OPTIMIZATION TECHNIQUES

QI Technique	Ref.	Remarks
OI Ant Colony		• Improved network-lifetime for energy constrained sensors.
Ontimization	[58]	• Lower node energy consumption.
optimization		• Significantly longer execution time per round than traditional counterparts.
Quantum Clone		• Robust clustering and extended network lifetime in energy system sensor networks.
Whale	[60]	• Highest count of working rounds of network.
Optimization	[00]	• Ensures high accuracy and stability in node energy control.
		• Lowest average data transmission delay.
		• Faster convergence of algorithm for video caching.
QI Evolutionary	[61],	• Lower content delivery delay.
Algorithm	[62]	• Offers superior peak-to-average power ratio and bit error rate performance.
		• Lower implementation complexity.
OI Genetic		• Lowest network control overhead in mobile Ad hoc networks.
Algorithm	[64]	• Superior reliability in terms of successful packet delivery rate.
Aigonum		• Lowest end-to-end packets transmission delay between nodes.
	[65]–	• Solves complex optimization tasks more accurately and quickly than classical methods.
Quantum Particle		• Improves scalability through parallelism.
Swarm		• Optimizes user satisfaction and resource efficiency.
Ontimization	[74]	• Improves IoV performance with high throughput, low packet loss, and minimal delays.
Optimization		• Enhances energy efficiency in device-to-device-assisted edge computing.
		• Involves higher decision-making overhead.
QI Seagull	[76]	• Superior LIAV path planning performance
Optimization	[/0]	• Superior OAV paul plaining performance.
OI Termite		• Enables diversity of population.
Colony	[77]	• Global information sharing leads to swift convergence.
Ontimization		• Achieves optimal transmission capacity in massive MIMO networks.
Optimization		• Stable and reliable in energy efficiency optimization.

and the classical backtracking search optimization algorithm. This QI backtracking search optimization algorithm has enhanced performance due to its ability to effectively incorporate information from current and historical quantum individuals, which mitigates the issues of local convergence. Therefore, the QI backtracking search optimization algorithm is robust in solving complex wireless communication systems related optimization problems.

**ii.** *Quantum-Inspired Bald Eagle Search:* It integrates principles of quantum computing with the natural behaviors of bald eagles, specifically their selection, searching, and swooping tactics. In this algorithm, a population of quantum bald eagles explores a D-dimensional search space, where each eagle's position is represented by quantum bits. The algorithm iteratively updates these positions based on fitness evaluations, facilitating efficient convergence toward global optima. The core mechanism of QI bald eagle search involves simulating the hunting behaviors of bald eagles, where they share information and adjust their positions using quantum rotation angles. Dynamic adaptation enhances search strategies, improving both efficiency and effectiveness.

This approach was used in [79] to maximize secrecy energy efficiency in IRS-aided wireless communication, exhibiting a computational complexity of  $O(iP_s(3D + 2))$ , where *i*,  $P_s$ , and *D* represent the number of iterations, population size, and dimension of the optimization problem, respectively. Upon comparison with benchmarks such as bald eagle search, PSO, and Dinkelbach algorithms, the complexities of all techniques were similar due to equivalent population sizes and dimensions. However, the QI bald eagle search showed robustness in prolonged iterations, whereas benchmark algorithms showed premature convergence.

iii. Quantum-Inspired Gravitational Search Algorithm: It integrates principles from gravitational search algorithm and quantum computing to address quadri-valent problems effectively. The individual qubits are represented as a pair of parameters  $(\alpha, \beta)$ , which can exist in states 0, 1, 2, 3, or a superposition of these states. Terms  $\alpha^2$  and  $\beta^2$  define the state probabilities with their sum equal to one. The population of agents is initialized for a D-dimensional search space. An observation function is employed by individual agents to collapse the quantum states to a specific value, leading to a set of candidate solutions. The fitness of solutions is evaluated, and upon finding a better fitness, the set of best solutions is updated. The mass of each agent is calculated based on its fitness relative to the worst solution, influencing its movement through a gravitationally-inspired mechanism. The angular velocity of each qubit is adjusted using a rotational quantum gate, which facilitates dynamic exploration of the solution space. Quadri-valent QI gravitational search algorithm was used in [80] to solve the NP-hard problem of operational mode assignment to wireless sensor nodes, and it increased the lifetime of the network compared to the classical benchmarks such as binary genetic algorithm and binary PSO.

iv. Quantum-Inspired Annealing: It is a computational technique that leverages principles of quantum mechanics to solve combinatorial optimization problems. The optimization problems are encoded into a suitable format for quantum annealers by transforming them into quadratic unconstrained binary optimization forms. The algorithm aims to find the lowest energy state of a Hamiltonian, which corresponds to an optimal solution. The NP-complete wavelength assignment problem in optical communications has been solved in [81] by employing a QI annealing algorithm called a simulator coherent Ising machine (a device using magnetic spin interactions to solve optimization problems). The algorithm reduces the number of quadratic unconstrained binary optimization problems and enhances robustness against suboptimal solutions. This transformation employs an extended binary vector incorporating both wavelength assignments and auxiliary variables, allowing for a more efficient minimization process.

QI annealing shows better scalability and has a lower average solution time, especially for larger problem sizes, as compared to classical benchmarks such as largest degree first, commercial Gurobi optimization software, and open-source mixed integer programming solver-GNU linear programming kit. In [82], an adaptive bit-rate control methodology was formulated as a quadratic unconstrained binary optimization problem, where increased bit-rate and reduced rebuffering maximize the quality of experience. The digital annealer performed better in solving the optimization problem in most cases compared to different benchmarks such as rate-based, buffer-based, and model-predictive control. The digital annealers have also been employed in [83] and [84] to identify optimized migration plans for legacy optical networks, where they exhibited superior performance compared to classical mixed integer programming. Similarly, several other applications of quantum annealing confirm the superiority in solving combinatorial optimization problems [30], [85], and [86]. This suggests that QI annealing may effectively shift the tractability boundaries for solving large-scale combinatorial problems.

v. *Quantum-Inspired Tabu Search:* It identifies suboptimal solutions by iteratively exploring the neighborhood of a current solution while employing a tabu list to avoid cycling back to previously explored solutions. This approach generally begins with an initial candidate solution, incorporating a greedy search to evaluate neighboring solutions, and employs a memory structure that guides the search away from previously visited states. In contrast, the QI tabu search algorithm enhances this framework by leveraging quantum mechanics to represent and manipulate potential solutions effectively. By measuring the probability distributions of qubits, QI tabu search identifies promising solutions while maintaining a diverse search space. This uniqueness enables QI tabu search to adaptively refine its search trajectory and improve solution space exploration.

QI tabu search was employed to identify a better duplex mode in [88], where it demonstrated superior efficacy compared to different benchmarks such as exhaustive search, greedy search, and fixed duplex mode. Specifically, it initializes a quantum population where each qubit can collapse into states representing different duplex modes for access points. It achieved higher secrecy spectral efficiency gains, which validate its effectiveness in meeting the secrecy requirements

 TABLE VI

 Key insights into the nature-inspired meta-heuristics QI optimization techniques.

QI Technique	Ref.	Remarks
QI Backtracking		• Enhanced diversity of solution and search ability.
Search	[/0]	• Superior secrecy capacity and secrecy energy efficiency.
QI Bald Eagle	[70]	• Significantly higher convergence performance.
Search	[//]	• Superior secrecy energy efficiency.
		• Superior lifetime of wireless sensor networks.
QI Gravitational	[80]	• The fitness value improves with increase in population size.
Search	[00]	• Parallel implementation increases throughput without impacting power of processor.
		• Effective in handling extensive networks.
	[81]	• Improves scalability and runtime for graph coloring and combinatorial optimization.
Quantum	[81], [82], [85]	• Comparable performance to industry-grade solvers for combinatorial tasks.
Annealing		• Stable increase in performance at higher levels of parallelism.
		• Superior performance for adaptive video streaming in static and dynamic scenarios.
ΟΙ		• Enables fast and accurate convergence through diversity and strong search capabilities.
Social Emotional	[87]	• Superior power control strategy for massive MIMO networks.
Optimization		• The power control strategy is highly sensitive to hardware impairments.
		• Classical counterpart is prone to local convergence.
OI Tabu Saarah	1991	• Higher number of possible solutions leads to faster convergence.
Qi iabu seafch	[00]	• Secrecy spectral efficiency comparable to exhaustive search in massive MIMO systems.

in cell-free massive MIMO systems. The iterative mechanism of QI tabu search, which includes updating the tabu list and applying quantum gates for state transitions, enables it to navigate the solution space with complexity often lower than that of exhaustive search methods.

Table VI highlights the advantages and disadvantages of naturally inspired QI meta-heuristics optimization approaches. It offers key insights into their strengths and limitations in diverse scenarios.

#### B. Quantum-Inspired Machine Learning

Classical ML techniques enable computers to learn from data and make autonomous decisions or predictions using classical bits (0 and 1). Figs. 15 and 16 present the generic flowcharts of classical ML and QI-ML algorithms, respectively. Classical ML relies on traditional data processing and model training, while OI-ML utilizes superposition for enriched data processing, enables mapping of higher-dimensional features, and improves database exploration and parameter tuning through QI optimization techniques. These techniques are categorized into supervised learning, which leverages labeled data for predictions; unsupervised learning, which uncovers patterns in unlabeled data; and reinforcement learning, where an agent maximizes rewards through trial and error interactions with its environment and various other methods. The QI-ML leverages principles of quantum mechanics like superposition and entanglement to enhance computational efficiency and tackle complex optimization problems across various learning paradigms. Although designed for classical computers, QI-ML can aid in solving large-scale optimization problems. In particular, the QI-ML can provide superior resource optimization solutions by efficiently handling large datasets, faster convergence, and reduced complexity.

Furthermore, OML requires quantum devices and fundamentally differs from QI-ML. In this context, the authors of [90] provided a feasibility analysis of integrating quantum resources with classical ML. They discussed issues using deep learning techniques in 6G networks under stringent latency requirements. There are various QI-ML techniques in literature [91]-[93], which can guide the transformation of different classical ML algorithms into their QI counterparts, including deep learning [94], federated learning, generative adversarial networks [95], neural networks (NN), reinforcement learning, and support vector machines. The QI conceptualization of ML techniques can enhance managing complex data and generalization. This section examines QI-ML approaches from the filtered sub-set that address radio resource optimization, highlighting core principles and distinct strategies.

**i.** *Quantum-Inspired Generative Artificial Intelligence:* Generative AI synthesizes new data by learning patterns from existing data through audio, images, and text. Generative adversarial networks are a specific type of generative AI that produces realistic outputs using two competing components: the generator and the discriminator. The synergy between generative AI and quantum computing networks can facilitate the creation of novel algorithms to solve complex optimization problems. In this context, the authors of [97] comprehensively reviewed quantum generative adversarial networks and emphasized their potential by leveraging the unique properties of quantum computing for advanced generative modeling.

Similarly, an intelligent resource allocation framework in uncertain fidelity was devised in [98] to manage quantum resources such as qubits, quantum channels, and entangled pairs. In [99], generative learning enhancements were explored with quantum models like quantum circuit-born and tensor



Fig. 15. General flow of a classical ML algorithm [89].

Fig. 16. General flow of a QI-ML algorithm [96].

network-born machines, which leverage quantum mechanics to capture data correlations more efficiently than classical models, potentially enhancing generative task performance. The authors of [100] highlighted that scaling up quantum systems will increase errors, and generative AI can help devise robust error correction schemes. In [101], generative AI was used in social media analysis and interactive learning to generate images from language inputs and evaluate content relevance via fuzzy similarity scores. QI generative models could enhance this by enabling advanced pattern recognition and data synthesis through quantum computational advantages.

**ii.** Quantum Federated Learning: It is a decentralized approach that enables multiple clients to train a shared model

while maintaining data privacy collaboratively. Each client processes its data and sends updates to a central server, aggregating them to improve the global model. Clients evaluate a variational quantum algorithm, accommodating varying data distributions and enhancing the model's robustness and accuracy in classification, combinatorial optimization, and quantum chemistry. In [102], a digital twin-assisted quantum federated learning algorithm was proposed for real-time training of personalized NNs across hospitals to maintain data privacy and enhance accuracy. Similarly, a flexible quantum federated learning method was developed specifically for satellite-ground communications in [103], achieving greater communication efficiency and computational gains compared to classical federated learning.

20

Furthermore, [104] highlighted the difficulty in copying quantum data and emphasized the necessity of a collaborative framework to aggregate data of different quantum machines. They proposed a robust quantum federated learning framework that performed well with a significant proportion of noisy data. Similarly, the quantum federated learning framework proposed in [105] effectively manages non-independent and identically distributed data, addressing challenges like unbalanced training frequencies through efficient communication protocols. Local models generate predictions, and the server creates a shared model that optimizes each local model. Model gradients are encrypted before transmission to ensure data privacy. Clients then update their local parameters and repeat the training process until convergence. Regarding benchmarks, quantum federated learning demonstrates significant advantages over classical centralized methods, especially in handling large, privacy-sensitive datasets. Additionally, kernel encoding methods for quantum data enhance data representation efficiency by requiring fewer qubits than classical techniques.

iii. Quantum Neural Networks (QNN): It utilizes qubits and quantum gates, which enable superposition and entanglement. Hence, multiple states can be explored simultaneously, improving optimization efficiency and introducing parallelism. In contrast, the classical NN deals with deterministic states, and its parallelism is constrained by the capabilities of classical hardware [106]. In [107], a hybrid quantum deep NN was proposed to enhance indoor localization accuracy through fingerprinting. The model incorporated gradient descent for optimization and could handle complex data while training on a small dataset with an accelerated training procedure. The scheme considered standalone measurements and a fusion of received signal strength and time of flight datasets. The hybrid quantum deep NN was evaluated in a real-time environment, and it achieved the lowest localization error compared to different benchmarks, such as Euclidean-based K-nearest neighbors, classical deep NN, and classical deep NN with dropout. QNNs are frequently employed in various scenarios, including communication [108], clustering [109], [110], security [111], [112], and user experience [113].

**iv.** *Quantum-Inspired Support Vector Machine:* It is a supervised learning algorithm for classification and regression that finds the optimal hyperplane to separate data points within a high-dimensional space. In this context, the authors of [114] proposed a QI counterpart of support vector machine that incorporates an improved fast sampling technique to sample the kernel matrix efficiently. The labeled data is transformed into an arborescent data structure, allowing for logarithmic time sampling of vectors. This enables classification with an arbitrary success probability in logarithmic time relative to the data dimension and the number of data points. The analysis showed the potential of an exponential speedup in classification as compared to the classical support vector machine.

v. *Quantum-Inspired Reinforcement Learning:* Classically, RL is modeled as a Markov decision process, which involves an agent learning to make decisions through interaction with

its environment. The agent aims to maximize cumulative rewards by exploring and exploiting actions based on observed states through trial and error. However, classical methods face challenges in exploring large state-action spaces due to the curse of dimensionality. On the contrary, the QI-RL enhances learning efficiency through quantum parallelism. In [116], a QI-RL approach was proposed for optimized wireless VR streaming within industrial IoT systems. It involved initializing quantum actions, executing observed actions, updating O-values based on rewards, and applying Grover's iteration to amplify successful actions. The proposed framework significantly enhanced average stalling rate and energy consumption compared to benchmarks such as static scheme, joint optimization without content correlation, and greedy offloading with random allocation. Furthermore, the QI-RL algorithm showed faster convergence than the Q-learning algorithm by exploiting quantum search instead of using an unstructured exploration and exploitation. Due to their improved efficiency and enhanced learning capabilities, QI-RL has been applied frequently in diversified scenarios, including IoV [117]-[120], [122], smart trains [121], UAV path planning [123], [126], wireless sensor networks [115] and wireless VR [124].

vi. Quantum-Inspired Residual Tensor Train: It is conventionally used for tensor decomposition. It enables efficient representation and compression of high-dimensional data through multilinear residual information. This approach enhances computational efficiency in ML tasks. However, the modeling of multilinear correlations is considered challenging due to computational and memory requirements. The tensor train was initially developed for many-body quantum systems to create multilinear models by embedding input features into a high-dimensional Hilbert space.

The authors of [125] proposed a robust QI residual tensor train model incorporating skip connections to learn multiple multilinear feature correlations. The proposed model was applied to practical examples with limited data, such as average localization error and Boston housing datasets. The average localization error dataset was used as an application in wireless sensor networks. It comprised four features: anchor ratio, iteration count, node density, and sensor transmission range. The QI residual train showed superior performance in terms of R2 score and root mean square error compared to different benchmark schemes, including linear regression, polynomial regression, ridge regression, polynomial ridge regression, lasso regression, polynomial lasso regression, support vector regression, and multilayer perception. The QI approach showed a higher time and memory complexity due to increased parameters for additional connections. It showed immunity to exploding and vanishing gradient problems observed in the classical tensor train models.

Table VII summarizes the advantages and disadvantages of ML-based QI optimization approaches. It provides valuable insights into their strengths and limitations across diverse scenarios, facilitating the assessment of their suitability for different optimization tasks.

 TABLE VII

 Key insights into the machine learning-based QI optimization techniques.

QI Technique	Ref.	Remarks
		• Quantum computing enhances generative AI model training.
QI Generative AI	[100], [101]	• It accelerates data preprocessing.
		• Quantum computational intelligence improves human-machine interaction.
Quantum Federated	[102]	• Enables accurate training without local data while preserving privacy.
Learning	[102]	• Robust against single and mixed quantum noise.
Quantum Neural	[106] [108]	• It has a lower complexity as compared to conventional method.
Networks	[112]	• Enables parallel training, simultaneously updating the weights.
INCLWOIKS	[115]	• Significantly enhances the accuracy of indoor localization.
		• The approach does not require quantum resources.
QI Support Vector	[114]	• Sampling technique enables exponential speedups.
Machines	[114]	• Enables arbitrary success probability in logarithmic time for data dimensions.
		• Performance should be explored on high-rank datasets without approximations.
		• Quantum search enables faster learning and convergence than classical method.
OI Reinforcement		• Adapts rapidly to highly dynamic and complex environments such as IoV.
Learning	[115]–[124]	• Coordinated unitary transformation allows concurrent state updates.
Learning		• Superior rewards with reduced delay, energy consumption and CPU cycles.
		• Optimizes relay and transmit power for high throughput and energy efficiency.
OI Residual Tensor		• Ensures reliable performance on problems with limited data.
Train	[125]	• Solves vanishing and exploding gradient issues seen in classical counterparts.
11/2011		• Capable to build robust classification boundary in high-dimensional space.

#### C. Quantum-Inspired Specialized Approaches:

These methodologies also leverage quantum mechanics principles to provide specialized solutions for specific tasks across various domains. This subsection presents an overview of underlying approaches identified in the filtered literature of QI optimization and highlights their distinctive strategies to address complex optimization problems.

i. Quantum-Inspired Game Theory: Classically, game theory provides an analytical framework to analyze competitive situations where the outcomes depend on the actions of multiple players. Key elements include players, strategies, payoffs, and Nash equilibria, where no player can benefit from unilaterally changing their strategy given the strategies of others. The authors of [127] analyzed the quantum Colonel Blotto game, where players allocate resources across multiple battlefields to maximize control. They modeled jamming in quantum communication networks due to the susceptibility to interference and jamming, which can disrupt communication. The players utilized quantum strategies, such as superposition and phase manipulation, to enhance tactical options compared to classical strategies. The quantum model enhanced classical methods by enabling jammers to add amplitude and phase noise. Additionally, the complexity of determining Nash equilibria increased significantly with the quantum approach as the number of potential strategies increased exponentially.

**ii.** *Grover Adaptive Search:* It utilizes classical probabilistic and heuristic methods inspired by the principles of Grover's algorithm; however, it does not achieve the quadratic speedup that Grover's quantum algorithm provides. The original Grover's search algorithm employs qubits and quantum gates

to amplify amplitude, resulting in a quadratic speedup for finding a target in an unstructured database. Grover's adaptive search algorithm can achieve significant speedup for quadratic and higher-order unconstrained binary optimization problems, reaching a query complexity of  $O(\sqrt{2^{n_b}})$  compared to the classical exhaustive search, which requires  $O(2^{n_b})$  operations, where  $n_b$  refers to the number of binary variables [128]. The Grover quantum search was applied iteratively in [129] to identify transmitted codewords swiftly.

**iii.** *Quantum Variational Algorithms:* These algorithms combine classical and quantum computing resources to solve complex optimization problems. They include variational quantum eigensolvers, which minimize the Hamiltonian's expected value; quantum approximate optimization algorithms, which address combinatorial optimization iteratively; variational quantum classifiers, which classify patterns using parameterized circuits; and hybrid variational algorithms, which tackle multi-objective problems. Each algorithm leverages quantum circuits and classical optimizers to address computationally demanding tasks. Fig. 17 presents the flowchart of generic quantum variational algorithms, highlighting an iterative process of parameter optimization and loss evaluation to refine a quantum model until convergence.

In [132], the quantum approximate optimization algorithm was employed to solve the NP-hard combinatorial optimization problem of maximum likelihood detection. The algorithm encoded the problem into a Hamiltonian framework and used parametrized quantum circuits and variational parameters. The quantum adiabatic evolution aided in preparing the initial state and achieving the ground state of Hamiltonian. The complexity of the quantum approximate optimization algorithm depends



Fig. 17. General flow of a quantum variational algorithm [130].

upon the number of quantum gates, and it showed comparable performance to classical maximum likelihood and minimum mean squared error detectors in terms of bit error rate. Similarly, variational quantum eigensolver with constraints Table VIII summarizes the advantages and disadvantages of specialized QI optimization approaches. It provides valuable insights into their strengths and limitations.

#### D. Quantum-Inspired Optimization: Techniques and Insights

The QI approaches have been applied to a multitude of domains and provide advantages such as lower complexity and faster convergence [134], [135], necessary to meet the demands of 6G networks. This section discussed the key QI techniques in the filtered dataset of papers that address the radio resource optimization issues, explains different QI optimization approaches to tackle complex optimization problems. For example, the digital annealer can solve complex combinatorial problems with large datasets [136]. The hybrid quantum-classical computing frameworks can alleviate challenges in resource optimization [137], and Gurobi's mixed integer programming models effectively solve complex scheduling problems [138]. The power network optimization and reconfiguration are also considered through quantumclassical hybrid approaches [139] and [140]. Hybrid classicalquantum optimization: Similarly, [141] also solves a QUBO problem by employing a hybrid classical-quantum approach that outperformed the classical simulated annealing method. The semiconducting magnetic energy storage systems can be used for global search in optimization problems by employing the modified QPSO technique [142].

Table IX provides an overview of QI techniques, detailing a specific focus on radio resource optimization and the frequency with which each technique is applied to various problem types. Notably, QI techniques based on generative AI, support vector machines, residual tensor train, and game theory have not been utilized for radio resource optimization tasks. In contrast, the QPSO and QI-RL algorithms have seen frequent application, with both used 10 times each. Figs. 18 and 19 show a breakdown of the annual usage of QPSO and QI-RL, respectively. QPSO has been applied in diverse scenarios including the optimization of channel assignment, edge devices, spectrum sensing, localization, and UAV-assisted networks. QI-RL has been utilized for optimizing edge devices, UAV-assisted networks, and power control.

TABLE VIII Key insights into the specialized QI optimization techniques.

QI Technique	Ref.	Remarks
QI Game Theory	[127]	• Quantum strategies efficiently counter jamming with effective mechanisms.
		• Complexity grows exponentially with servers due to the "Curse of Dimensionality."
Grover Adaptive Search	[126], [128], [131]	<ul> <li>Better balancing of exploration and exploitation as compared to classical counterparts.</li> <li>Quadratic speedups in solving higher-order unconstrained binary optimization tasks.</li> <li>Significantly lower number of qubits and query complexity.</li> <li>Higher number of quantum gates required.</li> </ul>
Quantum Approximate Optimization	[132]	<ul><li>Scalable to solve combinatorial problems.</li><li>High memory and computational power requirements.</li></ul>

#### TABLE IX

The details of different QI approaches used in literature to solve the radio resource management problems. The use cases are categorized according to Section V. A: Channel Assignment, B: Edge Devices, C: Reconfigurable Intelligent Surfaces, D: Spectrum Sensing and Localization, E: UAV-assisted Networks, and F: Power Control.

Quantum-Inspired	References	Rad	Radio Resource Optimization Focus				Frequency	
<b>Optimization Techniques</b>		A	В	C	D	E	F	
Quantum-Inspired Meta-Heuristics								
QI Ant Colony Optimization	[58]		<ul> <li>✓</li> </ul>					1
Quantum Clone Whale Optimization	[60]		1					1
QI Evolutionary Algorithm	[61], [62]		1				<ul> <li>✓</li> </ul>	2
QI Genetic Algorithm	[64]		1					1
Quantum Particle Swarm Optimization	[65]–[74]	1	1		<ul> <li>✓</li> </ul>	1		10
QI Seagull Optimization	[76]					1		1
QI Termite Colony Optimization	[77]						<ul> <li>✓</li> </ul>	1
QI Backtracking Search	[78]						1	1
QI Bald Eagle Search	[79]			1				1
QI Gravitational Search	[80]		1					1
Quantum Annealing	[81], [82], [85]	1	1	1				3
QI Social-Emotional Optimization	[87]	1						1
QI Tabu Search	[88]				1			1
	Quantum-Inspired M	<b>Aachin</b>	e Learr	ning				
QI Generative AI	[100], [101]							0
Quantum Federated Learning	[102]		1					1
Quantum Neural Networks	[106]–[108], [113]	1	1		<ul> <li>✓</li> </ul>		1	4
QI Support Vector Machines	[114]							0
QI Reinforcement Learning	[115]–[124]		1			1	1	10
QI Residual Tensor Train	[125]							0
Q	uantum-Inspired Spe	cialized	l Appro	oaches				
QI Game Theory	[127]							0
Grover Adaptive Search	[126], [128], [131]	1	1		1			3
Quantum Approximate Optimization	[132]	1						1



Fig. 18. Frequency of QPSO algorithm over the years.



Fig. 19. Frequency of QI-RL over the years.

### V. QUANTUM-INSPIRED RADIO RESOURCE OPTIMIZATION IN 6G NETWORKS

This section comprehensively explores the QI techniques applied to radio resource optimization within the context of 6G networks, systematically categorizing the existing body of research. Fig. 20 illustrates the categorization of different aspects of communication networks where the QI approaches of the filtered dataset have been applied. These include channel assignment, edge devices, power allocation, RIS, spectrum sensing and localization, and UAV-assisted networks. The papers of each category are reviewed to extract the objectives, constraints, solution techniques, and key contributions inherent



Fig. 20. Application areas of QI optimization in communication networks.

in the respective studies. Additionally, this section rigorously identifies existing research gaps and synthesizes the lessons learnt, thereby offering a thorough understanding and delineating future research trajectories in this evolving domain. Fig. 21 illustrates a comprehensive communication system with vertical segments, where each aspect requires distinct resource optimization solutions depending on the identified categories of radio resource optimization.

#### A. Quantum-Inspired Channel Assignment

Channel assignment refers to allocating frequency bands or time slots to devices in the communication network. It facilitates efficient and interference-free communication while optimizing the limited spectrum. Fig. 22 depicts a wireless networking scenario where diverse edge devices, such as mobile users, IoT sensors, UAVs, and smart computing elements, connect to base stations. It emphasizes the need for an efficient channel assignment scheme to meet heterogeneous requirements of these devices, ensuring low latency, high reliability, and massive connectivity in 6G networks.

The QI methods can enhance classical channel assignment solutions by drawing inspiration from quantum computing concepts, such as superposition, entanglement, and quantum parallelism. The concurrent exploration of broader solution spaces can yield optimal channel assignments. QI algorithms such as quantum annealing, genetic algorithms, quantum walks, and ML adaptations can offer novel solutions to complex dynamic communication networks. Leveraging QI methods on classical computers can augment the effectiveness of channel assignment algorithms and offers a viable path forward, although practical quantum computation is in its early stages. Channel assignment techniques require tailoring to address the unique challenges that quantum communication networks pose, including dealing with the quantum nature of information and the corresponding security requirements. This subsection examines papers that address channel assignment optimization using QI techniques.

Considering the inherent limitation of radio spectrum resources, the increasing number of user terminals and the surging demand for mobile traffic necessitates optimization of



Fig. 21. Different segments of the communication network requiring resource management.



Fig. 22. Wireless network scenario requiring optimal channel assignment.

wireless channels. The wireless channel assignment is an NPhard problem conventionally solved through the one-hot encoding of channel indices. The authors of [128] formulated this problem as a higher-order unconstrained binary optimization problem by considering the binary ascending and descending encoding of channel indices and solved it through the Grover adaptive search algorithm with a quadratic speedup in identifying the solution. The objective was to minimize interference between access points. The constraints for wireless channel assignment encompass two key aspects. Firstly, each access point is restricted to utilizing only a single channel, ensuring no channel sharing among access points. Secondly, it is crucial to avoid the usage of nonexistent channels, thereby limiting the assignment to available and valid channels. The analysis showed that the proposed approach employing descending encoding requires a significantly lower number of qubits and query complexity.

Similarly, efficient resource optimization in a heterogeneous networking scenario is necessary due to the varying quality of service requirements imposed by different operators. In this regard, the authors of [66] proposed a quality of service-aware virtualization resource optimization mechanism for converged network architecture and employed the QPSO algorithm to solve the global optimization problem. The objective was to minimize tuning overhead and the load balancing ratio. The constraints included ensuring that the traffic load on each wavelength does not exceed its maximum capacity, assigning at least one wavelength to each virtualization heterogeneous cloud radio access network, maintaining the unchanged state of optical network units within each heterogeneous cloud radio access network, and ensuring that each activated wavelength was assigned at least one optical network unit. A penalty function was used to transform the constrained optimization problem into an unconstrained one. The benchmarks considered in this paper include order acceptance and scheduling, and workflow scheduling algorithm.

In [132], the maximum likelihood detection of binary symbols over a MIMO channel was encoded into the Hamiltonian operator. The maximum likelihood detection task was transformed into a multi-level quantum approximate optimization algorithm circuit, and its analytical expression was derived. It was a combinatorial NP-hard optimization problem of selecting the most likely transmitted signal from the dataset. The performance of the proposed quantum approximate optimization algorithm-based maximum likelihood detector depended on the graph's structure associated with the Hamiltonian function. The benchmarks used for comparison included classical and minimum mean square error detectors.

In [81], the authors transformed the wavelength assignment task into a graph coloring problem and solved it through the QUBO algorithm by identifying the spins that minimize energy. The coherent Ising machine solver was used to solve the combinatorial NP-complete optimization problem, and Lagrange multipliers reduce hyperparameters. The two objectives of the proposed technique included reducing the number of QUBO problems to be solved and introducing robustness against finding suboptimal solutions. The constraints entailed assigning each vertex to only one wavelength and ensuring that two adjacent vertices are not assigned the same wavelength. The benchmarks considered for comparison included Gurobi optimization software, GNU linear programming kit-mixed integer programming solver, largest degree first, and original QUBO. Experimentation was performed on a synthetically generated dataset comprising 900 randomly generated graphs, with nodes ranging between 10 and 100.

The authors in [87] devised an optimized power control technique for the massive MIMO uplink using a QI socialemotional optimization algorithm. It developed on assuming a single-cell massive MIMO system with BS having hardware impairments. The objective was to perform antenna selection effectively and solve the multi-constrained non-convex power allocation problem that maximizes energy and spectral efficiency. There were constraints on the power and data rate of each user. The proposed technique leveraged quantum evolution and social-emotional optimization to achieve strong searching ability and fast convergence. The complexity of the proposed approach was  $O(i(5P_s + 3P_sn))$  where  $i, P_s$ , and n refer to iterations, population size, and maximum dimensional number, respectively. The benchmarks used for comparison included social-emotional optimization, fractional programming, and maximum power allocation.

UAV-assisted cellular networks were considered for adaptive bit-rate video streaming by performing the joint optimization of caching and user association in [61]. The proposed QI evolutionary algorithm was used iteratively to solve the formulated NP-hard problem while aiming to minimize the total content delivery delay. Several constraints included the total size of video chunks cached at each UAV being lower than the UAV's caching capacity, each user associating with only one UAV, and the optimization variables being binary. The benchmark algorithms used for comparison included solely QI evolutionary algorithm-based techniques for channel assignment and user association, as well as random channel assignment and user association techniques.

Cell-free MIMO systems can aid in coverage extension without cell boundaries. Therefore, authors of [108] proposed a quantum NN-based algorithm for optimized assignment of transmitter-user in cell-free MIMO systems. The objective was to enhance the minimum achievable sum rate among users with a fair distribution of resources. There was a constraint on the precoding vector, which enables optimized transmission of signals and manages interference. A Rayleigh channel with

 TABLE X

 CHANNEL ASSIGNMENT USING QI OPTIMIZATION TECHNIQUES.

Ref.	Objective	Constraints	Problem	Solution	Remarks
[61]	Minimize total content delivery delay	Caching capacity and binary optimization variables	NP-hard, non-linear integer pro- gramming	QI evolutionary algorithm	The content delivery delay minimization problem is formulated as a non-linear integer programming optimization problem, and QI evolutionary algorithm is proposed to obtain optimal caching and user association.
[66]	Minimize tuning overhead and load balancing ratio	Traffic load does not exceed wavelength's capacity, and virtualization heterogeneous cloud radio access network is assigned at least one wavelength	Mathematics model	QPSO and quality of service mapping algorithm	The paper formulated the wavelength assign- ment problem as a mathematical optimization model and solved it using a QPSO algorithm. It also proposed a quality-of-service mapping algorithm to allocate 5G traffic to priority queues in a time and wavelength division multiplexed passive optical network. It intro- duced a resource scheduling algorithm for the downstream traffic.
[81]	Reduce count of QUBO problems and avoid suboptimal solutions	Each vertex is assigned to only one wavelength and two adjacent vertices are not assigned the same wavelength	NP-hard	QUBO, QI annealing	In this paper the NP-hard, routing and wave- length assignment problem was solved by simulating coherent lising machine-based QI optimization solver.
[108]	Maximize the minimum achievable sum rate	Precoding vector	Transmitter- user assignment	Quantum NN	The paper proposed a quantum NN for fair allocation of communication resources.
[128]	Optimal channel assignment	Communication resource	NP-hard	Grover adaptive search algorithm	The NP-hard problem is formulated as a hybrid unconstrained optimization problem by considering channel indices as binary numbers, and the specific quantum circuit is created, while deriving exact number of qubits and quantum gates required by Grover adaptive search.
[132]	Selecting most likely transmitted signal from the dataset	Number of transmit symbols from dis- crete set	NP-hard, Maximum likelihood detection	Multi-level quantum approximate optimization algorithm	The paper aimed at maximum likelihood de- tection of binary symbols transmitted over MIMO channel.
[87]	Maximize energy and spectral efficiency	Power and data rate of each user	Non-convex	QI social- emotional optimization	The paper aimed to jointly optimize antenna selection, power allocation, and QoS con- straints for energy-efficient massive MIMO uplink networks through QI social-emotional optimization algorithm.

variations in mean value was considered for the performance of simulations between the average sum rate and transmission signal to noise ratio. The complexity of the proposed scheme was i.r, where i and r represent the number of iterations and single gradient calculation, respectively.

Table X illustrates the details of the reviewed papers, including problem objective identification, considered constraints, type of problem, solution approach, and remarks related to the contribution of research work.

Lessons Learnt: The comprehensive literature review of

the dataset has aided in identifying several significant gaps, paving the way for future research in channel assignment techniques. Various techniques for advancing research on radio resource optimization concerning channel assignment can be summarized as follows:

- *Minimizing Quantum Gates:* Efforts to minimize the number of quantum gates are identified as a future direction, mainly due to the increased quantum gates requirement outlined in [128].
- Energy Efficiency in Connectivity: Providing high en-

ergy efficiency in scenarios with massive connectivity poses a significant challenge, as highlighted in [66].

- *Heterogeneous Computing:* The proposed technique can undergo revision and evaluation to enhance applicability in heterogeneous computing scenarios, including multi-core central and graphics processing units [81].
- *From Single-Cell to Multi-Cell:* There is potential for extension of the proposed single-cell technique to more complex environments such as massive MIMO with imperfect channel state information in multi-cell settings, integrating cognitive radio and co-time co-frequency fully duplex systems [87]. Furthermore, the joint analysis of UAV deployment, resource optimization, content caching, and user association for adaptive bit-rate video streaming can be performed [61].

#### B. Quantum-Inspired Optimization for Edge Devices

Fig. 23 depicts various edge devices, including base stations, coverage extension drone, IoT sensors, IoV and mobile users. It emphasizes the need for efficient radio resource optimization and task offloading to overcome computational power limitations in edge platforms and ensure seamless performance in interconnected environments. Therefore, efficiently managing constrained resources such as bandwidth, computational power, energy, and navigation is essential. However, optimizing resources in these networks presents numerous challenges, including complexity stemming from bandwidth optimization, complex data, device heterogeneity, edge and fog computing, limited energy resources, mobility in IoV, varied quality of service requirements, and security and privacy concerns. The QI optimization techniques offer a promising approach by exploiting entanglement, parallelism, and superposition to explore multiple potential solutions concurrently, thus enhancing systematic performance in terms of efficiency and reliability. These techniques are adaptable to dynamic environments, making them favorable for optimizing resources in IoT and IoV systems with time-varying conditions. This subsection provides a review of the papers focusing on the QI IoT and IoV resource optimization.

Swarm intelligence and evolutionary algorithms are commonly employed for NP-hard task scheduling problems. The complex network topologies and diverse applications make task scheduling in cooperative environments challenging. The authors in [68] formulated the task device-edge-cloud cooperative scheduling problem through mixed integer nonlinear programming and proposed a QPSO based approach to identify the global best solution. The objective was to maximize the number of tasks while employing constraints on deadline and resource capacity. The proposed technique was compared with benchmark heuristic methods such as earliest finish time first, least average completion time, least slack time first, PSO, genetic algorithm, and variants of the QPSO algorithm. Performance metrics used in the paper included the number of finished tasks, computing length completion, input data size processed, computing resource utilization, cost efficiency, and time consumed in decision-making.

The constrained resources of mobile edge computing devices can be a bottleneck in delivering high-quality services to users, and the mobility and network state fluctuations can impose adverse effects on service quality. Hence, [67] presented a service migration scheme from initial nodes to other edge nodes while minimizing delay and energy consumption. The problem was formulated as a binary programming problem, and a particle swarm-based service migration scheme was proposed to provide better services. The proposed scheme developed a queuing delay prediction algorithm, a delay-aware computation resource allocation algorithm, and a modified QPSO. The queuing delay prediction algorithm identified average and individual task queuing delays, the delay-aware computation resource allocation algorithm performed dynamic allocation of computational resources, and the modified QPSO performed an efficient search that avoids local extreme point constraints. The motion of users was simulated through random walks. The system performance was evaluated before and after service migration for enhanced mobile broadband and URLLC tasks for the proposed and benchmark techniques, such as QPSO and the greedy algorithm [67].

A mixed integer non-linear programming problem for resource optimization was formulated in [65] to maximize energy efficiency. The constrained optimization problem was transformed into an unconstrained problem through the penalty function. The suboptimal solution was obtained using the QPSO algorithm. Several constraints include the offloaded task completion time of different users not exceeding the total computation delay, the energy consumption of different users being less than the energy harvested, transmitting power having an upper bound, and a distance decision for energy harvesting. The benchmarks used for comparison include localedge-auxiliary-transfer, local-edge-auxiliary, local-edge, localauxiliary, and local-only. However, this technique considered single relay unit for the transfer of computational tasks.

The relay and transmission power control for wireless sensors in IoT is crucial due to the continuous reduction in sensor size. The authors in [115] proposed QRL-based joint optimization of relay and transmission power selection using Grover's iterative method. The objective was to maximize the throughput of each link in the system, subjecting to minimized energy consumption. The system model incorporated constraints on transmission power in each time slot, energy flow, data rate, data buffer overflow, and boundary conditions on allocated power. The relay and transmission power selection has been modeled as a Markov decision process, and ORL was applied for optimization. The channels were assumed to be quasi-static block Rayleigh fading where the channel gain remains constant within one timeslot but varies over different timeslots. The benchmark schemes used for comparison include a deep RL-based relay selection scheme, a quantumenabled RL framework for joint optimal relay and transmit power selection, and random selection. The performance metrics included throughput, energy consumption, and utility.

Air-borne nanotechnology-based sensors operating over terahertz bands in vehicular networks necessitate optimizing energy efficiency and wireless power transfer due to battery constraints. Quantum-behaved and improved discrete particle swarm-based algorithms were proposed in [69] to maximize energy efficiency. The non-convex mixed integer non-linear



Fig. 23. Wireless network requiring optimal task offloading for edge devices.

programming optimization problem was converted into a series of energy efficiency resource optimization tasks over different time slots using the equivalent transformation method. The constraints included the orthogonality of terahertz channels and power. The benchmarks used for comparison include quantum-behaved particle swarm-based energy efficiency optimization and PSO. The performance metrics included energy efficiency, convergence time, and effective constraint handling.

Addressing battery size constraints and growing computational demands on IoT sensors complicates extending the network lifetime. Therefore, [58] proposed an undirected graph-based energy optimization model for IoT environments, where sensor nodes and their wireless links were represented as vertices and edges, respectively. This NP-hard QI green communication framework aimed to minimize the energy cost of relaying paths from a source node, subject to qubit state probability constraints. The qubit population generated binary solutions with a repair technique for constraint violations. Performance benchmarks included quantum-based and classical ant colony optimization methods, focusing on quantum representation, measurement, and rotation angle. The performance metrics included network lifetime, node energy consumption, and average node residual energy.

Furthermore, IoT devices are crucial in achieving maximum automation and enhancing efficiency. However, the reliability of the IoT environment can be compromised by data errors, missing segments, and falsified perceptions. Therefore, the authors in [143] proposed a novel quantum computing-inspired IoT optimization technique to maximize the data accuracy by acquiring the data in quantum form. Two parameters, such as sensors in the vicinity and optimal sensor space, were used to identify IoT-sensor space. The aim was to determine optimal sensor space that maximizes overall stato-dynamic sensing ability. This technique eliminated distant nodes prone to falsified data generation. The constraint on sensors in the vicinity reduced the sensor space for every object. The conventional optimization task was non-polynomial with an asymptomatic complexity of  $O(2^L)$ , where L is the number of qubits. The quantum formalization reduced this optimization problem into a bounded-error quantum polynomial time class with running time complexity of O(L), searching time of O(X), and time complexity of  $\Omega((L^3 log(1/\epsilon))/log(L))$ , where X and  $\epsilon$  are complex matrix of size  $2L \times 2L$  and computing error, respectively. Particularly, the search time complexity was reduced from  $O(2^X)$  to O(X) by incorporating quantum computing. The proposed technique was evaluated through three key performance indicators: data accuracy, cost, and temporal efficiency. A total of 11 crucial parameters, encompassing range, span, accuracy, and others, were considered in a stochastic environment to ensure optimal real-time delivery of IoT services. The proposed technique was compared with genetic bio-inspired optimization, differential evolution, and PSO. The performance evaluation metrics included temporal efficiency, statistical analysis, and reliability.

The productivity of an industrial IoT environment can be improved through wireless VR, enabling workers to engage in industrial activities flexibly and remotely. In this context, [116] formulated a joint optimization problem considering task offloading, computing, and spectrum resource optimization by incorporating content correlation between VR equipment, channel variations, and VR quality of experience. Dual approximation was employed to transform the problem into a Markov decision process, and an online QI-RL algorithm was designed to identify optimal policy and apply quantum parallelism for dimensionality reduction. The objective was to reduce the energy consumption of VR devices and ensure a smooth experience. There were constraints on the spectrum, computing resources, edge computing-assisted access point computational capacity, buffer size, and quality of experience. The system model considered a fiber link between edge computing-assisted access points and VR service providers. The performance of the proposed scheme was evaluated in comparison to various baseline schemes such as static, joint optimization without content correlation, and greedy offloading with random allocation. The performance evaluation metrics included energy consumption and average stalling rate.

The emerging applications undoubtedly increase users' quality of experience in the IoV but also impose a significant computational load on resource-constrained vehicles. In [117], an edge intelligence-assisted model was designed that integrated WiFi-based mobile edge computing and 5G-based cloud computing. The communication and computation resource optimization scheme was formulated as a Markov decision process. The aim was to minimize average task processing latency. The proposed QI-RL method considered the resource optimization of an NP-hard non-convex problem as a quantum uncertainty problem. The aim was to minimize the longterm average time delay through joint optimization. Several constraints were considered, such as the mobile edge computing server's computational resources, BS's bandwidth, total task execution delay, transmission strategy, offloading strategy, energy consumption, and economic cost of resources. The convergence rate was minimized through quantum parallelism. The benchmark schemes used for comparison include local computing, random selection, Q-learning, QRL without mobile edge computing, and QRL without the cloud. The metrics used to evaluate performance include learning convergence and processing delay.

Furthermore, the heterogeneity characteristics must be considered to harness the benefits of network coding in terms of error control and throughput improvement. In [70], the problem of multi-user accessing tasks and a multi-layered service scheduling framework between users and control entities was formulated. A statistical fairness measurement model was constructed to ensure the end user's independence. The proposed scheduling scheme in a multi-user IoV environment was called dynamic resource optimization scheduling, where the scheduling service aimed to maximize the total utility of all the network coding sets. Furthermore, a coding cache queue control algorithm called QPSO and the proportional integral model collectively achieved the global optimal solution for active queue control. The constraints included service budget for the end user from the system and service execution time for the end user. The benchmark schemes used for comparison included minimum latency aggregation scheduling, balanced shortest path tree, round robin pairing and scheduling, distributed data aggregation scheduling algorithm, and fast data aggregation protocol. The key performance indices evaluated include fairness index, network throughput, packet loss rate, time slice allocation, service scheduling rate, average system response time, and average end-to-end delay.

Similarly, the large amount of data processing required in the IoV imposes a serious need for task offloading. Hence, the authors in [71] considered an NP-hard resource optimization problem of mobile edge computing-enabled IoV. The joint offloading QPSO was proposed to perform optimal task offloading and minimize the execution time. The crossover probability was defined to improve the searchability and convergence speed of the proposed scheme. The constraints included maximum tolerance time, continuously connected time, and system utility. Collaborative offloading was considered for remote cloud, edge servers, and vehicles. The proposed technique was compared with different benchmarks, such as classical PSO, local computing, vehicle-to-vehicle offloading, and vehicle-toinfrastructure offloading. The performance evaluation metrics included average energy consumption, average time delay, and normalized system utility.

The authors of [118] ensured energy-efficient intelligence retrieval through a non-fungible tokens-based green network intelligence scheme utilizing the orthogonal frequency division of multiple access links between vehicles and BS. The intelligent networking problem was formulated as a Markov decision process, where intelligence can be retrieved from BS, dedicated short-range communication, and BS relay. The constraints included bandwidth, payment decisions, acquisition mode decisions, obtaining intelligence from pre-stored nodes, acquiring each block only once, delay, and energy. The NPhard joint optimization problem was solved through the QI-RL algorithm while aiming to maximize the long-term average revenue. The benchmarks used for comparison included hypertext transfer protocol-based retrieval schemes with/without cache and interplanetary file system-based schemes with random techniques. The metrics for performance evaluation included average reward and delays for the number of blocks and vehicles.

Considering the time-sensitivity of decision-making in connected and autonomous vehicles, the authors in [119] formulated a distributed vehicles selection problem according to energy, communication, computing, and quantification of intelligence as a many-to-many matching game. Gale-Shapely algorithm with a linear complexity was employed for the solution. Furthermore, spectrum resource optimization was formulated as a Markov decision process, and QI-RL was employed to identify the optimal policy. The QI-RL technique provided a good trade-off between exploration and exploitation. The Grover iterations aided in accelerating convergence and improving unordered search. A collective learning technique was proposed considering the many-to-many matching game and the QI-RL algorithm. The cooperative participants in the many-to-many matching game and the QI-RL technique included master participating vehicles, slave participating vehicles, and trusted participating roadside units. The benchmarks used for comparison included many-to-many matching with a random allocation technique and a random technique. The performance evaluation metrics included average delay and energy consumption.

The authors of [120] formulated an NP-hard joint optimization problem for non-fungible token-based distributed intelligence learning through a discrete Markov decision process. The QI-RL algorithm was employed to achieve fast convergence while considering constraints such as bandwidth, delay, and energy; only pre-stored nodes could provide intelligence, and each block was allocated only once. The benchmarks used for comparison included hypertext transfer protocolbased retrieval without cache, hypertext transfer protocolbased retrieval with cache, and random interplanetary file system-based retrieval.

The IoV can be considered a fog computing paradigm operating in an uncertain environment, demanding timely and intelligent decision-making. In this regard, the authors of [131] proposed a quantum exploration-based multi-armed bandit technique for task offloading, aiming to minimize the expected unit cost. Suboptimal amplitude amplification was anticipated due to unresolved issues, such as the lack of a direct mapping between phase and probability amplitudes and the non-smooth nature of arbitrary cost estimates, especially in environments with incomplete feedback and variability. To address this, the proposed quantum learning-based exploration technique, called quantum amplitude amplification, modified the probability amplitudes of potential actions through Grover iterations before collapsing. These iterations entailed oracle queries and diffusion operations. Quantum measurement and superposition also leverage trade-offs between exploitation and exploration to enhance learning.

Furthermore, [121] proposed a QI risk-sensitive RL algorithm and train-edge-cloud collaborative computing technique for smart trains. This work considered the real-time computation of several intelligent tasks, such as driver fatigue and infrastructure fault detection. Hence, the objective was to minimize the task average processing latency of all computing tasks while incorporating the constraints on edge intelligence servers, such as risk, computing capacity, and maximum number of offloaded tasks. Two Q-value functions were employed to assess the cost and risk. Independent updates were performed, and the weighted sum of Q-value functions was utilized to decide the action. Hence, the aim was to jointly minimize the risk and maximize the edge intelligence server computing resource. The general Q-learning and RL-based  $\epsilon$ greedy algorithm were used as benchmarks for comparison.

The IoV is expected to cater to capacity concerns of multidimensional spaces comprising actuators, edge devices, and IoT. In [122], a multi-objective technique was proposed to enhance the sensor state's quantization ratio through its coverage range, efficiency, and node-centric metrics. The objective was task offloading while ensuring low latency and resource utilization efficiency through accurate node deployment. It was an NPhard problem involving constraints of offloading decision rate, resource potentiality, channel, qubit state-error probability, and fusion variance. A QI online node consolidation algorithm was developed for a time-sensitive measurement reinforcement system with global positioning system (GPS) enabled active nodes and computation serving and relaying hypernodes. Latency concerns were tackled by performing node localization alongside an angle-based node position analysis using the Tgate. The simulation setup considered three qubits and a controlled combination of T-gates and H-gates to aid in mapping classical states. The benchmark schemes used for comparison included deep quantum routing agent, node deployment using quantum theory, quantum approximate optimization algorithm, and QI green communication framework for energy balancing in sensor-enabled cyber-physical systems. The performance metrics included consolidation error rate, iteration completion time, node selection error rate, offloading reduction, and service reliability.

A smart healthcare framework relies on the Internet of medical things and requires real-time training, efficient learning, and strong privacy protections. The authors of [102] proposed a digital twin-assisted quantum federated learning algorithm designed to enhance the training of variational QNN in various healthcare applications. The robustness of the proposed scheme was demonstrated through its superior performance in the presence of different types of quantum noise, including amplitude damping, depolarization, bit flips and phase flips.

The adaptive bit-rate algorithms are considered necessary to maintain a suitable quality of experience for different users while avoiding network congestion. The adaptive bitrate techniques are classified into rate-based, buffer-based, and hybrid. Wei et al. have considered evolving quantum technology and proposed a OI dynamic adaptive streaming over hypertext transfer protocol technique in [82] to optimize the bit-rate of video streaming. The problem was formulated as a QUBO task considering the video quality, bit-rate change, and rebuffering events. The one-bit-rate constraint was considered in the formulation. A digital annealer was used to minimize the objective function. The proposed technique was evaluated for three scenarios: static, walking, and moving on a bus. BigBuckBunny video content was encoded into six different bit-rates and used for evaluation. The benchmarks used for comparison included rate-based, buffer-based, modelpredictive control, and RL.

The mobile Ad hoc networks have a highly dynamic topology requiring an optimized routing scheme. In this regard, the authors of [64] proposed a quantum-genetic-based optimal link-state routing protocol augmented with a Q-learning strategy to maximize the selection of multi-point relay sets in such highly mobile networks. They optimized the routing decisions while considering node energy and link reliability. The proposed technique achieved superior performance regarding network control overhead, successful packet delivery, and end-to-end delay compared to benchmarks such as neighborhood state self-adaptive update and classical and cartographyenhanced optimal link-state routing protocols.

The authors of [60] proposed a high-performance clustering protocol-based on a quantum clone whale optimization algorithm. The clustering of integrated energy system-based wireless sensor nodes is an NP-hard problem due to several involved factors in obtaining the solution. The proposed algorithm utilized random search behavior within the global scope to explore the solution space, and each qubit needed to be updated according to the optimal individual after clonal expansion. Quantum probability amplitude and clonal expansion guide individuals in the population toward an optimal solution. There were constraints on intra-cluster distance and BS distance factor. The single-hop routing was adopted within a cluster, and single-hop or multiple-hop routing was considered for the BS. The proposed technique demonstrated superior accuracy and stability, with slightly increased complexity compared to several benchmark clustering schemes. The performance metrics included the largest number of surviving nodes because of considering the remaining energy of nodes and distance within the cluster.

Similarly, efficient clustering in wireless sensor networks has various benefits. These include reduced communication overhead, maximization, and increased network lifetime. However, the cluster heads are potentially exposed to heavy loads, which can cause premature death of cluster head nodes in a cluster and consequently reduce the lifetime of the wireless sensor network. Hence, cluster head load reduction is necessary for energy efficiency enhancement in wireless sensor networks. The system model proposed in [72] considered N nodes being grouped into disjoint clusters through K-means clustering. An approach was devised to optimize the cluster head node's load by formulating two objective functions to characterize the optimal part of the load of the cluster head, which shall be assigned to each node in the cluster and minimize the energy consumption of the nodes. There was a constraint on the cluster head's load distribution. A quadratic loss penalty function was incorporated to convert the constrained optimization problem into an unconstrained one, which enabled the application of a bio-inspired algorithm for optimization. The QPSO algorithm and game theory solved the optimization problem while considering a multi-hop energy transfer technique based on strong magnetic resonant coupling. The proposed QPSO and game theory-based techniques were compared with K-means and Leach algorithms.

Similarly, the operational mode assignment to wireless sensor nodes in wireless sensor networks is an NP-hard problem. Therefore, a Quadri-valent QI gravitational search algorithm was proposed in [80], which considered a heterogeneous computation framework combining the parallel processing capabilities of the central and graphics processing units. The work focused on identifying a network topology that maximized the network lifetime while considering connectivity constraints. The proposed technique enhanced the execution time through the quantum NOT gate and paralleling on the graphics processing unit, where the quantum NOT gate performed the NOT operation on qubits. The benchmark schemes used for comparative analysis included the binary genetic algorithm, binary PSO, improved binary quadri-valent gravitational search algorithm, and QI gravitational search algorithm. The performance metrics included speed, accuracy, network lifespan, and BS energy consumption.

The authors [113] proposed an adaptive virtualization for quantum services using NNs to enable reliable network virtualization and service replication under uncertainty. The objective was to devise a robust system for enhanced resource utilization and minimal delays to achieve customer satisfaction. Linear NNs were used for both virtualization processes and quantum state verification. The proposed technique showed a superior response ratio, processing rate, latency, utilization, and failure. The benchmarks used for comparison included stability and establishment QPSO, energy and performance-efficient task scheduling algorithm, and quantum-aware compositional scheduling framework.

Table XI illustrates various key details of the reviewed papers, including problem objective identification, the considered constraints, the type of problem, the solution approach, and the remarks related to the limitation of the proposed technique or an enhancement for the enrichment of the research work.

**Lessons Learnt:** The QI techniques can be utilized to further the research of radio resource optimization relating to the IoT and IoV. The research gaps and prospective research directions for IoT and IoV optimization are identified from the reviewed papers and listed as follows:

• Communication in Cloud and Edge Computing: The

challenges in edge computing environments include excessive task processing delays and potential congestion stemming from centralized data centers [67]. Furthermore, the dynamic nature of pervasive edge computing presents obstacles to effective computation and communication [116]. To address these challenges, there is a growing emphasis on energy-efficient collaboration within integrated networks [118], along with incorporating task priority and load balancing into the proposed scheme [121]. Additionally, there is a need to evaluate the feasibility and performance of proposed schemes on a larger scale and in mobile integrated energy systembased wireless sensor nodes [60].

- Algorithmic Enhancements and Evaluation: The effectiveness of the QPSO technique can be thoroughly examined by evaluating its performance on large-scale problems and exploring its optimization capabilities [68]. Similarly, comparing the QPSO technique with heuristic approaches can yield valuable insights into its efficacy [65]. Moreover, evaluating the proposed techniques across various domains, including large-scale optimization tasks, can provide a comprehensive understanding of their applicability and scalability [72], [80].
- Analysis and Scale Evaluation: The analysis of average delays for a larger set of vehicles in existing research [117] can provide valuable insights into the applicability of the proposed approach in real-time vehicular networks. Furthermore, evaluating the proposed dynamic resource optimization scheduling scheme considering end user cooperation [70] can lead to a practical assessment of its effectiveness. Additionally, analysis of the proposed scheme through convergence rate analysis in lightweight and complex environments can enhance its adaptability and robustness [122].
- *Emerging Research Avenues:* The key areas for future research include big data processing, visualization, and integrating classical optimization with quantum learning [115]. Additionally, the effect of variations in terrain and environment can be explored for the terahertz band, aiming for enhanced communication [69]. Implementing energy-efficient collaborative learning and the trajectory of collaborative learning are considered future enhancements in [119], [120], respectively.

### C. Quantum-Inspired Optimization of Reconfigurable Intelligent Surfaces

The intelligent surfaces of tunable meta-materials can significantly enhance wireless communication through coverage extension, improving energy efficiency, mitigating interference, and dynamically adapting to time-varying environments. Fig. 24 illustrates a scenario where an intelligent reflective surface (IRS) mounted on an aerial platform ensures reliable user connectivity by overcoming radio signal blockage from a building, facilitating seamless communication with the base station. These surfaces present various challenges, including deployment, channel estimation, resource allocation, interference, energy efficiency, and scalability. Realizing the benefits

 TABLE XI

 Edge devices resource optimization using QI techniques.

Ref.	Objective	Constraints	Problem	Solution	Remarks
[58]	Optimize the energy	Probability of qubit states	NP-hard	QI green communication framework for energy balancing in sensor-enabled IoT systems, QI ant colony optimization	This paper devised a quantum computing- oriented solution to optimize the energy con- sumption in sensor-enabled IoT environments considering energy-centric solution represen- tation, measurement, and rotation angle.
[60]	Enhance the QoS of wireless sensor network	Intra-cluster distance and BS distance factor	NP-hard	Quantum clone whale optimization algorithm	The paper aimed at enhancing the lifetime of integrated energy system wireless sensor networks by employing a high-performance clustering protocol.
[65]	Maximize energy efficiency	Computation, energy, power, and distance	Mixed integer non-linear program- ming	QPSO	This work formulated the resource optimiza- tion problem as a mixed integer non-linear programming problem and proposed a sub- optimal solution using QPSO algorithm in a device-to-device-assisted edge computing sys- tem with hybrid energy harvesting.
[67]	Minimize delay and energy con- sumption	Local extreme point constraints	Binary pro- gramming	Particle swarm- based service migration scheme	The service migration problem was formu- lated as a non-linear binary programming optimization problem, and particle swarm- based service migration scheme was proposed while considering the queuing delay predic- tion, delay-aware computation resource opti- mization, and modified QPSO algorithms.
[68]	Maximize the number of tasks	Deadline and resource capacity	Mixed integer non-linear program- ming	QPSO	This work formulated a task scheduling prob- lem and proposed QPSO based method iden- tify the solution in reasonable time.
[69]	Maximize energy efficiency	Power and orthogonality of tetra-hertz channels	Non-convex mixed integer non-linear program- ming	QPSO	This paper formulated the long-term energy efficiency optimization problem for nano- empowered vehicular networks as mixed in- teger non-linear programming problems over time slots and proposed quantum-behaved par- ticle swarm-based energy efficiency optimiza- tion algorithm and improved discrete particle swarm-based energy efficiency optimization algorithm to obtain suboptimal solutions.
[70]	Maximize the total utility of all network coding sets	Service bud- get and execu- tion time	Statistical fairness measurement model	QPSO and pro- portional integral model	The paper proposed a dynamic resource op- timization algorithm to schedule vehicles in a multi-user IoV system fairly. The QPSO algorithm combined the proportional integral model to achieve the global optimal solution.
[71]	Optimize task offloading and minimize execution time	Tolerance time, continuously connected time and system utility	NP-hard	QPSO for joint offloading	A mobile edge computing-based joint offload- ing technique for vehicular networks was pro- posed that utilized the QPSO algorithm to min- imize the task delay and energy consumption through tasks offloading to service nodes and roadside units.

Ref.	Objective	Constraints	Problem	Solution	Remarks
[72]	Minimize node en- ergy consumption and assign clus- ter head's load to nodes	Cluster head load distribution	Quadratic uncon- strained optimization	QPSO and game theory	The authors devised a load dividing model for cluster heads in the wireless sensor net- work to achieve enhanced network lifetime.
[80]	Maximize network secrecy energy ef- ficiency	Connectivity	Operational mode assignment, NP-hard	Quadri- valent QI gravitational search algorithm	The NP-hard problem of operational mode selection for nodes in wireless sensor net- works was tackled by employing QI algo- rithm.
[82]	Optimize bit-rate of video streaming	Buffer size and one-bit- rate	QUBO	Quantum Annealing	This paper focused on adaptive bit-rate con- trol by increasing the average bit-rate and decreasing the rebuffering events for maxi- mal user quality of experience.
[116]	Reduce the energy consumption of VR equipment	Spectrum, computing resources, buffer size and quality of experience	Markov deci- sion process	QI-RL	This paper formulated a joint optimization problem that incorporates viewport render- ing offloading, computing, and spectrum re- source allocation and aimed to reduce the energy consumed by VR equipment while ensuring a smooth, immersive VR experi- ence. RL-based online learning algorithm integrated with quantum parallelism was em- ployed to find the optimal policy.
[117]	Minimize average task processing la- tency	Computation, energy and resource cost	Non-convex, NP-hard	QI-RL	This work formulated the uplink resource optimization problem in IoVs as a time- varying Markov decision process, and pro- posed QI-RL algorithm for edge offloading that allowed vehicles to choose network ac- cess mode and offloading strategy flexibly.
[118]	Maximize long- term average revenue	Bandwidth, payment and acquisition mode decision, pre- stored nodes intelligence attainment, one-time block acquisition, delay, and energy	Markov deci- sion process, NP-hard	QI-RL	A non-fungible token-based green intelli- gence networking scheme was proposed for connected and autonomous vehicles in smart cities enabling efficient networking of intelli- gence and reduce energy consumption. The core problem was formulated as a discrete Markov decision process and solved using QI-RL algorithm.
[119]	Maximize the sum incomes from all vehicles	Delay, energy, number of master participating vehicles a slave participating vehicle can assist, and master participating vehicle and matching decision value	Markov deci- sion process	Gale- Shapely Algorithm, QI-RL	This work proposed a quantum collective learning and matching game-based scheme for connected autonomous vehicles in the metaverse and used the simulated environ- ment of metaverse to expand the sample size and quantify intelligence diversity for effec- tive learning. The vehicle selection and spec- trum allocation problems were formulated as a matching game and Markov decision process, respectively, and solved using Gale- Shapley and QI-RL algorithms to achieve high system revenue.

Ref.	Objective	Constraints	Problem	Solution	Remarks
[120]	Maximize reward (utility of intelligence, acquisition delay, and bandwidth cost)	Bandwidth, delay, energy, only nodes that pre- stored can provide intelligence, and each block is allocated only once	Discrete Markov decision process, NP- hard	Grover algorithm, QI-RL	The paper devised a non-fungible token- based distributed intelligence networking for connected autonomous vehicles by formu- lating the problem as a discrete Markov decision process and employed QI-RL to identify the optimal policy.
[122]	Accurate node deployment	Offloading decision rate, resource potentiality, channel, qubit state-error probability, and fusion variance	NP-hard	QI online node consolidation, QI-RL	The issue of reliable service deployment was solved through the proposed QI node consolidation algorithm that utilized time- sensitive measurement reinforcement and angular-based node position analysis.
[121]	Minimize task processing delay	Risk, computing capacity and maximum number of offloaded tasks	Markov decision process	Grover iteration, QI-RL	The paper proposed a train-edge-cloud col- laborative computing framework to minimize task delays through a risk-sensitive QI-RL algorithm for smart train systems and in- troduced a novel risk function considering edge intelligent servers load and rail transit characteristics.
[115]	Maximize throughput	Power, energy flow, data rate, and data buffer overflow	Markov decision process	QRL	This work modeled the relay and transmit power selection problem as a Markov deci- sion process and employed QRL algorithm based on Grover iterations to optimize the consumed energy and throughput.
[131]	Minimize expectation of unit cost	Learning in adversarial environment	Multi-armed bandit problem	Quantum amplitude amplification, Grover iterations	The authors solved tasks offloading problem through QI bandit learning, where the client learns the costs of each task by utilizing quantum computing principles.
[143]	Maximize stato- dynamic sensing ability	Sensors in vicinity	Bounded- error quantum polynomial	QI IoT optimiza- tion	This work incorporated quantum formaliza- tion of sensor-specific parameters in a real- time IoT environment and devised a quantum computing-inspired optimization technique to maximize the data accuracy.

of RIS in communication networks requires advancements in resource optimization plans, hardware design, channel estimate techniques, and optimization algorithms. QI solutions harness the ability to explore multiple configurations simultaneously due to inspiration drawn from quantum computing concepts. It facilitates effectively exploring the solution space through model optimization procedures similar to quantum annealing, creating associations among RIS components, and utilizing fast-converging data-driven ML techniques. These methods provide efficient optimization and ensure flexibility.

IRSs are important in 6G systems due to their energy efficiency. The authors of [30] formulated a combinatorial problem of radio resource optimization for a standalone IRS system. The proposed QUBO formulation provided an energysaving solution, and it does not require external information, thus eliminating the heavy overheads of channel estimation for each IRS's reflective element, phase shift optimization, and control mechanism. The proposed technique was compared with three methods: A, B, and C. In method A, a sub-set of operators could control the IRS and perform phase shifts. In method B, each operator sequentially gained the right to control the IRS in a single time slot cycle. In method C, simultaneous communication was considered among all operators to optimize phase shift, which caters to maximizing the throughput. There were several constraints, such as traffic for each user equipment, avoiding resource allocation to user equipment excluded in the radio link table in each time slot, upper bound on sub-carriers allocated to each user equipment,



Fig. 24. Intelligent reflecting surface providing reliable connectivity to user.

and the total count of sub-carriers allocated to each user equipment does not exceed the available range if same BS communicated with multiple user equipment in each time slot.

The IRS-based improved energy utilization in 6G networks refers to achieving the same throughput with a reduced transmission power and frequency bandwidth. In [86], the IRS allocation scheduling was formulated as a QUBO problem, and quantum computing was used to solve the combinatorial optimization problem. The objective was to identify the ground state to minimize the Hamiltonian. The ground state of the Ising model was identified through the adiabatic theorem. The system considered two user pieces of equipment communicating with a BS through a single IRS. The reflection coefficient of each IRS was adjusted to maximize the array gain for the user having the IRS assigned. D-Wave Advantage and hybrid solver were used to solve the formulated problem. The constraints included traffic, user equipment assignment to the IRS, traffic reduction avoidance due to interference, and prevention of IRS allocation in case of non-availability of wireless links. The performance of the proposed technique was compared with orthogonal frequency division multiple access and frequency sharing under various conditions such as IRS array gain, traffic reduction due to interference, and signal-tonoise ratio variations.

A larger IRS leads to limited improvements in performance. Hence, Zhang et al. proposed a flexible and secure intelligently mixed reflecting and relaying surface-aided communication system in [79] where the designed fast intelligent reflective elements working mode selection technique enabled individual intelligent reflecting elements to select their working modes, such as reflection or relaying. The network secrecy energy efficiency maximization task was formulated analytically, and secure energy-efficient communication was ensured by employing fast intelligent reflective elements working mode selection and secure dynamic hybrid resource optimization. Specifically, the QI bald eagle search algorithm was proposed to solve the secure dynamic hybrid resource optimization problem. There was a constraint on the target secrecy rate and power allowed by each BS and intelligent reflecting element antenna. The computational complexity of the QI bald eagle search algorithm was  $O(iP_s(3n+2))$ , where i,  $P_s$ , and n refer to the number of iterations, quantum bald eagle population, and dimensions, respectively. The performance of the proposed secure dynamic hybrid resource optimization based bald eagle search technique was compared with other heuristic and classical algorithms such as secure dynamic hybrid resource management-PSO, secure dynamic hybrid resource management-Dinkelbach, passive-IRS, active-IRS, and direct transmission.

Phase optimization is necessary in RISs to obtain the desired reflected wavefront. Hence, the authors of [85] transformed the beamforming problem into an NP-hard problem of identifying the target Ising Hamiltonian's ground state. The formulated problem had an exponential solution space. Two sub-Hamiltonians relating to energy and enforcing weighing constraints simplified the formulation due to multiple spinspin interactions. There were constraints on RIS array size, material thickness, and beam weight. The proposed RIS-Ising-quantum annealing technique was compared with two benchmark schemes: simulated annealing provided by the D-Wave neal package and branch and bound solver of IBM's complex linear programming expert optimizer. However, the proposed quantum annealing can solve relatively smaller-sized problems due to a mismatch between the fully connected graph generated by RIS Ising models and the sparse Pegasus graph used in D-Wave quantum processing unit hardware. Therefore, a hybrid quantum annealing scheme was proposed to mitigate the mismatch of quantum annealing.

Table XII provides the key details of the reviewed sub-set of papers related to the resource optimization of RIS. The tabulated information includes objectives, constraints, problem type, solution, and remarks.

**Lessons Learnt:** The research gaps and future directions listed below have been identified from the dataset used for the literature review. The QI approaches can be explored to address these appealing research pathways to enrich the radio resource optimization of RIS.

- *IRS Deployment and Optimization:* Current research can be improved by focusing on three key areas: optimizing the placement of IRS alongside BSs, refining patterns in phase shift and switching, and integrating a control technique for managing multiple BSs and IRSs [30]. Similarly, real-time optimization of large-scale RIS structures can be investigated [85].
- *Evaluation of Secrecy Energy Efficiency:* The proposed technique's effectiveness in maintaining secrecy energy efficiency in scenarios with imperfect channel state information can be assessed [79].
- Throughput Maximization: To maximize throughput, simultaneous optimization can be conducted for each base station's transmission power, transmission precoding, and IRS reflection coefficient [86].

#### D. Quantum-Inspired Spectrum Sensing and Localization

Spectrum sensing optimization aims to better identify and manage the radio frequency spectrum for improved wireless communication. Similarly, in localization, the position of a device or user is determined through signal strength measurements or GPS. Fig. 25 illustrates integrated spectrum sensing and localization of target objects within the coverage area of

Ref.	Objective	Constraints	Problem	Solution	Remarks
[30]	Optimize the energy	Traffic, and sub- carriers	Complex com- binatorial opti- mization prob- lem	QUBO	The authors applied quantum computing for radio resource op- timization of the proposed stan- dalone IRS and alleviated the con- cerns of impractical overhead in designing phase shift for IRSs.
[79]	Maximize network secrecy energy efficiency	Target secrecy rate and power	Secure dynamic hybrid resource optimization	QI bald eagle search	Increasing the number of ele- ments in IRS leads to minor im- provement in performance, hence a fast reflecting element's work- ing mode selection method was devised based on QI algorithm.
[85]	IdentifythegroundstateoftargetIsingHamiltonian	RIS array size, ma- terial thickness and weighing of beams	NP-hard	Hybrid quantum annealing	This paper solved the problem of optimizing the phase configura- tion of RIS.
[86]	Maximize the array gain	Traffic, user equip- ment assignment to IRS, traffic reduction avoidance due to in- terference, and pre- vention of IRS allo- cation in case of non- availability of wire- less links	Combinatorial optimization problem	QUBO	The optimization of IRS alloca- tion scheduling was considered while limiting the number of users and IRS array gain was maxi- mized by configuring the reflec- tion coefficients.

TABLE XII RIS optimization using QI techniques.



Fig. 25. Spectrum sensing and localization of target objects [144].

a wireless network. It shows three base stations, a smart car positioned at the intersection of their coverage areas, twodimensional coordinates (x, y), distance measurements (di) from each base station, and a cloud platform to coordinate the sensing and localization tasks across all stations. Key challenges include adaptability to heterogeneous environments, coordinated spectrum sensing, detection accuracy, energy efficiency optimization, interference, and security against malicious attacks. Resolving these issues is crucial to optimizing spectrum use and improving the general performance of communication networks. Technological advancements in signal processing, AI, and ML aid in overcoming these obstacles and enhancing spectrum sensing abilities.

Incorporating quantum principles into classical computing approaches enables QI techniques to optimize spectrum sensing effectively. These techniques improve spectrum sensing parameter optimization efficiency by exploring numerous solutions simultaneously through quantum annealing, superposition, and parallelism. Spectrum sensing techniques get coherence and data-driven adaptation from quantum entanglement and ML algorithms. These methods effectively handle the complexity of spectrum sensing optimization in communication networks and provide flexibility in response to changing spectrum conditions. While QI methods on classical computers help improve spectrum sensing capabilities, real quantum computers may offer further benefits.

Cognitive wireless medical sensor networks are also beneficial in healthcare systems for remote transmission of information and monitoring purposes. However, the classical static spectrum allocation technique limits the wireless sensors' communication using a single channel and could cause packet loss or interference issues. Addressing these issues, a cognitive spectrum allocation technique binary quantum-behaved elite PSO algorithm was proposed in [73], which followed the bird colony behavior. The objective was to optimize network revenue and spectrum allocation. The designed interference matrix had a constraint called non-interference. An undirected graph denoted the network, and the spectrum allocation was modeled as a graph coloring problem. Genetic and classical PSO algorithms were considered benchmarks, and the performance metrics included network reward and spectrum allocation efficiency.

The spectral efficiency can be achieved in 5G networks through co-frequency self-interference cancellation techniques. However, the cross-link interference is a performancelimiting factor. Therefore, the authors in [88] devised a twoloop technique to maximize spectral efficiency. The objective was to optimize duplex mode selection and secrecy transceivers jointly. It was a non-convex mixed integer problem that incorporated binary variables and was regarded as an NP-hard problem. A QI tabu search algorithm was proposed for duplex mode enhancement in the outer loop. A successive convex approximation was employed in the inner loop to optimize transceivers and artificial noise. QI tabu search diversified and aimed to identify solutions through intensive search. The successive convex approximation and QI tabu search algorithms showed fast convergence within ten to fifteen and ten iterations, respectively. The colluding and non-colluding eavesdroppers were considered, and the constraints included antenna operational mode, transmission power, and uplink receivers. The benchmark schemes used for comparison included greedy search, network-assisted full duplex with fixed duplex mode, and time-division duplex. The overall complexity of the proposed technique was  $O(C(V_a^{2.5}C_o^2 + V_a^{3.5}))$ , where  $C, V_a$ , and  $C_o$  refer to the count of possible solutions, scalar variables, and constraints respectively.

Considering the need for resource optimization in a 6G network scenario, power-domain non-orthogonal multiple access (NOMA) developing on RL-inspired QNN was proposed in [145] for user grouping and meeting spectral efficiency. The objective was to maximize the user sum rate while incorporating constraints on power. The proposed RL-QNN and QNN approaches had a time complexity of  $O(n_{layer}n_{neuron})$ , which was superior to the conventional NN's time complexity of  $O(n_{layer}(n_{neuron})^2)$ , where  $n_{layer}$  and  $n_{layer}$  represent the number of neurons and neural layers, respectively.

Sensor errors, noise, and multipath effects usually impact the accuracy of indoor localization. In [107], a hybrid quantum deep NN model was proposed to achieve accelerated learning through quantum computing and handle complex data through deep NN. The proposed scheme showed superior localization through received signal strength and time of flight metrics, compared to benchmark techniques, including a Euclideanbased K-nearest neighbor, a classical deep NN with and without dropout. Furthermore, the complexity of the proposed scheme was reduced by decreasing the number of variational parameters. Therefore, the QI techniques have the potential to address complex localization tasks.

Table XIII presents key details, including the objectives, constraints, problem, type, solution, and remarks for the

sub-set of research papers focused on resource optimization techniques related to spectrum sensing.

**Lessons Learnt:** The research gaps and prospective research directions based on the dataset used for the spectrum sensing optimization literature review are listed below:

- *Integration with Emerging Technologies:* The cognitive wireless medical sensor networks can be integrated with emerging technologies such as 6G networks, edge computing, and the IoT, aiming to enhance spectrum allocation, improve network performance, and design new healthcare applications [73].
- Imperfections in Channel State Information and Noise: Analysis of integrating imperfect channel state information into massive MIMO systems can lead to enhancing performance and reliability [88]. Decentralized QNN can be considered for resource optimization while catering to the performance-related challenges posed by noise [145].

### E. Quantum-Inspired Resource Optimization in UAV-Assisted Networks

UAVs are indispensable components of wireless communication networks, providing an economical and flexible solution for emergency deployments. Fig. 26 illustrates a UAV-assisted emergency coverage scenario, emphasizing the need for an optimized flight path. The UAV starts at a hovering point within a cell with intact network resources and a control center, then follows an optimal trajectory towards an emergency-struck area, previously served by a base station, while accounting for obstacles such as buildings. Trajectory planning optimizes UAV flight routes in communication networks, enhancing communication capabilities. Some challenges include avoiding obstacles, adapting to different environments, optimizing energy efficiency, ensuring good communication range and quality, maintaining network connections, using complex path planning algorithms, and accounting for weather conditions. The OI techniques simultaneously explore multiple trajectory options, simulate quantum annealing-like optimization, create associations between planning elements, employ data-driven ML algorithms, explore solution space effectively, and provide novel hybrid approaches. They offer effective optimization methods and guarantee flexibility in dynamic environments. The UAV trajectory planning process is improved by QI techniques, allowing for efficient and safe navigation in dynamic communication networks.

Specifically, trajectory planning can be performed in a cellular-connected UAV environment to minimize flight time while ensuring satisfactory wireless transmission quality. In [126], the UAV's flight path was formulated as a Markov decision process, and optimized flight direction within each time slot was identified through a deep RL-based approach. The trajectory planning problem was too sophisticated to be solved through standard optimization techniques. Hence, a QI experience replay framework was proposed, which inter-linked the experienced transition to the associated qubit and applied Grover's technique iteratively for amplitude amplification. The

TABLE XIII Spectrum sensing and Localization using QI techniques.

Ref.	Objective	Constraints	Problem	Solution	Remarks
[73]	Optimize network revenue and spectrum allocation	Non-interference	Graph color- ing problem	Binary quantum- behaved elite particle swarm algorithm	This paper addressed the problem of efficient spectrum allocation in cogni- tive medical wireless sensor network through quantum computing algorithm.
[88]	Maximize secrecy spectral efficiency	Antenna operational mode, transmission power, and uplink receivers	NP-hard	Successive con- vex approxima- tion, QI tabu search	This paper addressed the joint opti- mization of duplex mode selection and secrecy transceivers to maximize over- all secrecy spectral efficiency.
[107]	Minimize mean square error	None	Indoor local- ization	Hybrid quantum deep NN	The paper aimed at minimizing the localization error in a real environment by using low-cost devices for received signal strength measurements and limited time of flight.
[145]	Maximize user sum rate	Power	Grouping of users	Quantum NN	The paper uses quantum NN to achieve the required spectral efficiency of next- generation wireless communication.



Fig. 26. UAV-assisted networking scenario.

objective was to minimize each UAV's flight time cost and corresponding expected ergodic outage duration. The constraints included sector association, UAV location, flying direction, mobility, normalization, and priorities. The proposed approach was compared with several other deep RL-based benchmarks, such as standard experience relay, prioritized experience replay, deep curriculum RL, and simultaneous navigation and radio mapping. The performance metrics included an average weighted sum of ergodic outage duration and time cost.

In contrast to terrestrial networks, UAVs are considered a favorable means of wireless service provisioning due to their movability and on-demand deployment. The authors of [123] performed UAV's flight trajectory planning, aiming at maximizing the ground users' expected sum uplink transmission rate. A QI-RL technique was proposed to optimize the UAV's trajectory without knowing user information, such as location, channel state information, and transmit power. There were constraints on flying distance between arbitrary adjacent time slots, the UAV's trajectory was within a feasible area, maximum exploration time was constrained by onboard power capacity, and the sum of individual ground user bandwidths must not exceed available bandwidth. Classical optimization approaches cannot solve such a problem due to the unavailability of environmental information. Model-free RL methods can be incorporated to solve such problems by employing the "trial and error" method. The proposed technique was compared with two conventional RL benchmarks, such as Q-learning with  $\epsilon$ -greedy and the Boltzmann exploration technique.

The authors of [76] proposed a UAV path planning model and multi-objective QI seagull optimization algorithm based on decomposition. They devised a logic to disintegrate the multi-objective task into multiple scalar optimization tasks. The concept of neighbors was incorporated into the proposed algorithm to enhance the connection between seagulls. There was a constraint on the independent variable's feasible region, and flight height has upper and lower bounds. The performance metrics included convergence, distributivity, and extensiveness, which were evaluated through parameters such as hypervolume and spread. The benchmark algorithms used for comparison included multi-objective PSO, non-dominated sorting genetic algorithm II, multi-objective evolutionary algorithm based on decomposition, multi-objective vortex search algorithm, multi-objective artificial algae algorithm, and multiobjective spotted hyena optimizer. The standard test function sets from Congress on evolutionary computation-2009, Deb-Thiele-Laumanns-Zitzler, and Zitzler-Deb-Thiele were used for performance evaluation of the aforelisted techniques and incorporated properties such as convex, concave, disconnected, and linear. The time complexity of the proposed technique was approximated as  $O(P_s^2)$ , where  $P_s$  refers to the population size (and count of weight vectors).

The UAV path planning is usually decomposed into two steps, including viable path identification between two points and trajectory optimization. Fernandes et al. proposed an enhanced diversity PSO technique in [74] aimed at efficient and safe path identification, energy wastage avoidance, and system integrity. The proposed scheme was evaluated in twodimensional static and dynamic environments. The cost function was devised based on the route length, terrain crossed, and obstacle avoidance. The stochastic ranking was used for invalid paths, such as crossing through obstacles. Such paths were considered constraints and treated with a death penalty, which referred to assigning a high value to the collision parameter. The performance of the proposed technique was evaluated in comparison to various benchmark algorithms, such as enhanced evolutionary PSO, cooperative multi-objective PSO, grey wolf optimizer PSO, adaptive genetic PSO, global and cooperative PSO, QPSO, PSO, genetic algorithm, and cuckoo search. The metrics used for performance evaluation included the reciprocal of Pareto sets proximity, reciprocal of hypervolume, inverted generational distance in decision space, and inverted generational distance in objective space. The aforementioned metrics were evaluated through the Congress on evolutionary computation benchmarking suite, comprising 22 functions with different characteristics. In particular, the QPSO algorithm showed a higher average convergence cost than the enhanced diversity PSO algorithm.

A motion-encoded PSO algorithm was proposed in [146] to identify a fast-moving target through UAVs. The objective of the NP-hard optimization task was to maximize the cumulative probability of target detection. The search trajectory was encoded as a sequence of UAV motion paths while preserving the cognitive and social coherence of the swarm. The problem considered the target's last known location and employed a normal probability distribution function through a

grid mapping called a belief map. There was a constraint, such that the UAV could move in one out of eight directions during each step. The QPSO algorithm was used as a benchmark and showed inferior performance compared to the proposed technique in identifying the fitness value of accumulated detection probability.

Similarly, the UAV target tracking control can provide intelligent consciousness remotely due to its major role in patrolling, surveillance, and target monitoring. Ma et al. proposed a QI experience relay deep deterministic policy gradient technique based on deep RL framework for vertical take-off and landing aircraft in [147]. The tracking control problem was formulated as a Markov decision process. The aim was to learn an optimal tracking control policy without prior knowledge for autonomous tracking of a target in flight through UAVs. The reward function for UAVs comprised two tasks: effectively reaching the desired position in less time and providing high precision target tracking guidance. The average reward of the proposed QI experience relay deep deterministic policy gradient technique increased more quickly with an effective sampling of high-priority samples and fewer fluctuations, compared to the benchmark techniques such as prioritized experience replay deep deterministic policy gradient and deep deterministic policy gradient.

Table XIV shows the objectives, constraints, problem, type, solution, and remarks for the sub-set of research papers focused on resource optimization techniques related to UAV trajectory.

Lessons Learnt: The UAV trajectory path planning research can be enriched by considering the following research gaps

Ref.	Objective	Constraints	Problem	Solution	Remarks
[74]	Efficient and safe path identification, energy wastage avoidance, and ensuring system integrity	Collisions, paths with obstacles	Trajectory planning problem for autonomous mobile robots	Enhanced diver- sity PSO	This paper solved mobile robotic vehicles trajectory planning prob- lem for static and dynamic envi- ronments.
[76]	Multi-target optimization	Feasible region of independent variable, upper and lower bound on flight height	Multi- objective problem	Multi-objective QI seagull optimization algorithm based on decomposition	The proposed technique trans- formed multi-objective problem into several scalar sub-problems and tested its effectiveness for UAV path planning.
[126]	Minimize UAV flight time and expected ergodic outage duration	Sector association, UAV location, fly- ing direction, mo- bility, and priorities	Markov deci- sion process	Grover iteration based amplitude amplification, Deep RL	The authors formulated a UAV navigation approach to address the problem of minimizing the weighted sum of time cost and expected outage duration.
[123]	Maximize expected sum uplink transmission rate through UAV flight trajectory planning	Flying distance, power and bandwidth	Markov deci- sion process	QI-RL	The authors solved trajectory planning problem of UAV through QI-RL to optimize expected sum uplink transmission rate.

TABLE XIV QI RESOURCE OPTIMIZATION IN UAV-ASSISTED NETWORKS.

and directions identified from the dataset of papers used in this survey.

- GPS and UAV Navigation: The research can consider integrating GPS technology to enhance the navigation capabilities of UAVs, enabling more accurate and reliable positioning, trajectory planning, and overall flight control [126]. Additionally, Deep RL-based formation control in GPS-denied environments can be explored to address challenges UAVs face, such as maintaining formation and performing coordinated tasks effectively [147].
- *Transmission Rate Optimization:* A feasible direction worth exploring is maximizing the expected sum uplink transmission rate for ground users in multi-UAV scenarios [123]. This involves optimizing the system to achieve higher average data transmission rates from multiple users to the BS or access point, enhancing uplink performance and overall throughput.
- *Power Systems Dispatch:* A prospective area for future research is applying the proposed technique to power systems dispatch [76]. It can aid in optimizing power resource allocation, leading to enhanced efficiency and improved performance in power grids.
- *Complexity Reduction:* The research can evaluate the proposed technique for multiple mobile robots and reducing computational complexity [74]. This involves optimizing coordination and control by developing algorithms and techniques to ensure efficient and effective operation.

#### F. Quantum-Inspired Power Control Optimization

Optimal power control is crucial in wireless networks to extend the battery life of edge devices and ensure reliable data transmission. Fig. 27 illustrates a base station with dynamic transmission power allocation. The system optimizes communication with three UAVs, each providing coverage to their respective zones. UAVs cohesively account for multiple power-affecting parameters, including transmission, flight, and data processing power, while managing their battery status to maintain coverage in their respective zones.

The MIMO systems employ optimal power control to achieve high transmission quality and reduced computational complexity through linear processing, but the security of transmitted information is a significant concern. Therefore, a secure co-time co-frequency massive MIMO network was designed in [78] that considered passive eavesdroppers and aimed to improve time-frequency resource utilization while ensuring physical layer security. The proposed joint antenna selection and power allocation problem was NP-hard. The objective was to maximize the secrecy capacity and energy efficiency. The optimized joint antenna selection and power allocation scheme was obtained through hybrid optimization functions through a QI backtracking search algorithm that had a computational complexity of  $O(i(5P_s+4P_sn))$ , where  $i, P_s$ , and n refer to iterations, population size, and dimension (of quantum individuals), respectively. There were constraints on power, uplink, and downlink transmission rates. The proposed technique was compared with benchmarks including PSO, Discrete PSO, artificial physics optimization, and backtracking search optimization algorithm.



Fig. 27. Scenario depicting the necessity of optimized power allocation.

The multicarrier communication system performance depends on the peak-to-average power ratio. Selected mapping significantly reduces the peak-to-average power ratio without losing spectrum resources. Hou et al. proposed a QI evolutionary technique in [62] to solve this NP-hard problem through combinatorial optimization of angle increment updating and characteristics information ranking. The objective was to minimize the maximum signal of discrete-time transmitted orthogonal frequency division multiple access signal, and the sign sequence was constrained. The proposed scheme had a computational complexity of  $P_s \times G \times O(N_c \log N_c)$ , where  $P_s$ , G, and  $N_c$  are population size, generation value, and sub-carriers, respectively. The benchmark techniques used for comparison included artificial bee colony selected mapping, cross-entropy selected mapping, and selected mapping. The performance metrics included average peak-to-average power ratio, bit error rate, and raw cubic metric.

The authors of [77] proposed an efficient power allocation and antenna selection technique called continuous QI termite colony optimization algorithm. It ensured reliable communication in a co-frequency full duplex massive MIMO system. The precoding and reception at the base station were done through maximum ratio transmission and maximum ratio combination. The simultaneous transmission and reception lead to selfinterference, and the degree of self-interference cancellation ranged from 0 to 1, where 0 meant no self-interference. The aim was to maximize the transmission capacity and energy efficiency. It was a non-convex NP-hard problem. The number of transmitting antennas and the uplink transmission power of each user were discrete and continuous variables, respectively. They belonged to different optimization domains and were transformed into a continuous optimization problem. There were constraints on the transmission rate, power, and search range of quantum termites. The benchmarks used for comparison included backtracking search optimization, half-power random resource allocation, PSO, and termite colony optimization. The computational complexity of the continuous QI

termite colony optimization algorithm was  $O(i(5P_s + 3P_sn))$ , where *i*, *n*, and *P<sub>s</sub>* represent iterations, dimensions, and population size (of quantum termites), respectively.

The optimization of achievable data rate and network coverage in wireless communication were considered key performance metrics in [106]. A QNN employing parallel training was proposed to optimize transmit precoding and power allocation in NOMA with MIMO. The objective was to maximize the sum rate of edges while incorporating the constraints on power allocation range and transmit precoding. The slope of the sum rate of edges decreased with respect to the transmit signal-to-noise ratio due to inter-user interference. The proposed parallel training QNN scheme was compared with the conventional gradient-descent-based approach and static power allocation-maximum rate precoding. The overall complexity of the feed-forward process and unsupervised parallel training for QNN was  $O(N_{shot}N_{layer}N_{neuron})$ , which was found to be superior to the conventional approach having overall complexity of  $O(N_{edge}.(N_{shot}N_{layer}N_{neuron} +$  $N_{shot}N_{layer}^2N_{neuron}^2)$ , respectively.  $N_{edge}$ ,  $N_{layer}$ ,  $N_{neuron}$ , and  $N_{shot}$  denote the number of edges, layers in NN, neurons in NN, and quantum measurements, respectively. The drawback of the proposed approach lied in higher memory utilization for the parallel training process.

VR videos impose significantly greater power requirements than other video types due to viewport rendering. The key techniques for wireless VR include content-correlation-based computation offloading, quality of experience-aware content transmission, and buffer-aware content caching. Therefore, Lin et al. proposed a joint intelligence framework in [124] considering content correlation such as caching, computing, and transmission. The three-layered framework comprised reality access, user access, and a virtual environment layer. A OI-RL scheme was proposed to solve the multidimensional resource provisioning problem by jointly optimizing the streaming. In the proposed technique, the agent performed online learning and decided about task offloading and multidimensional resource allocation decisions relating to computation, spectrum, and storage. The environment was referred to as a time-varying channel and was modeled as a Markov decision process. The policy referred to rewards, such as energy consumption and quality of experience of VR equipment. The objective was to minimize the energy consumption of VR equipment. There were constraints on resource capacity, flow stability, offloading capacity, and resource utilization assurance.

Table XV shows the objectives, constraints, problem, type, solution, and remarks for the sub-set of research papers focused on resource optimization techniques related to optimized power control.

**Lessons Learnt:** The research related to optimal power allocation can be enriched by considering the following research gaps and directions identified from the dataset of papers used in this survey.

• Ultra-Dense Scenarios and Heterogeneous Networks: The proposed technique can be evaluated for ultra-dense scenarios and heterogeneous networks in the presence of many eavesdroppers [77] and [78]. As a future direction, • VR Streaming Services: While targeting joint optimization of streaming services in VR, the security concerns can be considered a future direction [124].

of system parameters variation can be explored [62].

#### VI. USE CASE: QUANTUM-INSPIRED VERTICAL FARMING

The 6G networks will deal with compute-intensive and dense sensor environments, offering new opportunities for integrating QI optimization into smart farming practices, including vertical and digital farming. These practices optimize urban spaces by using multi-layered structures within controlled environments, such as unused buildings or repurposed warehouses, to boost crop productivity and reduce the carbon footprint [148]. For example, a vertical farming setup spanning  $50 m^2$  might include crop stacks arranged in rows, each featuring five vertical layers with five sensors installed in each layer that monitor plant health, soil moisture, and environmental conditions. Each row, measuring 1 m in width and spaced 1 m apart, extends 24 m in length. This setup will result in approximately 1,250 sensors, many of which would be in hard-to-reach positions.

The sensors in such an environment generate tasks with low computational complexity and flexible latency requirements. However, managing a high density of sensors in large farms necessitates robust and scalable mobile edge computing techniques to handle data influx and provide wireless energy harvesting, thereby maximizing sensor lifetime. Fig. 28 illustrates the effectiveness of QI optimization in vertical farms for efficiently managing the extensive network of IoT sensors, ensuring effective monitoring of environmental conditions and plant health. UAVs can utilize the QI algorithms to dynamically optimize flight paths for real-time data acquisition and edge processing, enhancing data collection efficiency and energy harvesting. Additionally, agricultural robots and IoV for the transportation of goods to and from the farm can benefit from QI optimization for automated resource management and maintenance tasks, ensuring efficient scheduling and routing.

QPSO is particularly effective in balancing exploration and exploitation with minimal computational overhead, making it ideal for managing high-dimensional sensor data and dynamic farming environments [67], [71]. It integrates seamlessly with IoT and edge computing systems, ensuring robust resource allocation. The QI simulated annealing is well-suited for realtime environmental adjustments, though it may exhibit slower convergence in highly dynamic scenarios [136]. QI-RL can enhance IoV systems for agricultural robots and vehicles [117], while the QI Genetic algorithm addresses complex, multi-objective problems such as UAV flight path optimization. However, its higher computational demands and hardware requirements may limit its real-time applicability, making it preferable where high precision is prioritized over speed [11].

 TABLE XV

 Power Allocation Optimization using QI techniques.

Ref.	Objective	Constraints	Problem	Solution	Remarks
[62]	Minimizethemaximumsignalofdiscrete-timetransmittedorthogonalfrequencydivisionmultipleaccesssignal	Sign sequence is con- strained	NP-hard	QI evolution- ary algorithm	This paper solved the problem of high peak-to-average power ratio through selected mapping technique.
[77]	Maximize transmis- sion capacity and en- ergy efficiency	Transmission rate, power and search range of quantum termites	Non- convex, NP-hard	Continuous QI termite colony optimization algorithm	A QI algorithm was proposed for energy-efficient resource optimiza- tion while aiming to achieve a bal- ance among energy consumption, system resource utilization, and over- all transmission capacity.
[78]	Maximize secrecy capacity and energy efficiency	Power, uplink, and downlink transmission rates	NP-hard	QI backtrack- ing search al- gorithm	The authors aimed at joint optimiza- tion of antenna selection and power allocation in co-time co-frequency full duplex MIMO network.
[106]	Maximize edge sum rate	Power and transmit precoding	NP-hard	Parallel train- ing QNN	The wireless resources optimization was achieved by requiring the edges to transmit only statistical parameters.
[124]	Minimize energy consumption of VR equipment	Resourcecapacity,flowstability,offloadingcapacity,andresourceutilizationassurance	Markov decision process	QI-RL	This paper tackled multidimensional resource provisioning issues in wire- less VR through a joint framework comprising of computation, storage and communication resources.



Fig. 28. Vertical farming system model for QI optimization.

Therefore, the selection of the QI optimization technique depends on the system's specific requirements and available resources. Conclusively, these QI optimization techniques offer seamless integration into classical networks, eliminating the need for specialized infrastructure.

#### VII. FUTURE RESEARCH DIRECTIONS AND CHALLENGES

Considering the exponential increase in communication devices, there is a dire need for novel optimization techniques to achieve the desired communication network performance through efficient resource optimization and resilience to noise. This section organizes future directions to offer clear insights from the reviewed dataset of QI optimization papers.

1) Applications in 6G networks: The quantum technology is favorable to addressing various communication problems, including cryptosystems for blockchain, hybrid algorithms, IoT and IoV optimization, multimedia processing, satellite communication, and UAV trajectory planning [12]. The vehicular routing problem addressed by the proposed scheme inspired by quantum computing can also integrate objectives such as minimizing distributed network latency and ensuring the authenticity and security of the quantum platform [143]. The quality of service enhancing clustering technique proposed in [60] can be applied to large-scale and mobile integrated energy system-based wireless sensor nodes. The marine predators algorithm with crossover proposed in [149] can be reformulated and evaluated in a QI scenario while hybridizing with fuzzy logic or artificial NNs. The QPSO can show slow convergence in large-scale scenarios. Therefore, it is necessary to reduce the convergence time for the proposed QPSO technique for multi-hop wireless power transfer in [72]. The dynamic resource optimization scheduling technique proposed in [70] developing on QPSO and proportional integral

can be employed on a large-scale and in cooperative user scenarios. Analyzing how individual user response patterns influence scheduling decisions helps identify discrepancies between service requests and resource allocation, enhancing dynamic resource optimization scheduling performance.

2) Complexity Reduction Techniques: The QI optimization techniques offer significant speedups compared to classical optimization approaches. Further reducing the complexity of QI optimization algorithms is a favorable domain [8]. The gradient-free optimization technique can optimize the circuit parameters of the quantum approximate optimization algorithm [150]. The hybrid classical-quantum optimization approaches, such as QUBO and higher-order unconstrained binary optimization, lead to tractable solutions, which may not be possible through classical approaches. Therefore, analyzing the possibility of further enhancements involves integrating the bits, neurons, and qubits [6].

The quantum energy balancing technique for IoT sensors proposed in [58] can be compared with other QI metaheuristics techniques for green communication in IoT networks. The algorithms proposed for maximizing energy efficiency, such as QPSO and improved discrete PSO in [69], can consider evaluation in complex terrain and terahertz bands. The QPSO technique proposed for energy efficiency maximization in [65] only considers one relay unit. Considering multiple relay units for transferring computational tasks can enhance the system. Similarly, the iteration process significantly influences the clustering performance of proposed single-objective cat swarm optimization and multiple objective cat swarm optimization techniques in [110]. Implementing alternate progressive optimization techniques can further enhance the running speed. The QI evolutionary algorithm proposed in [61] aims to optimize caching and user association jointly. Enhancing the techniques involves considering additional parameters, such as user association in adaptive bit-rate scenarios, resource allocation, and UAV deployment strategies.

3) Machine Learning-based Approaches: **Ouantum** learning-based green computing is an appealing technique for connected vehicular networks, big data processing, visualization, and integration of classical optimization with quantum learning [115]. In the IoV scenario, the QI-RL framework presented in [117] can facilitate self-adaptive resource optimization on a broader scale, allowing for the evaluation of average delays. Similarly, researchers can extend the QI-RL technique proposed for optimizing a single UAV trajectory path in [123] to simultaneously optimize the trajectory paths of multiple UAVs, aiming to maximize the aggregate uplink rate. Expanding the QI experience replay technique proposed in [126] involves extending it into deep RL-based models, such as deep deterministic policy gradient, soft-actor centric, and Rainbow. The multi-party quantum devices can harness benefits from a collective utilization of federated and transfer learning-based techniques [105]. Examining the proposed QI support vector machine algorithm and error propagation process can help reduce complexity and tighten error bounds [151].

4) **Peak-to-average Power Reduction:** The 3rd generation partnership project establishes stringent standards for telecommunication networks, particularly for error vector magnitude. It is a metric that assesses the accuracy of transmitted signals by measuring the deviation from their ideal form, and a lower value is crucial for robust data transmission. However, the challenge arises with the peak-to-average power ratio, highlighting the variation between the peak signal power and its average power. Its high value can lead to signal distortion and inefficient transmission, severely affecting communication quality [62].

Classical optimization methods, such as mixed integer programming, will struggle to address the complexity of 6G networks, such as real-time, dynamic datasets on a massive scale. Quantum annealing solves large-scale combinatorial problems by simulating quantum behaviors like tunnelling to escape local optima. This capability enables it to explore numerous solutions simultaneously. It offers an efficient way to minimize the peak-to-average power ratio while complying with the error vector magnitude constraints set by 3rd generation partnership project. Therefore, quantum annealing can optimize network performance to meet the increasingly complex demands of future 6G standards, outperforming classical methods in both speed and accuracy [152].

5) **Ouantum Key Distribution and Entangled pairs**: The medoids-based clustering technique proposed in [49] for distributing quantum keys over a fiber optic network can undergo evaluation in a QI wireless network scenario. Furthermore, exploring the effect of inter-network connectivity, extension of key distribution territory, and physical features on QKD quality is also possible. The dynamic programming-based algorithm proposed in [32] only targets latency minimization of entangled pairs; analyzing aggregated trees, pipelining, purification, and multimode memories can also be performed. The technique proposed in [31] for optimized entanglement generation distribution can extend to a large-scale network incorporating quantum repeaters. Incorporating security provisioning for highly mobile IoT devices, maximizing the integration of narrowband IoT systems, and evolving machine-type communications can enhance the quantum-resistant access authentication and data distribution technique for IoT devices proposed in [153].

6) **Resilience to Noise:** Investigating the noise performance of the photonics-inspired recurrent NN proposed in [154] is possible through its implementation on photonic hardware.

7) Transitioning from Classical to Quantum Infrastructure: The transition to quantum infrastructure in networking necessitates the development of hybrid architectures to maintain service continuity during the transition phase. Recent advancements in routing optimization for 6G networks, particularly through applying QML approaches, necessitate adaptive and intelligent network management protocols capable of dynamically adjusting to the complexities of quantum communication [155]. As scalability remains a significant challenge, ensuring that quantum hardware can support largescale networks is crucial for a successful transition.

8) Challenges: Realizing a quantum-integrated communication networking scheme requires addressing several key challenges, which include learning QML models, decoherence, noise, and classical-quantum system integration [22]. The secret-key flow model proposed in [47] considers fiber optic networks where infrastructure upgrades are necessary for QKD integration while accounting for the distance limitations for QKD transmission. The fiber optic-based scalable and reconfigurable quantum network architecture proposed in [34] for photon-pair distribution would encounter challenges related to quantum resource constraints, such as the availability and generation of entangled photon pairs, quantum noise and loss, and compatibility with existing infrastructure. The quantum frequency processors categorized in [156] target full process tomography, and their experimental characterization would give rise to several challenges, such as non-ideal models for integrated platforms that potentially undermine the openbox assumptions, the computational complexity of black-box methods, and longer integration time due to noise.

#### VIII. CONCLUSION

As the number of devices increases and the demand for reliable communication grows, optimizing communication networks is becoming increasingly complex. The large-scale optimization tasks inherently require more time to identify the optimal solution, leading to increased latency. The classical optimization techniques also tend to get stuck in local optima due to inefficient exploration of the solution spaces. URLLC is crucial in 6G, enabling services like AR, VR, intelligent transportation, and remote surgery. This importance drives the use of QI optimization techniques in communication networks. The reviewed research papers demonstrate that QI metaheuristics, ML and specialized techniques outperform classical benchmark techniques by providing faster and more efficient solutions. The QI techniques for optimizing radio resources leverage quantum physics principles but do not require quantum hardware. Classical entities imitate quantum principles like entanglement, superposition, and tunneling through transformations. This technique enables a faster convergence of meta-heuristics, ML and specialized optimization approaches compared to their classical counterparts. Furthermore, continually exploring and refining such techniques is essential in addressing the dynamic and evolving optimization tasks relating to communication networks.

#### REFERENCES

- [1] C.-X. Wang, X. You, X. Gao, X. Zhu, Z. Li, C. Zhang, H. Wang, Y. Huang, Y. Chen, H. Haas, J. S. Thompson, E. G. Larsson, M. D. Renzo, W. Tong, P. Zhu, X. Shen, H. V. Poor, and L. Hanzo, "On the road to 6G: Visions, requirements, key technologies, and testbeds," *IEEE Communications Surveys & Tutorials*, vol. 25, no. 2, pp. 905– 974, Secondquarter 2023.
- [2] S. Sharif, S. Żeadally, and W. Ejaz, "Space-aerial-ground-sea integrated networks: Resource optimization and challenges in 6G," *Journal of Network and Computer Applications*, p. 103647, Jun. 2023.
- [3] H. Liang, Z. Yang, G. Zhang, and H. Hou, "Resource allocation for space-air-ground integrated networks: A comprehensive review," *Journal of Communications and Information Networks*, vol. 9, no. 1, pp. 1–23, Mar. 2024.
- [4] F. Kooshki, M. A. Rahman, M. M. Mowla, A. G. Armada, and A. Flizikowski, "Efficient radio resource management for future 6G mobile networks: A cell-less approach," *IEEE Networking Letters*, vol. 5, no. 2, pp. 95–99, Jun. 2023.

- [5] X. Song, M. Cheng, L. Lei, and Y. Yang, "Multitask and multiobjective joint resource optimization for UAV-assisted air-ground integrated networks under emergency scenarios," *IEEE Internet of Things Journal*, vol. 10, no. 23, pp. 20342–20357, Dec. 2023.
- [6] A. Singh, K. Dev, H. Siljak, H. D. Joshi, and M. Magarini, "Quantum internet-applications, functionalities, enabling technologies, challenges, and research directions," *IEEE Communications Surveys & Tutorials*, vol. 23, no. 4, pp. 2218–2247, Oct. 2021.
- [7] Z. Li, K. Xue, J. Li, L. Chen, R. Li, Z. Wang, N. Yu, D. S. Wei, Q. Sun, and J. Lu, "Entanglement-assisted quantum networks: Mechanics, enabling technologies, challenges, and research directions," *IEEE Communications Surveys & Tutorials*, vol. 25, no. 4, pp. 2133–2189, Oct. 2023.
- [8] T. Q. Duong, L. D. Nguyen, B. Narottama, J. A. Ansere, D. Van Huynh, and H. Shin, "Quantum-inspired real-time optimisation for 6G networks: Opportunities, challenges, and the road ahead," *IEEE Open Journal of the Communications Society*, vol. 3, pp. 1347–1359, Aug. 2022.
- [9] G. G. Rozenman, N. K. Kundu, R. Liu, L. Zhang, A. Maslennikov, Y. Reches, and H. Y. Youm, "The quantum internet: A synergy of quantum information technologies and 6G networks," *IET Quantum Communication*, vol. 4, no. 4, pp. 147–166, Oct. 2023.
- [10] P. Botsinis, D. Alanis, Z. Babar, H. V. Nguyen, D. Chandra, S. X. Ng, and L. Hanzo, "Quantum search algorithms for wireless communications," *IEEE Communications Surveys & Tutorials*, vol. 21, no. 2, pp. 1209–1242, Apr. 2019.
- [11] O. H. M. Ross, "A review of quantum-inspired metaheuristics: Going from classical computers to real quantum computers," *IEEE Access*, vol. 8, pp. 814–838, Dec. 2019.
- [12] T. Q. Duong, J. A. Ansere, B. Narottama, V. Sharma, O. A. Dobre, and H. Shin, "Quantum-inspired machine learning for 6G: Fundamentals, security, resource allocations, challenges, and future research directions," *IEEE Open Journal of Vehicular Technology*, vol. 3, pp. 375– 387, Aug. 2022.
- [13] L. Huynh, J. Hong, A. Mian, H. Suzuki, Y. Wu, and S. Camtepe, "Quantum-inspired machine learning: A survey," *arXiv preprint* arXiv:2308.11269, Aug. 2023.
- [14] Z. Yang, M. Zolanvari, and R. Jain, "A survey of important issues in quantum computing and communications," *IEEE Communications Surveys & Tutorials*, vol. 25, no. 2, pp. 1059–1094, Secondquarter, 2023.
- [15] S. Koudia, A. S. Cacciapuoti, K. Simonov, and M. Caleffi, "How deep the theory of quantum communications goes: Superadditivity, superactivation and causal activation," *IEEE Communications Surveys* & *Tutorials*, vol. 24, no. 4, pp. 1926–1956, Fourthquarter, 2022.
- [16] A. Kumar, S. Bhatia, K. Kaushik, S. M. Gandhi, S. G. Devi, A. D. J. Diego, and A. Mashat, "Survey of promising technologies for quantum drones and networks," *IEEE Access*, vol. 9, pp. 125 868–125 911, Sep. 2021.
- [17] D. Pan, G.-L. Long, L. Yin, Y.-B. Sheng, D. Ruan, S. X. Ng, J. Lu, and L. Hanzo, "The evolution of quantum secure direct communication: On the road to the Qinternet," *IEEE Communications Surveys & Tutorials*, vol. 26, no. 3, pp. 1898–1949, Thirdquarter, 2024.
- [18] P. Tedeschi, S. Sciancalepore, and R. Di Pietro, "Satellite-based communications security: A survey of threats, solutions, and research challenges," *Computer Networks*, vol. 216, p. 109246, Oct. 2022.
- [19] M. S. Kumar, R. Ramanathan, and M. Jayakumar, "Key less physical layer security for wireless networks: A survey," *Engineering Science* and Technology, an International Journal, p. 101260, Nov. 2022.
- [20] D. T. Uysal, P. D. Yoo, and K. Taha, "Data-driven malware detection for 6G networks: A survey from the perspective of continuous learning and explainability via visualisation," *IEEE Open Journal of Vehicular Technology*, vol. 4, pp. 61–71, Nov. 2022.
- [21] L. Malina, P. Dzurenda, S. Ricci, J. Hajny, G. Srivastava, R. Matulevičius, A.-A. O. Affia, M. Laurent, N. H. Sultan, and Q. Tang, "Post-quantum era privacy protection for intelligent infrastructures," *IEEE Access*, vol. 9, pp. 36038–36077, Feb. 2021.
- [22] B. Narottama, Z. Mohamed, and S. Aïssa, "Quantum machine learning for next-G wireless communications: Fundamentals and the path ahead," *IEEE Open Journal of the Communications Society*, vol. 4, pp. 2204–2224, Aug. 2023.
- [23] H.-H. Lu, J. M. Lukens, M. Alshowkan, B. T. Kirby, and N. A. Peters, "Building a controlled-NOT gate between polarization and frequency," *Optica Quantum*, vol. 2, no. 4, pp. 282–287, Aug. 2024.
- [24] M. Antesberger, C. M. Richter, F. Poletti, R. Slavík, P. Petropoulos, H. Hübel, A. Trenti, P. Walther, and L. A. Rozema, "Distribution of

- [25] C. Hughes, J. Isaacson, A. Perry, R. F. Sun, and J. Turner, *Quantum computing for the quantum curious*. Springer Nature, 2021.
- [26] I. B. Djordjevic, Quantum communication, quantum networks, and quantum sensing. Academic Press, 2022.
- [27] M. Hajdušek and R. Van Meter, "Quantum communications," arXiv preprint arXiv:2311.02367, Nov. 2023.
- [28] M. CLOUÂTRÉ, M. J. Khojasteh, and M. Z. WIN, "Model-predictive quantum control via hamiltonian learning," *IEEE Transactions on Quantum Engineering*, vol. 3, pp. 1–23, May 2022.
- [29] S. Shanmugam and S. Kalyani, "Unrolling SVT to obtain computationally efficient SVT for n-qubit quantum state tomography," *IEEE Transactions on Signal Processing*, vol. 71, pp. 178–188, Feb. 2023.
- [30] T. Ohyama, Y. Kawamoto, and N. Kato, "Resource allocation optimization by quantum computing for shared use of standalone IRS," *IEEE Transactions on Emerging Topics in Computing*, vol. 11, no. 4, pp. 950–961, Dec. 2023.
- [31] M. Chehimi and W. Saad, "Entanglement rate optimization in heterogeneous quantum communication networks," in *17th International Symposium on Wireless Communication Systems (ISWCS)*, Berlin, Germany, Oct. 2021, pp. 1–6.
- [32] M. Ghaderibaneh, C. Zhan, H. Gupta, and C. Ramakrishnan, "Efficient quantum network communication using optimized entanglement swapping trees," *IEEE Transactions on Quantum Engineering*, vol. 3, pp. 1–20, Apr. 2022.
- [33] W. Dai, T. Peng, and M. Z. Win, "Optimal remote entanglement distribution," *IEEE Journal on Selected Areas in Communications*, vol. 38, no. 3, pp. 540–556, Mar. 2020.
- [34] F. Laudenbach, B. Schrenk, M. Achleitner, N. Vokić, D. Milovančev, and H. Hübel, "Flexible cloud/user-centric entanglement and photon pair distribution with synthesizable optical router," *IEEE Journal of Selected Topics in Quantum Electronics*, vol. 26, no. 3, pp. 1–9, Jun. 2020.
- [35] U. Pereg, "Communication over entanglement-breaking channels with unreliable entanglement assistance," *Physical Review A*, vol. 108, no. 4, p. 042616, Oct. 2023.
- [36] U. Khalid, M. S. Ulum, A. Farooq, T. Q. Duong, O. A. Dobre, and H. Shin, "Quantum semantic communications for metaverse: Principles and challenges," *IEEE Wireless Communications*, vol. 30, no. 4, pp. 26–36, Aug. 2023.
- [37] A. Mittelstädt, L. A. T. Greif, S. T. Jagsch, and A. Schliwa, "Terahertz lasing at room temperature: A numerical study of a vertical-emitting quantum cascade laser based on a quantum dot superlattice," *Physical Review B*, vol. 103, no. 11, p. 115301, Mar. 2021.
- [38] R. Brieger, I. Roth, and M. Kliesch, "Compressive gate set tomography," *Prx quantum*, vol. 4, no. 1, p. 010325, Mar. 2023.
- [39] U. Khalid, J. U. Rehman, S. N. Paing, H. Jung, T. Q. Duong, and H. Shin, "Quantum network engineering in the NISQ age: Principles, missions, and challenges," *IEEE Network*, vol. 38, no. 1, pp. 112–123, Jan. 2024.
- [40] P. Bermejo and R. Orús, "Variational quantum and quantum-inspired clustering," *Scientific Reports*, vol. 13, no. 1, p. 13284, Aug. 2023.
- [41] Y. Fu, Y. Hong, T. Q. Quek, H. Wang, and Z. Shi, "Scheduling policies for quantum key distribution enabled communication networks," *IEEE Wireless Communications Letters*, vol. 9, no. 12, pp. 2126–2129, Dec. 2020.
- [42] B. J. Rollick, G. Siopsis, and B. Qi, "Dynamic attenuation scheme in measurement-device-independent quantum key distribution over turbulent channels," *Physical Review A*, vol. 106, no. 3, p. 032405, Sep. 2022.
- [43] N. Alshaer, A. Moawad, and T. Ismail, "Reliability and security analysis of an entanglement-based QKD protocol in a dynamic ground-to-UAV FSO communications system," *IEEE Access*, vol. 9, pp. 168052– 168067, Dec. 2021.
- [44] M. Winczewski, T. Das, and K. Horodecki, "Limitations on a deviceindependent key secure against a nonsignaling adversary via squashed nonlocality," *Physical Review A*, vol. 106, no. 5, p. 052612, Nov. 2022.
- [45] M. Xu, D. Niyato, Z. Xiong, J. Kang, X. Cao, X. S. Shen, and C. Miao, "Quantum-secured space-air-ground integrated networks: Concept, framework, and case study," *IEEE Wireless Communications*, vol. 30, no. 6, pp. 136–143, Dec. 2023.
- [46] R. Lin and J. Chen, "Modeling and minimizing spontaneous raman scattering for QKD secured DWDM networks," *IEEE Communications Letters*, vol. 25, no. 12, pp. 3918–3921, Dec. 2021.

- [47] H. Wang, Y. Zhao, X. Yu, A. Nag, Z. Ma, J. Wang, L. Yan, and J. Zhang, "Resilient quantum key distribution (QKD)-integrated optical networks with secret-key recovery strategy," *IEEE Access*, vol. 7, pp. 60 079–60 090, May 2019.
- [48] R. Kaewpuang, M. Xu, W. Y. B. Lim, D. Niyato, H. Yu, J. Kang, and X. S. Shen, "Cooperative resource management in quantum key distribution (QKD) networks for semantic communication," *IEEE Internet* of *Things Journal*, vol. 11, no. 3, pp. 4454–4469, Feb. 2024.
- [49] I. Garcia-Cobo and H. D. Menéndez, "Designing large quantum key distribution networks via medoid-based algorithms," *Future Generation Computer Systems*, vol. 115, pp. 814–824, Feb. 2021.
- [50] S. Tarantino, B. Da Lio, D. Cozzolino, and D. Bacco, "Feasibility study of quantum communications in aquatic scenarios," *Optik*, vol. 216, p. 164639, Aug. 2020.
- [51] C. Liu, C. Zhu, X. Liu, M. Nie, H. Yang, and C. Pei, "Multicarrier multiplexing continuous-variable quantum key distribution at terahertz bands under indoor environment and in inter-satellite links communication," *IEEE Photonics Journal*, vol. 13, no. 4, pp. 1–13, Aug. 2021.
- [52] S. N. Paing, J. W. Setiawan, M. A. Ullah, F. Zaman, T. Q. Duong, O. A. Dobre, and H. Shin, "Counterfactual quantum byzantine consensus for human-centric metaverse," *IEEE Journal on Selected Areas in Communications*, vol. 42, no. 4, pp. 905–918, Apr. 2024.
- [53] J. ur Rehman, H. Al-Hraishawi, T. Q. Duong, S. Chatzinotas, and H. Shin, "On estimating time-varying pauli noise," *IEEE Transactions* on Communications, vol. 72, no. 4, pp. 2079–2089, Apr. 2024.
- [54] P. Fuentes, J. E. Martinez, P. M. Crespo, and J. Garcia-Frias, "Design of low-density-generator-matrix-based quantum codes for asymmetric quantum channels," *Physical Review A*, vol. 103, no. 2, p. 022617, Feb. 2021.
- [55] P. Fuentes, J. Etxezarreta Martinez, P. M. Crespo, and J. Garcia-Frias, "Approach for the construction of non-calderbank-steane-shor lowdensity-generator-matrix–based quantum codes," *Physical Review A*, vol. 102, no. 1, p. 012423, Jul. 2020.
- [56] L. Mu, Z. Li, W. Xiao, R. Zhang, P. Wang, T. Liu, G. Min, and K. Li, "A fine-grained end-to-end latency optimization framework for wireless collaborative inference," *IEEE Internet of Things Journal*, vol. 11, no. 4, pp. 5840–5853, Feb. 2024.
- [57] S. Hakemi, M. Houshmand, E. KheirKhah, and S. A. Hosseini, "A review of recent advances in quantum-inspired metaheuristics," *Evolutionary Intelligence*, vol. 17, no. 2, pp. 627–642, Oct. 2022.
- [58] S. Kumar, O. Kaiwartya, M. Rathee, N. Kumar, and J. Lloret, "Toward energy-oriented optimization for green communication in sensor enabled IoT environments," *IEEE Systems Journal*, vol. 14, no. 4, pp. 4663–4673, Dec. 2020.
- [59] H. Y. Chong, H. J. Yap, S. C. Tan, K. S. Yap, and S. Y. Wong, "Advances of metaheuristic algorithms in training neural networks for industrial applications," *Soft Computing*, vol. 25, no. 16, pp. 11209– 11233, May, 2021.
- [60] Y. Liu, C. Li, Y. Zhang, M. Xu, J. Xiao, and J. Zhou, "HPCP-QCWOA: High performance clustering protocol based on quantum clone whale optimization algorithm in integrated energy system," *Future Generation Computer Systems*, vol. 135, pp. 315–332, Oct. 2022.
- [61] J. Xie, Z. Wang, and Y. Chen, "Joint caching and user association optimization for adaptive bitrate video streaming in UAV-assisted cellular networks," *IEEE Access*, vol. 10, pp. 106275–106285, Oct. 2022.
- [62] J. Hou, W. Wang, Y. Zhang, X. Liu, and Y. Xie, "Multi-objective quantum inspired evolutionary SLM scheme for PAPR reduction in multi-carrier modulation," *IEEE Access*, vol. 8, pp. 26 022–26 029, Feb. 2020.
- [63] H. M. Saad, R. K. Chakrabortty, S. Elsayed, and M. J. Ryan, "Quantum-inspired genetic algorithm for resource-constrained projectscheduling," *IEEE Access*, vol. 9, pp. 38488–38502, Feb. 2021.
- [64] D. Zhang, Y. Cui, and T. Zhang, "New quantum-genetic based OLSR protocol (QG-OLSR) for mobile ad hoc network," *Applied Soft Computing*, vol. 80, pp. 285–296, Jul. 2019.
- [65] J. Chen, Y. Zhao, Z. Xu, and H. Zheng, "Resource allocation strategy for D2D-assisted edge computing system with hybrid energy harvesting," *IEEE Access*, vol. 8, pp. 192 643–192 658, Oct. 2020.
- [66] H. Zhang, C. Huang, J. Zhou, and L. Chen, "QoS-aware virtualization resource management mechanism in 5G backhaul heterogeneous networks," *IEEE Access*, vol. 8, pp. 19479–19489, Jan. 2020.
- [67] L. Liang, J. Xiao, Z. Ren, Z. Chen, and Y. Jia, "Particle swarm based service migration scheme in the edge computing environment," *IEEE Access*, vol. 8, pp. 45 596–45 606, Mar. 2020.

- [68] B. Wang, Z. Zhang, Y. Song, M. Chen, and Y. Chu, "Application of quantum particle swarm optimization for task scheduling in deviceedge-cloud cooperative computing," *Engineering Applications of Artificial Intelligence*, vol. 126, p. 107020, Nov. 2023.
- [69] L. Feng, A. Ali, M. Iqbal, F. Ali, I. Raza, M. H. Siddiqi, M. Shafiq, and S. A. Hussain, "Dynamic wireless information and power transfer scheme for nano-empowered vehicular networks," *IEEE Transactions* on *Intelligent Transportation Systems*, vol. 22, no. 7, pp. 4088–4099, Jul. 2021.
- [70] C. Huang, J. Cao, S. Wang, and Y. Zhang, "Dynamic resource scheduling optimization with network coding for multi-user services in the internet of vehicles," *IEEE Access*, vol. 8, pp. 126988–127003, Jun. 2020.
- [71] W. Shu and Y. Li, "Joint offloading strategy based on quantum particle swarm optimization for MEC-enabled vehicular networks," *Digital Communications and Networks*, vol. 9, no. 1, pp. 56–66, Feb. 2023.
- [72] A. Hassan, A. Anter, and M. Kayed, "A robust clustering approach for extending the lifetime of wireless sensor networks in an optimized manner with a novel fitness function," *Sustainable Computing: Informatics and Systems*, vol. 30, p. 100482, Jun. 2021.
- [73] J. Zhou, H. Qin, Y. Zhang, R. Yang, Y. Liu, and C. Li, "Binary quantum elite particle swarm optimization algorithm for spectrum allocation in cognitive wireless medical sensor network," in *Journal of Physics: Conference Series*, vol. 1924, no. 1, May 2021, p. 012030.
- [74] P. B. Fernandes, R. Oliveira, and J. F. Neto, "Trajectory planning of autonomous mobile robots applying a particle swarm optimization algorithm with peaks of diversity," *Applied Soft Computing*, vol. 116, p. 108108, Feb. 2022.
- [75] M. S. Alvarez-Alvarado, F. E. Alban-Chacón, E. A. Lamilla-Rubio, C. D. Rodríguez-Gallegos, and W. Velásquez, "Three novel quantuminspired swarm optimization algorithms using different bounded potential fields," *Scientific Reports*, vol. 11, no. 1, pp. 1–22, Jun. 2021.
- [76] P. Wang and Z. Deng, "A multi-objective quantum-inspired seagull optimization algorithm based on decomposition for unmanned aerial vehicle path planning," *IEEE Access*, vol. 10, pp. 110497–110511, Oct. 2022.
- [77] Y. Su, H. Gao, and S. Zhang, "Energy-efficient resource management for CCFD massive MIMO systems in 6G networks," *Journal of Systems Engineering and Electronics*, vol. 33, no. 4, pp. 877–886, Aug. 2022.
- [78] H. Gao, Y. Su, S. Zhang, Y. Hou, and M. Jo, "Joint antenna selection and power allocation for secure co-time co-frequency full-duplex massive MIMO systems," *IEEE Transactions on Vehicular Technology*, vol. 70, no. 1, pp. 655–665, Jan. 2021.
- [79] S. Zhang, H. Gao, Y. Su, J. Cheng, and M. Jo, "Intelligent mixed reflecting/relaying surface-aided secure wireless communications," *IEEE Transactions on Vehicular Technology*, vol. 73, no. 1, pp. 532–543, Jan. 2024.
- [80] M. Mirhosseini, M. Fazlali, H. T. Malazi, S. K. Izadi, and H. Nezamabadi-pour, "Parallel quadri-valent quantum-inspired gravitational search algorithm on a heterogeneous platform for wireless sensor networks," *Computers & Electrical Engineering*, vol. 92, p. 107085, Jun. 2021.
- [81] A. S. Boev, S. R. Usmanov, A. M. Semenov, M. M. Ushakova, G. V. Salahov, A. S. Mastiukova, E. O. Kiktenko, and A. K. Fedorov, "Quantum-inspired optimization for wavelength assignment," *Frontiers in Physics*, vol. 10, p. 1092065, Jan. 2023.
- [82] B. Wei, H. Song, M. Nakamura, K. Kimura, N. Togawa, and J. Katto, "QuDASH: Quantum-inspired rate adaptation approach for DASH video streaming," *IEEE Access*, vol. 11, pp. 118462–118473, Oct. 2023.
- [83] M. Sugimura, M. Kobayashi, H. Matsumura, X. Wang, and P. Palacharla, "Accelerate optical network modernization through quantum-inspired digital annealing," in *European Conference on Opti*cal Communication (ECOC), Basel, Switzerland, Sep. 2022, pp. 1–4.
- [84] M. Sugimura, M. Javad-Kalbasi, H. Matsumura, X. Wang, P. Palacharla, and S. Valaee, "Speedy and cost-efficient optical network modernization through quantum-inspired computing," in *Optical Fiber Communications Conference and Exhibition (OFC)*, San Diego, CA, USA, Mar. 2023, pp. 1–3.
- [85] Q. J. Lim, C. Ross, A. Ghosh, F. Vook, G. Gradoni, and Z. Peng, "Quantum-assisted combinatorial optimization for reconfigurable intelligent surfaces in smart electromagnetic environments," *IEEE Transactions on Antennas and Propagation*, vol. 72, no. 1, pp. 147–159, Jan. 2024.
- [86] T. Ohyama, Y. Kawamoto, and N. Kato, "Intelligent reflecting surface (IRS) allocation scheduling method using combinatorial optimization

by quantum computing," *IEEE Transactions on Emerging Topics in Computing*, vol. 10, no. 3, pp. 1633–1644, Sep. 2022.

- [87] H. Gao, Y. Su, S. Zhang, and M. Diao, "Antenna selection and power allocation design for 5G massive MIMO uplink networks," *China Communications*, vol. 16, no. 4, pp. 1–15, Apr. 2019.
- [88] X. Xia, Z. Fan, W. Luo, A. Lu, D. Wang, X. Zhao, and X. You, "Joint uplink power control, downlink beamforming, and mode selection for secrecy cell-free massive MIMO with network-assisted full duplexing," *IEEE Systems Journal*, vol. 17, no. 1, pp. 720–731, Mar. 2023.
- [89] M. Y. Shams, A. M. Elshewey, E.-S. M. El-Kenawy, A. Ibrahim, F. M. Talaat, and Z. Tarek, "Water quality prediction using machine learning models based on grid search method," *Multimedia Tools and Applications*, vol. 83, no. 12, pp. 35 307–35 334, Apr. 2024.
- [90] F. Zaman, A. Farooq, M. A. Ullah, H. Jung, H. Shin, and M. Z. Win, "Quantum machine intelligence for 6G URLLC," *IEEE Wireless Communications*, vol. 30, no. 2, pp. 22–30, Apr. 2023.
- [91] J. Adu Ansere, D. T. Tran, O. A. Dobre, H. Shin, G. K. Karagiannidis, and T. Q. Duong, "Energy-efficient optimization for mobile edge computing with quantum machine learning," *IEEE Wireless Communications Letters*, vol. 13, no. 3, pp. 661–665, Mar. 2024.
- [92] J. A. Ansere, T. Q. Duong, S. R. Khosravirad, V. Sharma, A. Masaracchia, and O. A. Dobre, "Quantum deep reinforcement learning for 6G mobile edge computing-based IoT systems," in *International Wireless Communications and Mobile Computing (IWCMC)*, Marrakesh, Morocco, Jul. 2023, pp. 406–411.
- [93] J. A. Ansere, E. Gyamfi, V. Sharma, H. Shin, O. A. Dobre, and T. Q. Duong, "Quantum deep reinforcement learning for dynamic resource allocation in mobile edge computing-based IoT systems," *IEEE Transactions on Wireless Communications*, vol. 23, no. 6, pp. 6221–6233, Jun. 2024.
- [94] D. K. Jain, Y. Li, M. J. Er, Q. Xin, D. Gupta, and K. Shankar, "Enabling unmanned aerial vehicle borne secure communication with classification framework for industry 5.0," *IEEE Transactions on Industrial Informatics*, vol. 18, no. 8, pp. 5477–5484, Aug. 2022.
- [95] S. Sharif, S. Zeadally, and W. Ejaz, "Resource optimization in UAV-assisted IoT networks: The role of generative AI," 2024. [Online]. Available: https://arxiv.org/abs/2405.03863
- [96] H. K. Ahmed, B. Tantawi, M. Magdy, and G. I. Sayed, A quantum machine learning model for medical data classification. Springer Nature Switzerland, Aug. 2023, pp. 95–114.
- [97] T. Li, S. Zhang, and J. Xia, "Quantum generative adversarial network: A survey," *Computers, Materials & Continua*, vol. 64, no. 1, pp. 401– 438, May 2020.
- [98] M. Xu, D. Niyato, J. Kang, Z. Xiong, Y. Cao, Y. Gao, C. Ren, and H. Yu, "Generative AI-enabled quantum computing networks and intelligent resource allocation," Jan. 2024. [Online]. Available: https://arxiv.org/abs/2401.07120
- [99] K. Gili, "Do quantum models make good generative learners?" Ph.D. dissertation, University of Oxford, Nov. 2023.
- [100] S. Pise, A. A. Agarkar, and S. Jain, "Unleashing the power of generative AI and quantum computing for mutual advancements," in *3rd Asian Conference on Innovation in Technology (ASIANCON)*, Ravet, India, Aug. 2023, pp. 1–7.
- [101] C.-S. Lee, M.-H. Wang, J.-K. Chiang, N. Kubota, E. Sato-Shimokawara, Y. Nojima, G. Acampora, P.-Y. Wu, S.-C. Chiu, S.-C. Yang *et al.*, "Quantum computational intelligence with generative AI image for human-machine interaction," in *IEEE International Conference on Fuzzy Systems (FUZZ-IEEE)*, Yokohama, Japan, Jul. 2024, pp. 1–8.
- [102] Z. Qu, Y. Li, B. Liu, D. Gupta, and P. Tiwari, "DTQFL: A digital twinassisted quantum federated learning algorithm for intelligent diagnosis in 5G mobile network," *IEEE Journal of Biomedical and Health Informatics*, pp. 1–10, Aug. 2023.
- [103] S. Park, S. Jung, and J. Kim, "Dynamic quantum federated learning for satellite-ground integrated systems using slimmable quantum neural networks," *IEEE Access*, vol. 12, pp. 58 239–58 247, Apr. 2024.
- [104] Q. Xia and Q. Li, "QuantumFed: A federated learning framework for collaborative quantum training," in *IEEE Global Communications Conference (GLOBECOM)*, Madrid, Spain, Dec. 2021, pp. 1–6.
- [105] M. Chehimi and W. Saad, "Quantum federated learning with quantum data," in *IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP)*, Singapore, May 2022, pp. 8617–8621.
- [106] B. Narottama, T. Jamaluddin, and S. Y. Shin, "Quantum neural network with parallel training for wireless resource optimization," *IEEE Transactions on Mobile Computing*, pp. 1–13, Oct. 2023.
- [107] N. P. Eberechukwu, M. Jeong, H. Park, S. W. Choi, and S. Kim, "Fingerprinting-based indoor localization with hybrid quantum-deep

neural network," IEEE Access, vol. 11, pp. 142276-142291, Dec. 2023.

- [108] B. Narottama and T. Q. Duong, "Quantum neural networks for optimal resource allocation in cell-free MIMO systems," in *IEEE Global Communications Conference (GLOBECOM)*, Rio de Janeiro, Brazi, Jan. 2023, pp. 2444–2449.
- [109] O. P. Patel, N. Bharill, A. Tiwari, and M. Prasad, "A novel quantuminspired fuzzy based neural network for data classification," *IEEE Transactions on Emerging Topics in Computing*, vol. 9, no. 2, pp. 1031–1044, Feb. 2019.
- [110] D. Yan, H. Cao, Y. Yu, Y. Wang, and X. Yu, "Singleobjective/multiobjective cat swarm optimization clustering analysis for data partition," *IEEE Transactions on Automation Science and Engineering*, vol. 17, no. 3, pp. 1633–1646, Jul. 2020.
- [111] M. Islam, M. Chowdhury, Z. Khan, and S. M. Khan, "Hybrid quantumclassical neural network for cloud-supported in-vehicle cyberattack detection," *IEEE Sensors Letters*, vol. 6, no. 4, pp. 1–4, Apr. 2022.
- [112] M. S. Salek, P. K. Biswas, J. Pollard, J. Hales, Z. Shen, V. Dixit, M. Chowdhury, S. M. Khan, and Y. Wang, "A novel hybrid quantumclassical framework for an in-vehicle controller area network intrusion detection," *IEEE Access*, vol. 11, pp. 96081–96092, Aug. 2023.
- [113] C. N. Selvin, V. Varadarajan, A. Panneerselvam, R. Jothilakshmi et al., "AVQS-NN: Adaptive virtualization for quantum services for application support using neural networks," *Applied Soft Computing*, vol. 136, p. 110075, Mar. 2023.
- [114] C. Ding, T.-Y. Bao, and H.-L. Huang, "Quantum-inspired support vector machine," *IEEE Transactions on Neural Networks and Learning Systems*, vol. 33, no. 12, pp. 7210–7222, Dec. 2022.
- [115] A. Jaiswal, S. Kumar, O. Kaiwartya, P. K. Kashyap, E. Kanjo, N. Kumar, and H. Song, "Quantum learning-enabled green communication for next-generation wireless systems," *IEEE Transactions on Green Communications and Networking*, vol. 5, no. 3, pp. 1015–1028, Sep. 2021.
- [116] P. Lin, Q. Song, D. Wang, F. R. Yu, L. Guo, and V. C. Leung, "Resource management for pervasive-edge-computing-assisted wireless VR streaming in industrial internet of things," *IEEE Transactions on Industrial Informatics*, vol. 17, no. 11, pp. 7607–7617, Nov. 2021.
- [117] D. Wang, B. Song, P. Lin, F. R. Yu, X. Du, and M. Guizani, "Resource management for edge intelligence (EI)-assisted IoV using quantuminspired reinforcement learning," *IEEE Internet of Things Journal*, vol. 9, no. 14, pp. 12588–12600, Jul. 2022.
- [118] Y. Ren, R. Xie, F. R. Yu, T. Huang, and Y. Liu, "Green intelligence networking for connected and autonomous vehicles in smart cities," *IEEE Transactions on Green Communications and Networking*, vol. 6, no. 3, pp. 1591–1603, Sep. 2022.
- [119] Y. Ren, R. Xie, F. R. Yu, T. Huang, and Y. Liu, "Quantum collective learning and many-to-many matching game in the metaverse for connected and autonomous vehicles," *IEEE Transactions on Vehicular Technology*, vol. 71, no. 11, pp. 12128–12139, Nov. 2022.
- [120] Y. Ren, R. Xie, F. R. Yu, T. Huang, and Y. Liu, "NFT-based intelligence networking for connected and autonomous vehicles: A quantum reinforcement learning approach," *IEEE Network*, vol. 36, no. 6, pp. 116–124, Dec. 2022.
- [121] S. Lin, L. Zhu, F. R. Yu, and Y. Li, "Collaborative computing optimization in train-edge-cloud-based smart train systems using risk-sensitive reinforcement learning," *IEEE Transactions on Vehicular Technology*, vol. 73, no. 3, pp. 3129–3141, Mar. 2024.
- [122] M. Mekala, G. Srivastava, A. H. Gandomi, J. H. Park, and H.-Y. Jung, "A quantum-inspired sensor consolidation measurement approach for cyber-physical systems," *IEEE Transactions on Network Science and Engineering*, vol. 11, no. 1, pp. 511–524, Feb. 2024.
- [123] Y. Li, A. H. Aghvami, and D. Dong, "Intelligent trajectory planning in UAV-mounted wireless networks: A quantum-inspired reinforcement learning perspective," *IEEE Wireless Communications Letters*, vol. 10, no. 9, pp. 1994–1998, Sep. 2021.
- [124] P. Lin, Q. Song, F. R. Yu, D. Wang, A. Jamalipour, and L. Guo, "Wireless virtual reality in beyond 5G systems with the internet of intelligence," *IEEE Wireless Communications*, vol. 28, no. 2, pp. 70– 77, Apr. 2021.
- [125] Y. Chen, Y. Pan, and D. Dong, "Residual tensor train: A quantuminspired approach for learning multiple multilinear correlations," *IEEE Transactions on Artificial Intelligence*, vol. 4, no. 5, pp. 1101–1113, Oct. 2023.
- [126] Y. Li, A. H. Aghvami, and D. Dong, "Path planning for cellularconnected UAV: A DRL solution with quantum-inspired experience replay," *IEEE Transactions on Wireless Communications*, vol. 21, no. 10, pp. 7897–7912, Oct. 2022.

- [127] S. K. Borah, S. Anand, R. V. Agarwal, and S. Bitragunta, "Analysis of adversarial jamming from a quantum game theoretic perspective," *IEEE Systems Journal*, vol. 17, no. 1, pp. 881–891, Mar. 2022.
- [128] Y. Sano, M. Norimoto, and N. Ishikawa, "Qubit reduction and quantum speedup for wireless channel assignment problem," *IEEE Transactions* on *Quantum Engineering*, vol. 4, pp. 1–12, Jul. 2023.
- [129] M. D. Migliore, "Classical and quantum processing in the deep physical layer," *IEEE Access*, vol. 11, pp. 52 969–52 982, May 2023.
- [130] N. Innan, O. I. Siddiqui, S. Arora, T. Ghosh, Y. P. Koçak, D. Paragas, A. A. O. Galib, M. A.-Z. Khan, and M. Bennai, "Quantum state tomography using quantum machine learning," *Quantum Machine Intelligence*, vol. 6, no. 1, p. 28, May 2024.
- [131] B. Cho, Y. Xiao, P. Hui, and D. Dong, "Quantum bandit with amplitude amplification exploration in an adversarial environment," *IEEE Transactions on Knowledge and Data Engineering*, vol. 36, no. 1, pp. 311–317, Jan. 2024.
- [132] J. Cui, Y. Xiong, S. X. Ng, and L. Hanzo, "Quantum approximate optimization algorithm based maximum likelihood detection," *IEEE Transactions on Communications*, vol. 70, no. 8, pp. 5386–5400, Aug. 2022.
- [133] T. V. Le and V. Kekatos, "Solving constrained optimization problems via the variational quantum eigensolver with constraints," *Physical Review A*, vol. 110, no. 2, p. 022430, Aug. 2024.
- [134] M. Li, C. Liu, K. Li, X. Liao, and K. Li, "Multi-task allocation with an optimized quantum particle swarm method," *Applied Soft Computing*, vol. 96, p. 106603, Nov. 2020.
- [135] Q. Wei, H. Ma, C. Chen, and D. Dong, "Deep reinforcement learning with quantum-inspired experience replay," *IEEE Transactions on Cybernetics*, vol. 52, no. 9, pp. 9326–9338, Sep. 2022.
- [136] O. Şeker, N. Tanoumand, and M. Bodur, "Digital annealer for quadratic unconstrained binary optimization: A comparative performance analysis," *Applied Soft Computing*, vol. 127, p. 109367, Sep. 2022.
- [137] L. Fan and Z. Han, "Hybrid quantum-classical computing for future network optimization," *IEEE Network*, vol. 36, no. 5, pp. 72–76, Oct. 2022.
- [138] D. Leib, T. Seidel, S. Jäger, R. Heese, C. Jones, A. Awasthi, A. Niederle, and M. Bortz, "An optimization case study for solving a transport robot scheduling problem on quantum-hybrid and quantuminspired hardware," *Scientific Reports*, vol. 13, no. 1, p. 18743, Oct. 2023.
- [139] G. Colucci, S. van der Linde, and F. Phillipson, "Power network optimization: A quantum approach," *IEEE Access*, vol. 11, pp. 98926– 98938, Sep. 2023.
- [140] F. F. Silva, P. M. Carvalho, L. A. Ferreira, and Y. Omar, "A QUBO formulation for minimum loss network reconfiguration," *IEEE Transactions on Power Systems*, vol. 38, no. 5, pp. 4559–4571, Sep. 2023.
- [141] M. Y. Naghmouchi and W. d. S. Coelho, "Mixed-integer linear programming solver using Benders decomposition assisted by a neutralatom quantum processor," *Physical Review A*, vol. 110, no. 1, p. 012434, Jul. 2024.
- [142] S. Tu, O. U. Rehman, S. U. Rehman, S. Ullah, M. Waqas, and R. Zhu, "A novel quantum inspired particle swarm optimization algorithm for electromagnetic applications," *IEEE Access*, vol. 8, pp. 21909–21916, Jan. 2020.
- [143] M. Bhatia and S. K. Sood, "Quantum computing-inspired network optimization for IoT applications," *IEEE Internet of Things Journal*, vol. 7, no. 6, pp. 5590–5598, Jun. 2020.
- [144] S. M. Asaad and H. S. Maghdid, "A comprehensive review of indoor/outdoor localization solutions in IoT era: Research challenges and future perspectives," *Computer Networks*, vol. 212, p. 109041, Jul. 2022.
- [145] B. Narottama and S. Y. Shin, "Quantum neural networks for resource allocation in wireless communications," *IEEE Transactions on Wireless Communications*, vol. 21, no. 2, pp. 1103–1116, Feb. 2022.
- [146] M. D. Phung and Q. P. Ha, "Motion-encoded particle swarm optimization for moving target search using UAVs," *Applied Soft Computing*, vol. 97, p. 106705, Dec. 2020.
- [147] B. Ma, Z. Liu, W. Zhao, J. Yuan, H. Long, X. Wang, and Z. Yuan, "Target tracking control of UAV through deep reinforcement learning," *IEEE Transactions on Intelligent Transportation Systems*, vol. 24, no. 6, pp. 5983–6000, Jun. 2023.
- [148] Y. Zhang, D. J. Love, J. V. Krogmeier, C. R. Anderson, R. W. Heath, and D. R. Buckmaster, "Challenges and opportunities of future rural wireless communications," *IEEE Communications Magazine*, vol. 59, no. 12, pp. 16–22, Dec. 2021.
- [149] M. Abdel-Basset, L. Abdel-Fatah, K. A. Eldrandaly, and N. M. Abdel-Aziz, "Enhanced computational intelligence algorithm for coverage

optimization of 6G Non-terrestrial networks in 3D space," *IEEE Access*, vol. 9, pp. 70419–70429, May 2021.

- [150] G. Acampora, A. Chiatto, and A. Vitiello, "Genetic algorithms as classical optimizer for the quantum approximate optimization algorithm," *Applied Soft Computing*, vol. 142, p. 110296, Jul. 2023.
- [151] C. Ding, T.-Y. Bao, and H.-L. Huang, "Quantum-inspired support vector machine," *IEEE Transactions on Neural Networks and Learning Systems*, vol. 33, no. 12, pp. 7210–7222, Dec. 2022.
- [152] D. Bern, "Quantum annealing algorithms for PAPR minimisation in wireless networks," M.S. Thesis, Department of Physics and Astronomy, Uppsala University, Uppsala, Sweden, 2022.
- [153] J. Cao, P. Yu, X. Xiang, M. Ma, and H. Li, "Anti-quantum fast authentication and data transmission scheme for massive devices in 5G NB-IoT system," *IEEE Internet of Things Journal*, vol. 6, no. 6, pp. 9794–9805, Dec. 2019.
- [154] H.-T. Peng, J. C. Lederman, L. Xu, T. F. De Lima, C. Huang, B. J. Shastri, D. Rosenbluth, and P. R. Prucnal, "A photonics-inspired compact network: Toward real-time AI processing in communication systems," *IEEE Journal of Selected Topics in Quantum Electronics*, vol. 28, no. 4, pp. 1–17, Aug. 2022.
- [155] O. Bouchmal, B. Cimoli, R. Stabile, J. J. Vegas Olmos, and I. Tafur Monroy, "From classical to quantum machine learning: Survey on routing optimization in 6G software defined networking," *Frontiers in Communications and Networks*, vol. 4, p. 1220227, Nov. 2023.
- [156] H.-H. Lu, N. A. Peters, A. M. Weiner, and J. M. Lukens, "Characterization of quantum frequency processors," *IEEE Journal of Selected Topics in Quantum Electronics*, vol. 29, no. 6, pp. 1–12, Dec. 2023.



**Trung Q. Duong** (Fellow, IEEE) is a Canada Excellence Research Chair (CERC) and a Full Professor at Memorial University, Canada. He is also the adjunct Chair Professor in Telecommunications at Queen's University Belfast, UK and a Research Chair of Royal Academy of Engineering, UK. He was a Distinguished Advisory Professor at Inje University, South Korea (2017-2019), an Adjunct Professor and the Director of Institute for AI and Big Data at Duy Tan University, Vietnam (2012-present), and a Visiting Professor (under Eminent Scholar program)

at Kyung Hee University, South Korea (2023-2025). His current research interests include quantum communications, wireless communications, quantum machine learning, and optimisation.

Dr. Duong has served as an Editor/Guest Editor for the IEEE TRANS-ACTIONS ON WIRELESS COMMUNICATIONS, IEEE TRANSACTIONS ON COMMUNICATIONS, IEEE TRANSACTIONS ON VEHICULAR TECHNOLOGY, IEEE COMMUNICATIONS LETTERS, IEEE WIRELESS COMMUNICATIONS LETTERS, IEEE WIRELESS COMMUNICATIONS, IEEE COMMUNICATIONS MAGAZINES, and IEEE JOURNAL ON SELECTED AREAS IN COMMUNICA-TIONS. He received the Best Paper Award at the IEEE VTC-Spring 2013, IEEE ICC 2014, IEEE GLOBECOM 2016, 2019, 2022, IEEE DSP 2017, IWCMC 2019, 2023, 2024 and IEEE CAMAD 2023, 2024. He has received the two prestigious awards from the Royal Academy of Engineering (RAEng): RAEng Research Chair (2021-2025) and the RAEng Research Fellow (2015-2020). He is the recipient of the prestigious Newton Prize 2017. He is a Fellow of Engineering Institute of Canada (EIC).



Muhammad Omair Butt (Student Member, IEEE) received the bachelor's degree in telecommunication engineering from FAST University, Pakistan, and the master's degree in electrical engineering from King Fahd University of Petroleum and Minerals, Saudi Arabia. He is currently pursuing the Ph.D. degree with the Department of Electrical and Computer Engineering, Lakehead University, Canada. His research interests include resource management in wireless networks, quantum communications, mobile edge computing, quantum-inspired algorithms,

ultrareliable and low-latency communication.



Waleed Ejaz is an Associate Professor in the Department of Electrical and Computer Engineering at Lakehead University, Barrie Campus, ON, Canada. He obtained his Ph.D. in Information and Communication Engineering from Sejong University, Republic of Korea. Dr. Ejaz has co-authored over 150 papers in prestigious journals and conferences and 3 books. His research interests include communication and networking technologies for the Internet of Things (IoT), 6G and beyond networks, machine learning and AI for wireless networks, and mobile edge

computing. He is serving as an Associate Editor of the IEEE Open Journal of Communication Society. He served as an Associate Editor of IEEE Communications Magazine and IEEE Canadian Journal of Electrical and Computer Engineering. He is a registered Professional Engineer (P.Eng.) in Ontario, Canada and a senior member of IEEE.



Nazar Waheed is a Senior Lecturer in Faculty of CIS at the Higher Colleges of Technology (HCT), UAE. He holds a Ph.D. in Engineering and Information Technology from the University of Technology Sydney (UTS), Australia. Previously, he served as a Lecturer and Senior Lecturer positions at several Australian universities. Dr. Waheed's research focuses on cybersecurity, Internet of Things (IoT) security, and data privacy. He has published extensively in peer-reviewed journals and conferences, contributing to advancements in cyber security and

secure IoT frameworks. He is a Cisco Certified Academy Instructor (CCAI) and has actively contributed to competency-based education programs through curriculum development and training. Dr. Waheed is passionate about mentoring students and fostering innovation in secure and efficient networking technologies.